## IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:	Case No. 08-12606 (BLS)
VERASUN ENERGY CORPORATION, <u>et al.</u> , )  Debtors. 1	Chapter 11 Jointly Administered
)	Related Docket No. 1642, 1950

## FINDINGS OF FACT, CONCLUSIONS OF LAW, AND ORDER UNDER 11 U.S.C. §§ 1129(a) AND (b) AND FED. R. BANKR. P. 3020 CONFIRMING THE JOINT PLAN OF LIQUIDATION OF VERASUN ENERGY CORPORATION AND ITS AFFILIATED DEBTORS

On October 23, 2009, this Court conducted a hearing (the "Confirmation Hearing")<sup>2</sup> to consider confirmation of the Joint Plan of Liquidation of VeraSun Energy Corporation and Its Affiliated Debtors (collectively, the "Debtors") (Docket No. 1642) as modified by the modifications set forth in Exhibit A hereto (as amended, modified or supplemented, the "Plan"), a copy of which is attached hereto as Exhibit B and the Court having reviewed and considered (i) the Plan and the Disclosure Statement (as that term is defined

Unless otherwise defined, capitalized terms used herein shall have the meanings ascribed to them in the Plan. Any term used in the Plan or this order (the "Confirmation Order") that is not defined in the Plan or this Confirmation Order, but that is used in the United States Bankruptcy Code, 11 U.S.C. "101-1330, as amended (the "Bankruptcy Code"), or the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), shall have the meaning ascribed to that term in the Bankruptcy Code or the Bankruptcy Rules.



0812606091023000000000003

The Debtors consist of: VeraSun Energy Corporation (EIN: 20-3430241); ASA OpCo Holdings, LLC (EIN: 68-0609122); US BioEnergy Corporation (EIN: 20-1811472); VeraSun Albert City, LLC (EIN: 20-2264707); VeraSun Albion, LLC (EIN: 55-0907221); VeraSun Aurora Corporation (EIN: 46-0462174); VeraSun BioDiesel, LLC (EIN: 20-3790860); VeraSun Bloomingburg, LLC (EIN: 55-0907224); VeraSun Central City, LLC (EIN: 55-0816855); VeraSun Charles City, LLC (EIN: 20-3735184); VeraSun Dyersville, LLC (EIN: 26-3594945); VeraSun Fort Dodge, LLC (EIN: 42-1630527); VeraSun Granite City, LLC (EIN: 20-5909621); VeraSun Hankinson, LLC (EIN: 20-2852457); VeraSun Hartley, LLC (EIN: 20-5381200); VeraSun Janesville, LLC (EIN: 20-4420290); VeraSun Linden, LLC (EIN: 55-0907228); VeraSun Litchfield, LLC (EIN: 20-8621370); VeraSun Marion, LLC (EIN: 20-3477343); VeraSun Marketing, LLC (EIN: 20-3693800); VeraSun Ord, LLC (EIN: 75-3204878); VeraSun Reynolds, LLC (EIN: 20-5914827); VeraSun Tilton, LLC (EIN: 26-1539139); VeraSun Welcome, LLC (EIN: 20-4115888); VeraSun Woodbury, LLC (EIN: 20-0647425).

below), (ii) the Affidavit of Service of Kurtzman Carson Consulting, LLC ("KCC") on Mailing of Solicitation Packages (the "KCC Declaration"), filed on September 28, 2009 (Docket No. 1875) and the Affidavit of Romelia A. Edwards With Respect to the Tabulation of Votes on the Joint Plan of Liquidation of VeraSun Energy Corporation and its Affiliated Debtors Under Chapter 11 of the Bankruptcy Code (the "Edwards Declaration"), filed on October 20, 2009 (Docket No. 1940), (iii) the Memorandum in Support of the Joint Plan of Liquidation of VeraSun Energy Corporation and its Affiliated Debtors (the "Memorandum"), filed on October 21, 2009 (Docket No. 1950, (iv) the Declaration of Mark Dickey, the Senior Vice President, General Counsel and Secretary of VeraSun Energy Corporation (the "Dickey Declaration") filed by the Debtors on October 21, 2009 (Docket No. 1949, (v) all of the evidence proffered or adduced at, objections filed in connection with, and arguments of counsel made at, the Confirmation Hearing; having taken judicial notice of the entire record of these Chapter 11 Cases; and after due deliberation thereon and good and sufficient cause appearing therefor, the Court hereby makes the following findings of fact and conclusions of law in respect of confirmation of the Plan.<sup>3</sup>

## THE COURT FINDS AND CONCLUDES THAT:

- A. <u>Filing of Plan</u>. On July 31, 2009, the Debtors filed both the Disclosure Statement with Respect to the Joint Plan of Liquidation of VeraSun Energy Corporation and Its Affiliated Debtors (the "<u>Disclosure Statement</u>") and the Plan (Docket Nos. 1641 and 1642, respectively).
- B. <u>Solicitation Procedures Order</u>. On September 10, 2009, this Court entered its Order Under Bankruptcy Code Sections 105, 502, 1125 and 1126, Bankruptcy Rules 2002, 3003, 3017, 3018, 3020 and 9007 and Local Bankruptcy Rule 3017-1 (I) Approving the Form and

Findings of fact shall be construed as conclusions of law and conclusions of law shall be construed as findings of fact when appropriate. See Fed. R. Bankr. P. 7052.

Manner of Notice of the Disclosure Statement Hearing; (II) Determining Adequacy of the Disclosure Statement; (III) Establishing a Record Date and Voting, Objection and Other Deadlines with Respect to Confirmation of Plan; (IV) Scheduling Hearing on Confirmation of Plan; (V) Approving (A) the Form and Manner of Notice of the Hearing on Confirmation and Related Matters and (B) the Deadline for Filing Confirmation Objections; (VI) Approving the Solicitation Procedures; and (VII) Granting Certain Other Relief (the "Solicitation Procedures Order") (Docket No. 1797) that, among other things, (i) approved the Disclosure Statement as containing adequate information within the meaning of section 1125 of the Bankruptcy Code and Fed. R. Bankr. P. 3017, (ii) fixed October 23, 2009 as the date for the commencement of the Confirmation Hearing, (iii) approved the form and method of notice of the Confirmation Hearing (the "Confirmation Hearing Notice"), (iv) established certain procedures for soliciting and tabulating votes with respect to the Plan, and (v) established a voting record date and voting deadline (the "Yoting Record Date" and "Yoting Deadline" respectively).

C. Transmittal of Solicitation Package. Solicitation packages containing (i) the applicable ballot; (ii) the Solicitation Procedures Order; (iii) Notice of (a) Approval of Disclosure Statement, (b) Hearing on Confirmation of Plan, (c) Deadline and Procedures for Filing Objections to Confirmation of Plan, (d) Record Date and (e) Voting Deadline for Receipt of Ballots; and (iv) a CD-ROM containing (a) the Confirmation Hearing Notice, (b) Disclosure Statement with Respect to the Joint Plan of Liquidation of VeraSun Energy Corporation and its Affiliated Debtors (Docket No. 1641), (c) the Plan and (d) the Solicitation Procedures Order (the "Solicitation Packages"), were transmitted to Classes 2E, 4E through 11E, 1F through 11F, 1G through 11G and 3H (collectively, the "Voting Classes"), in accordance with Fed. R. Bankr. P. 3017(d) and the Solicitation Procedures Order, all as set forth in the KCC Declaration. In

addition, Solicitation Packages were transmitted to the Unimpaired Classes in Classes 2A, 4A through 11A, 1B through 11B, 1C, 1D through 11D that contained notices of nonvoting status in lieu of Ballots, also as set forth in the KCC Declaration.

- D. <u>Publication of Confirmation Hearing Notice</u>. The Debtors published the Confirmation Hearing Notice on September 15, 2009, in the <u>Wall Street Journal</u> (national edition) and the <u>Argus Leader</u> as evidenced by the affidavits of publication filed by representatives from the respective newspapers (the "<u>Affidavits of Publication</u>") on October 8, 2009 (Docket Nos. 1899 and 1898).
- E. Edwards Declaration. On October 20, 2009, the Voting Agent, Kurtzman Carson Consultants, filed the Edwards Declaration certifying the method and results of the Ballot tabulation for each of the Voting Classes voting to accept or reject the Plan (Docket No. 1940).
- F. Exclusive Jurisdiction; Venue; Core Proceeding (28 U.S.C. §§ 157(b)(2) and 1334(a)). The Court has jurisdiction over the Chapter 11 Cases pursuant to 28 U.S.C. §§ 157 and 1334. Venue is proper pursuant to 28 U.S.C. §§ 1408 and 1409. Confirmation of the Plan is a core proceeding under 28 U.S.C. § 157(b)(2), and the Court has exclusive jurisdiction to determine whether the Plan complies with the applicable provisions of the Bankruptcy Code and should be confirmed.
- G. <u>Judicial Notice</u>. The Court takes judicial notice of the docket of these Chapter 11 Cases maintained by the Clerk of the Court and/or its duly-appointed agent, including, without limitation, all pleadings and other documents filed, all orders entered, and all evidence and arguments made, proffered or adduced at, the hearings held before the Court during the pendency of these Chapter 11 Cases.

- H. Transmittal and Mailing of Materials; Notice. Due, adequate and sufficient notice of the Disclosure Statement and Plan and of the Confirmation Hearing, along with all deadlines for voting on or filing objections to the Plan, has been given to all known Holders of Claims and Interests in accordance with the procedures set forth in the Solicitation Procedures Order. The Disclosure Statement, Plan, Ballots, Solicitation Procedures Order and the Confirmation Hearing Notice were transmitted and served in substantial compliance with the Solicitation Procedures Order and the Bankruptcy Rules, and such transmittal and service were adequate and sufficient. Adequate and sufficient notice of the Confirmation Hearing and the other bar dates and hearings described in the Solicitation Procedures Order was given in compliance with the Bankruptcy Rules and the Solicitation Procedures Order, and no other or further notice is or shall be required.
- I. <u>Solicitation</u>. Votes for acceptance or rejection of the Plan were solicited in good faith and complied with sections 1125 and 1126 of the Bankruptcy Code, Bankruptcy Rules 3017 and 3018, the Disclosure Statement, the Solicitation Procedures Order, all other applicable provisions of the Bankruptcy Code, and all other rules, laws, and regulations.
- J. <u>Ballots</u>. All procedures used to distribute solicitation materials to the applicable Holders of Claims and Interests and to tabulate the Ballots were fair and conducted in accordance with the Solicitation Procedures Order, the Bankruptcy Code, the Bankruptcy Rules, the local rules of the Bankruptcy Court for the District of Delaware (the "<u>Local Rules</u>"), and all other applicable rules, laws, and regulations.
- K. Impaired Classes that Have Voted to Accept the Plan. As evidenced by the Edwards Declaration and evidence proffered or adduced at the Confirmation Hearing, which certified both the method and results of the voting, the following Classes that were entitled to vote have accepted the Plan pursuant to the requirements of sections 1124 and 1126 of the

Bankruptcy Code: Classes 2E and 4E through 10E, 2F, 6F, 7F, 10F, 1G through 4G, 6G through 8G, 10G, 11G, and 3H. Thus, at least one Impaired Class of Claims, determined without including any acceptance by an insider of any of the Debtors, has voted to accept the Plan with respect to the Debtors.

- L. <u>Classes 1I-11I Deemed to Have Rejected the Plan</u>. Classes 1I-11I are not entitled to receive any Distribution under the Plan on account of their Interests and, therefore, votes to accept or reject the Plan were not solicited from Holders of Interests in such Classes.
- M. <u>Burden of Proof.</u> The Debtors, as proponents of the Plan, have met their burden of proving the elements of sections 1129(a) and (b) of the Bankruptcy Code, by a preponderance of evidence, which is the applicable evidentiary standard in this Court. The Court also finds that the Debtors have satisfied the elements of sections 1129(a) and (b) of the Bankruptcy Code under the clear and convincing standard of proof.
- N. Plan Compliance With Bankruptcy Code (11 U.S.C. § 1129(a)(1)). The Plan complies with the applicable provisions of the Bankruptcy Code, thereby satisfying section 1129(a)(1) of the Bankruptcy Code.
- 1. Proper Classification (11 U.S.C. §§ 1122, 1123(a)(1)). In addition to Administrative Claims and Tax Priority Claims (which are not required to be classified), Article III of the Plan designates sixty-five (65) Classes of Claims and eleven (11) Classes of Interests for the Debtors. The Claims and Interests placed in each Class are substantially similar to other Claims or Interests in each such Class. Valid business, factual and legal reasons exist for separately classifying the various Classes of Claims and Interests created under the Plan, and such Classes do not unfairly discriminate between Holders of Claims or Interests. Thus, the Plan satisfies sections 1122 and 1123(a)(1) of the Bankruptcy Code.
- 2. Specification of Unimpaired Classes (11 U.S.C. § 1123(a)(2)). Section 4.1 of the Plan specifies the Classes of Claims and Interests that are Unimpaired. Thus, the Plan satisfies section 1123(a)(2) of the Bankruptcy Code.
- 3. Specification of Treatment of Impaired Classes (11 U.S.C. § 1123(a)(3)). Section 4.2 of the Plan specifies the Classes of Claims and Interests that are Impaired under the

Plan. Article V of the Plan specifies the treatment of Claims in all such Classes. Thus, the Plan satisfies section 1123(a)(3) of the Bankruptcy Code.

- 4. <u>No Discrimination (11 U.S.C. § 1123(a)(4))</u>. The Plan provides for the same treatment by the Debtors for each Claim in each respective Class unless the Holder of a particular Claim has agreed to less favorable treatment with respect to such Claim. Thus, the Plan satisfies section 1123(a)(4) of the Bankruptcy Code.
- 5. Adequate Means for Implementation of Plan (11 U.S.C. § 1123(a)(5)). Article VII of the Plan, entitled "Means for Implementation of the Plan," sets forth numerous provisions to facilitate implementation of the Plan, such as the continued corporate existence of the Reorganized Debtors, the revesting of all property of the Estates in the Reorganized Debtors, generally allowing for all corporate action to effectuate the Plan, the appointment, rights and obligations of the Plan Administrator, the establishment, rights and obligations of the Plan Committee, cancellation of securities, instruments and agreements evidencing Claims and Interests, and the settlement of certain Intercompany Claims. Thus, the Plan satisfies section 1123(a)(5) of the Bankruptcy Code.
- 6. Prohibition Against Issuance of Non-voting Equity Securities and Provisions for Voting Power of Classes of Securities (11 U.S.C. § 1123(a)(6)). Section 7.3 of the Plan provides that the certificate of incorporation and by-laws of each Debtor, among other things, will prohibit the issuance of non-voting equity securities to the extent prohibited by section 1123(a)(6) of the Bankruptcy Code. These documents were filed on October 13, 2009 (Docket No. 1920). Thus, the requirements of section 1123(a)(6) of the Bankruptcy Code are satisfied.
- of the Plan properly and adequately discloses or otherwise identifies the procedures for determining the identities and affiliations of all individuals or entities proposed to serve on or after the Effective Date as officers and directors of the Reorganized Debtors. Specifically, section 7.4 of the Plan provides that the Plan Administrator, to be selected by the Creditors' Committee, will serve as the sole officer and director or manager, as applicable, of each of the Reorganized Debtors. Section 7.5 of the Plan provides great detail regarding the Plan Administrator as well as the Plan Administrator's rights, powers and duties. These provisions are consistent with the interests of the creditors and interest holders and with public policy. Thus, section 1123(a)(7) of the Bankruptcy Code is satisfied.
- 8. <u>Discretionary Contents of the Plan (11 U.S.C. § 1123(b))</u>. The Plan's provisions are appropriate and consistent with the applicable provisions of the Bankruptcy Code, including, without limitation, provisions for (a) Distributions to Holders of Claims and Interests, (b) the disposition of Executory Contracts and Unexpired Leases, (c) the retention of, and right to enforce, sue on, settle or compromise (or refuse to do any of the foregoing with respect to) certain claims or causes of action against third parties, to the extent not waived and released under the Plan, (e) indemnification obligations, (f) releases by the Debtors and Debtors-in-Possession and (g) releases by Holders of Claims and Interests.

- 9. <u>Fed. R. Bankr. P. 3016(a)</u>. The Plan is dated and identifies the entities submitting it, thereby satisfying Fed. R. Bankr. P. 3016(a).
- O. <u>Debtors' Compliance with Bankruptcy Code (11 U.S.C. § 1129(a)(2))</u>. The Debtors have complied with the applicable provisions of the Bankruptcy Code, thereby satisfying section 1129(a)(2) of the Bankruptcy Code. Specifically, the Debtors are proper debtors under section 109 of the Bankruptcy Code the Debtors and the Creditors' Committee are and proper proponents of the Plan under section 1121(a) of the Bankruptcy Code. The Debtors have complied with the applicable provisions of the Bankruptcy Code, including as provided or permitted by orders of the Court. The Debtors complied with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, the Local Rules for the United States Bankruptcy Court for the District of Delaware, and the Solicitation Procedures Order in transmitting the Plan, the Disclosure Statement, the Ballots and related documents and notices, and in soliciting and tabulating votes on the Plan.
- P. Plan Proposed In Good Faith (11 U.S.C. § 1129(a)(3)). The Debtors have proposed the Plan in good faith and not by any means forbidden by law, thereby satisfying section 1129(a)(3) of the Bankruptcy Code. In determining that the Plan has been proposed in good faith, the Court has examined the totality of the circumstances surrounding the filing of the Chapter 11 Cases and the formulation of the Plan. The Plan was proposed with the legitimate and honest purposes of maximizing the recovery to Holders of Claims and Interests under the circumstances of these Chapter 11 Cases.
- Q. Payments for Services or Costs and Expenses (11 U.S.C. § 1129(a)(4)). The Court-appointed professionals in these Chapter 11 Cases are subject to the requirements of sections 330 and 331 of the Bankruptcy Code and, therefore, have been approved by, or are subject to approval of the Court as reasonable. Further, the Plan provides that all unpaid

Professional Fees incurred prior to the Effective Date shall be subject to final allowance or disallowance upon application to the Bankruptcy Court pursuant to sections 330 or 503(b)(4) of the Bankruptcy Code.

- R. <u>Directors, Officers, and Insiders (11 U.S.C. § 1129(a)(5))</u>. The Debtors have complied with section 1129(a)(5) of the Bankruptcy Code and have disclosed the initial officers of the Reorganized Debtors. The Plan Administrator shall serve as the sole officer and director or manager, as applicable, of each of the Reorganized Debtors. The Plan Administrator shall be selected by the Creditors' Committee. Through the Plan Supplement the Debtors have disclosed publicly the identity and any affiliation of the Plan Administrator, KDW Restructuring & Liquidation Services LLC ("KDW"). KDW is not an insider of the Debtors. The employment of KDW as the sole officer and director or manager, as applicable, of the Reorganized Debtors is consistent with the interests of Holders of Claims against and Interests in the Debtors and with public policy.
- S. No Rate Changes (11 U.S.C. § 1129(a)(6)). Section 1129(a)(6) of the Bankruptcy Code is satisfied because the Plan does not provide for any change in rates over which a governmental regulatory commission has jurisdiction.
- T. Best Interests Test (11 U.S.C. §1129(a)(7)). The Plan satisfies the requirements of section 1129(a)(7) of the Bankruptcy Code. Article XI of the Disclosure Statement and the other evidence proffered or adduced at the Confirmation Hearing (i) are persuasive and credible, (ii) have not been controverted by other evidence or have not been challenged, (iii) are based upon reasonable and sound assumptions; and (iv) establish that each Holder of an impaired Claim or Interest either has accepted the Plan or will receive or retain under the Plan, on account of such Claim or Interest, property of a value, as of the Effective Date, that is not less than the

amount that such Holder would receive or retain if the Debtors were liquidated under chapter 7 of the Bankruptcy Code on such date.

- through 10E, 2F, 6F, 7F, 10F, 1G through 4G, 6G through 8G, 10G, 11G, and 3H have voted to accept the Plan. Classes 8F, 5G and 9G voted to reject the Plan. Classes 1I through 11I (Interests) will neither receive nor retain any property under the Plan and, therefore, are deemed to have rejected the Plan (Classes 1I through 11I, together with Classes 8F, 5G and 9G, the "Rejecting Classes"). In addition, no Holder of a Claim in Classes 11E, 4F, 9F, and 11F timely submitted a Ballot. As discussed more fully below, the Debtors have met the "cramdown" requirements in section 1129(b) of the Bankruptcy Code with respect to the Non-Accepting Classes necessary to obtain Confirmation of the Plan.
- V. Treatment of Administrative Claims, Tax Priority Claims and Non-Tax Priority

  Claims (11 U.S.C. § 1129(a)(9)). The treatment of Administrative Claims and Non-Tax Priority

  Claims under the Plan satisfies the requirements of section 1129(a)(9)(A) and (B) of the

  Bankruptcy Code, and the treatment of Tax Priority Claims under the Plan satisfies the

  requirements of section 1129(a)(9)(C) of the Bankruptcy Code.
- W. Acceptance By Impaired Class (11 U.S.C. § 1129(a)(10)). Classes 2E and 4E through 10E, 2F, 6F, 7F, 10F, 1G through 4G, 6G through 8G, 10G, 11G, and 3H are each Impaired Classes of Claims or Interests that have voted to accept the Plan and, to the best of the

The Voting Agent received no timely submitted Ballots from Holders of Claims classified in Classes 11E, 4F, 9F and 11F. For purposes of Plan confirmation, the Debtors treat these abstaining classes as if they had voted to reject the Plan. Accordingly, the Debtors demonstrate in Section IV below that the "cramdown" requirements for confirmation are satisfied with respect to Classes 11E, 4F, 9F and 11F (together with the Rejecting Classes, the "Non-Accepting Classes").

Debtors' knowledge, do not contain "insiders" of any significant magnitude. Thus, section 1129(a)(10) of the Bankruptcy Code is satisfied.

- X. Feasibility (11 U.S.C. § 1129(a)(11)). The Plan satisfies section 1129(a)(11) of the Bankruptcy Code. The Plan is feasible because the proceeds the Debtors realized from the sale of substantially all of their assets will be sufficient to satisfy the Reorganized Debtors' very limited post-Effective Date obligations. Apart from the distribution obligations to Holders of Allowed Claims and Administrative Claims, the Reorganized Debtors will have few financial obligations. Thus, the Plan is feasible and confirmation is not likely to be followed by the need for further financial reorganization of the Reorganized Debtors.
- Y. Payment of Fees (11 U.S.C. § 1129(a)(12)). The Debtors have paid or, pursuant to Section 14.4 of the Plan, will pay by the Effective Date fees payable under 28 U.S.C. § 1930, thereby satisfying section 1129(a)(12) of the Bankruptcy Code.
- Z. Continuation of Retiree Benefits (11 U.S.C. § 1129(a)(13)). The Debtors have not obligated themselves to provide retiree benefits to any person. Accordingly, the requirements of section 1129(a)(13) of the Bankruptcy Code are satisfied.
- Classes. Pursuant to section 1129(b) of the Bankruptcy Code, the Plan may be confirmed notwithstanding the fact that the Non-Accepting Classes have not voted to accept the Plan. All of the requirements of section 1129(a) of the Bankruptcy Code other than section 1129(a)(8), with respect to such Classes, have been met. Holders of unsecured Claims in Non-Accepting Classes will receive treatment identical to that received by other unsecured Classes: a pro rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed. In addition, Classes 11 through 111, consisting of Interests in the Debtors, will

not receive any Distribution under the Plan and are therefore deemed to reject the Plan. No Holders of Claims or Interests junior to the Holders of such Interests will receive or retain any property under the Plan on account of such Interests, and, as evidenced by the uncontroverted valuations and estimates contained in the Disclosure Statement and put into evidence at the Confirmation Hearing, no Class of Claims or Interests senior to any such Class is receiving more than full payment on account of such Claims or Interests. Accordingly, as required by section 1129(b) of the Bankruptcy Code, the Plan is fair and equitable and does not discriminate unfairly.

- BB. Principal Purpose of Plan (11 U.S.C. § 1129(d)). The principal purpose of the Plan is not the avoidance of taxes or the avoidance of the application of section 5 of the Securities Act of 1933 (15 U.S.C. § 77e).
- ASA Debtors and the VSE Debtors, as set forth in Section 7.1 of the Plan, will (i) facilitate and maximize prompt distribution to creditors and (ii) permit creditors to avoid the harm that likely would result absent substantive consolidation and confirmation of the Plan embodying it. The Court further finds that the limited substantive consolidation will benefit all creditors of the ASA Debtors and the VSE Debtors and is in the best interests of the Debtors, their creditors, and the Estates. Based upon the acceptance of the Plan by creditors entitled to vote and the absence of any objection by a party in interest to the proposed substantive consolidation of the ASA Debtors and the VSE Debtors set forth in the Plan, substantive consolidation is consensual and, under the circumstances of these Chapter 11 Cases, is appropriate under the Bankruptcy Code and applicable precedent.
- DD. <u>Modifications to the Plan</u>. The modifications to the Plan described and/or set forth beginning on Exhibit A hereto constitute non-material or technical changes and/or changes

with respect to particular Claims or Interests by agreement with Holders of such Claims or Interests, and do not materially adversely affect or change the treatment of any Claims or Interests. Accordingly, pursuant to Bankruptcy Rule 3019, these modifications do not require additional disclosure under section 1125 of the Bankruptcy Code or re-solicitation of votes under section 1126 of the Bankruptcy Code, nor do they require that Holders of Claims or Interests be afforded an opportunity to change previously cast acceptances or rejections of the Plan.

EE. Good Faith Solicitation (11 U.S.C. § 1125(e)). The Debtors and their agents, representatives, attorneys, and advisors, and other Persons involved in the solicitation process have solicited votes on the Plan in good faith and in compliance with the applicable provisions of the Bankruptcy Code and the Solicitation Procedures Order and are entitled to the protections afforded by section 1125(e) of the Bankruptcy Code and the exculpation provisions set forth in Section 12.5 of the Plan.

FF. Executory Contracts. The Debtors have exercised reasonable business judgment in determining that they will reject all of the prepetition executory contracts and unexpired leases to which any Debtor is a party, unless such contract or lease (a) previously shall have been assumed or rejected by the Debtors, (b) previously shall have expired or terminated pursuant to its own terms before the Effective Date or (c) is identified in the Plan Supplement as a contract to lease to be assumed. The Debtors have also exercised reasonable business judgment in determining that they will assume their director and officer liability insurance policies (the "Assumed Contracts") pursuant to § 365(a) of the Bankruptcy Code as set forth in Article IX of the Plan. There is no default in any of the Assumed Contracts. The assumption of the Assumed Contracts pursuant to section 9.2 of the Plan shall be legal, valid and binding upon the Debtors or Reorganized Debtors and all non-debtor parties to such executory contract or unexpired lease, all

to the same extent as if such assumption had been effectuated pursuant to an appropriate authorizing order of the Court entered before the Confirmation Date under section 365 of the Bankruptcy Code.

- GG. Rule 9019 Settlement and Releases. Pursuant to Fed. R. Bankr. P. 9019 and in consideration for the distributions and other benefits provided under, described in, contemplated by and/or implemented by the Plan, the releases described in Article XII of the Plan and the Intercompany Claims Settlement described in Section 7.12 of the Plan constitute integral elements of good faith compromises and settlements of the matters covered thereby. Such compromises and settlements are made in exchange for consideration and are in the best interests of Holders of Claims, are within the reasonable range of possible litigation outcomes, are fair, equitable and reasonable and are essential elements of the resolution of the Chapter 11 Cases in accordance with the Plan.
- HH. Approval of the Intercompany Claims Settlement. The Intercompany Claims
   Settlement, set forth in Section 7.12 of the Plan, is hereby approved in its entirety.
- II. Releases and Exculpation. Each of the release, indemnification and exculpation provisions set forth in the Plan: (1) is within the jurisdiction of the Court under 28 U.S.C. §§ 1334(a), (b), and (d); (2) is an essential means of implementing the Plan pursuant to section 1123(a)(5) of the Bankruptcy Code; (3) is an integral element of the transactions incorporated into the Plan; (4) confers a material benefit on, and is in the best interest of, the Debtors, their estates and their creditors; (5) is important to the overall objectives of the Plan to finally resolve all Claims among or against the parties-in-interest in the Chapter 11 Cases with respect to the Debtors, their organization, capitalization, operation and reorganization; and (6) is consistent with sections 105, 1123, 1129 and other applicable provisions of the Bankruptcy Code.

- JJ. <u>Conditions to Confirmation</u>. The conditions to Confirmation set forth in Section 11.1 of the Plan have been satisfied, waived or will be satisfied by entry of this Confirmation Order.
- KK. Conditions to Effective Date. Each of the conditions to the Effective Date, as set forth in Sections 11.2 of the Plan, is reasonably likely to be satisfied. The conditions to the Effective Date, set forth in Section 11.2 of the Plan, shall be subject to waiver by the Debtors (subject to the limitations set forth in Section 11.2 of the Plan), with the consent of the Creditors' Committee, which consent shall not be unreasonably withheld, without notice or a hearing, and as otherwise provided in Section 11.3 of the Plan.
- LL. Retention of Jurisdiction. The Court properly may retain jurisdiction over the matters set forth in Article XIII of the Plan.
- MM. Agreements and Other Documents. The Debtors have made adequate and sufficient disclosure of: (1) the adoption of new or amended and restated certificate of incorporation and bylaws or similar constituent documents for the Debtors or the Reorganized Debtors; (2) the Distributions to be made pursuant to the Plan; (3) the adoption, execution, delivery and implementation of all contracts, leases, instruments, releases and other agreements or documents related to any of the foregoing; and (4) the other matters provided for under the Plan involving the corporate structure of the Debtors or the Reorganized Debtors.
- NN. <u>Preservation of Causes of Action</u>. It is in the best interests of the creditors and interest holders that the Litigation Claims be retained by the Reorganized Debtors pursuant to Section 7.7 of the Plan to maximize the value of the Debtors' Estates.
- OO. <u>Election Pursuant to 11 U.S.C. § 1111(b)</u>. No secured creditor has elected the treatment provided by section 1111(b) of the Bankruptcy Code.

## ACCORDINGLY, THE COURT HEREBY ORDERS THAT:

- 1. Confirmation. The Plan, which consists of the Plan (and all exhibits and supplements thereto) and the modifications set forth in Exhibit A hereto, which are hereby incorporated into and constitute a part of the Plan, is hereby approved and confirmed under section 1129 of the Bankruptcy Code. The exhibits to the Plan, the Plan Schedules and the Plan Supplement (as may be modified pursuant to the terms of the Plan and/or such exhibit, as applicable) are incorporated by reference into and comprise an integral part of the Plan and this Confirmation Order.
- 2. <u>Objections</u>. All objections to confirmation of the Plan that have not been withdrawn, waived, settled, or addressed in the Plan or in this Confirmation Order and all reservations of rights included therein are overruled on the merits.
- 3. Provisions of Plan and Order Non-severable and Mutually Dependent.

  The provisions of the Plan and this Confirmation Order, including the findings of fact and conclusions of law set forth herein, are non-severable and mutually dependent. This Confirmation Order constitutes a judicial determination that each term and provision of the Plan, as it may have been altered, modified or interpreted at the Confirmation Hearing or the confirmation of the Plan, is valid and enforceable pursuant to its terms.
- 4. <u>Plan Classification Controlling</u>. The classification of Claims and Interests for purposes of the Distributions to be made under the Plan shall be governed solely by the terms of the Plan. The classifications set forth on the Ballots tendered to or returned by the Debtors' creditors and Interest Holders in connection with voting on the Plan (a) were set forth on the Ballots solely for purposes of voting to accept or reject the Plan, (b) do not necessarily represent, and in no event shall be deemed to modify or otherwise affect, the actual classification of such

Claims or Interests under the Plan for Distribution purposes, (c) may not be relied upon by any creditor or Interest Holder as representing the actual classification of such Claims or Interests under the Plan for Distribution purposes, and (d) shall not be binding on the Reorganized Debtors, the Estates, or the Debtors.

- 5. Effects of Confirmation; Successors and Assigns. Subject to the provisions of Sections 12.1 and 12.2 of the Plan, and notwithstanding any otherwise applicable law, upon the Effective Date, the terms of the Plan (including the Plan Exhibits, Plan Supplements, Plan Schedules, and all documents and agreements executed pursuant to the Plan) and this Confirmation Order are deemed binding upon (a) the Debtors, (b) the Reorganized Debtors, (c) all present and former Holders of Claims against and Interests in the Debtors, whether or not Impaired under the Plan and whether or not, if Impaired, such Holders accepted the Plan, (d) each Person acquiring property under the Plan, (e) any other party-in-interest, (f) any Person making an appearance in these Chapter 11 Cases, and (g) each of the foregoing's respective heirs, successors, assigns, trustees, executors, administrators, affiliates, officers, directors, agents, representatives, attorneys, beneficiaries, or guardians.
- 6. <u>Intercompany Claims and the Equity Interests in Affiliated Debtors</u>. The treatment of Intercompany Claims provided in Section 7.12 of the Plan is deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety.
- 7. Continued Corporate Existence; Revesting of Assets. Except as otherwise provided in the Plan, each Reorganized Debtor shall continue to exist after the Effective Date as a separate corporate or other legal entity, with all the powers of a corporation or legal entity under applicable law in the jurisdiction in which each applicable Debtor is incorporated or organized and pursuant to the respective certificate of incorporation and bylaws or other

organizational documents in effect prior to the Effective Date, except to the extent such certificate of incorporation and bylaws or other organizational documents are amended by the Plan. Except as otherwise explicitly provided in the Plan or in this Confirmation Order, including, without limitation, Section 7.6 of the Plan, on the Effective Date, all property of each Debtor's Estate shall revest in each applicable Reorganized Debtor on the Effective Date.

- 8. <u>Directors and Officers of Reorganized Debtors.</u> Pursuant to section 7.4 of the Plan, the Plan Administrator shall serve as the sole officer and director or manager, as applicable, of each of the Reorganized Debtors. This Court approves the appointment of the Plan Administrator as disclosed in the Plan Supplement or as otherwise announced at the Confirmation Hearing, as of and immediately following the Effective Date. As set forth in Section 7.8 of the Plan, the Plan Administrator shall be authorized to execute, deliver, file or record such documents, instruments, releases and other agreements and to take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.
- 9. Cancellation of Existing Securities. Except as otherwise provided in the Plan, on the Effective Date and concurrently with the applicable Distributions made pursuant to Article VII of the Plan, the promissory notes, share certificates (including treasury stock), the Unsecured Notes, the Unsecured Note Indenture, other instruments evidencing any Claims or Interests, and all options, warrants, calls, rights, puts, awards, commitments or any other agreements of any character to acquire such Interests shall be deemed automatically extinguished, canceled and of no further force and effect, without any further act or action under any applicable agreement, law, regulation, order or rule, and the obligations of the Debtors under the notes, share certificates, Unsecured Notes, the Unsecured Note Indenture and other agreements

and instruments governing such Claims and Interests shall be automatically discharged. The Holders of or parties to such canceled notes, share certificates, Unsecured Notes, the Unsecured Note Indenture and other agreements and instruments shall have no rights arising from or relating to such notes, share certificates, Unsecured Notes, the Unsecured Note Indenture, and other agreements and instruments or the cancellation thereof, except the rights provided pursuant to the Plan; provided, however, that the Unsecured Notes and the Unsecured Note Indenture shall continue in effect solely for the purposes of (i) allowing the Unsecured Note Indenture Trustee or its agents to make Distributions to Holders of Unsecured Notes; (ii) allowing Holders of the Unsecured Notes to receive Distributions hereunder; and (iii) preserving the rights and liens of the Unsecured Note Indenture Trustee with respect to the Unsecured Note Indenture Trustee Fees and Expenses to the extent not otherwise paid. The Unsecured Note Indenture shall terminate completely upon the completion of all Distributions to the Holders of the Unsecured Notes and the payment in full of the Unsecured Note Indenture Trustee Fees and Expenses. Notwithstanding the foregoing, the indenture governing the Prepetition Makewhole Claims and any related documents thereto governing the rights and obligations of the Debtors and non-Debtor parties with respect to the Prepetition Makewhole Claims shall survive for purposes of the Prepetition Makewhole Claim litigation, until such time as the Prepetition Makewhole Claims have been Allowed by a Final Order and the obligations are satisfied in accordance with the terms of the Plan or the Prepetition Makewhole Claim have been disallowed by Final Order or expunged. After the performance by the Unsecured Note Indenture Trustee or its agents of all duties that are required under the Plan, the Confirmation Order and/or under the terms of the Unsecured Note Indenture, the Unsecured Note Indenture Trustee and its agents and advisors shall be relieved of, and released from, all obligations associated with the Unsecured Notes

arising under the Unsecured Note Indenture or under other applicable agreements or law, and the Unsecured Note Indenture Trustee shall be deemed fully released and discharged.

- Disclosure Statement and in the Memorandum, and based upon the evidence and arguments made, proffered, or adduced at the Confirmation Hearing and in the Dickey Declaration, the Estates of the ASA Debtors and the VSE Debtors, respectively, shall be substantively consolidated as set forth in Section 7.1 of the Plan, solely for the purposes of voting on the Plan and making Distributions to Holders of Claims under the Plan.
- 11. Releases, Limitations of Liability and Indemnification. The releases set forth in Sections 12.2 and 12.5 of the Plan, and the exculpation and limitation of liability provisions set forth in Section 12.4 of the Plan, are deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety.
- Persons who have held, hold or may hold Claims against or Interests in the Debtors are (i) permanently enjoined from taking any of the following actions against the Estate(s), the Plan Administrator, the Creditors' Committee (or any of its members from time to time), the Plan Committee (or any of its members from time to time), the Unsecured Note Indenture Trustee, or any of their property on account of any such Claims, Interests or Precluded Claims and (ii) preliminarily enjoined from taking any of the following actions against any of the Debtors, the Reorganized Debtors, the Plan Administrator, the Creditors' Committee (or any of its members from time to time), the Plan Committee (or any of its members from time to time) or their property on account of such Claims, Interests or Precluded Claims: (A) commencing or continuing, in any manner or in any place, any action or other proceeding; (B) enforcing,

attaching, collecting or recovering in any manner any judgment, award, decree or order; (C) creating, perfecting or enforcing any lien or encumbrance; (D) asserting a setoff, right of subrogation or recoupment of any kind against any debt, liability or obligation due to the Debtors; and (E) commencing or continuing, in any manner or in any place, any action that does not comply with or is inconsistent with the provisions of the Plan. By accepting Distributions pursuant to the Plan, each Holder of an Allowed Claim receiving any Distribution pursuant to the Plan will be deemed to have specifically consented to the injunctions set forth in Article XII of the Plan.

- 13. <u>Injunction</u>. The satisfaction, release, and discharge pursuant to Article XII of the Plan shall act as an injunction against any Person commencing or continuing any action, employment of process, or act to collect, offset, or recover any Claim, Interest or cause of action satisfied, released, or discharged under the Plan to the fullest extent authorized or provided by the Bankruptcy Code, including, without limitation, to the extent provided for or authorized by sections 524 and 1141 thereof.
- 14. Plan Supplement. There being no objections to any of the documents contained in the Plan Supplement and any amendments, modifications and supplements thereto and all documents and agreements introduced into evidence by the Debtors at the Confirmation Hearing (including all exhibits and attachments thereto and documents referred to therein), the execution, delivery and performance thereof by the Reorganized Debtors is authorized and approved, including, but not limited to, the Plan Administrator Agreement. Without need for further order or authorization of the Court, the Debtors or Reorganized Debtors are authorized and empowered to make any and all modifications to any and all documents included as part of the Plan Supplement that do not materially modify the terms of such documents and are

consistent with the Plan; provided, however, that such modifications are reasonably acceptable to the Creditors' Committee or the Plan Committee, as applicable.

- appointment of KDW as the Plan Administrator hereby is approved and the Debtors and Reorganized Debtors are authorized to execute the Plan Administrator Agreement and to take any action necessary or appropriate to implement, effectuate or consummate the plan Administrator Agreement. The Reorganized Debtors are authorized and empowered to make any and all modifications to the Plan Administrator Agreement that do not materially modify the terms of such agreement, provided that such modifications are reasonably acceptable to the Plan Committee.
- approvals and consents required, if any, by the laws, rules or regulations of any State or any other governmental authority with respect to the implementation or consummation of the Plan and any documents, instruments or agreements, and any amendments or modifications thereto, and any other acts referred to in or contemplated by the Plan, the Disclosure Statement and any documents, instruments or agreements, and any amendments or modifications thereto.
- 17. <u>Plan Committee</u>. Pursuant to Section 7.8 of the Plan, the Plan Committee shall consist of one three (3) members who shall be appointed by the Creditors' Committee, unless otherwise determined in the sole discretion of the Creditors' Committee. After the Effective Date, the Plan Committee shall have the rights, duties and obligations set forth in Section 7.8 of the Plan and the Plan Administrator Agreement, and will exist for the time period set forth in Section 7.8 of the Plan. If the Effective Date does not occur as provided in Article XI

of the Plan, the Plan Committee shall be null and void ab initio, and neither the Plan Committee nor its members or professionals shall have any rights under Section 7.8 of the Plan.

- 18. Term of Bankruptcy Injunction or Automatic Stay. The stay in effect in the Chapter 11 Cases pursuant to sections 105 or 362(a) of the Bankruptcy Code shall continue to be in effect until the Effective Date, and at that time shall be dissolved and of no further force or effect, subject to the injunction set forth in the preceding paragraphs and/or sections 524 and 1141 of the Bankruptcy Code; provided, however, that nothing herein shall bar the filing of financing documents or the taking of such other actions as are necessary to effectuate the transactions specifically contemplated by the Plan or by this Confirmation Order prior to the Effective Date.
- 19. Matters Relating to Implementation of the Plan; General Authorizations.

  The approvals and authorizations specifically set forth in this Confirmation Order are nonexclusive and are not intended to limit the authority of the Debtors or Reorganized Debtors or the Plan Administrator to take any and all actions necessary or appropriate to implement, effectuate and consummate any and all documents or transactions contemplated by the Plan or this Confirmation Order. In addition to the authority to execute and deliver, adopt, assign or amend, as the case may be, contracts, leases, instruments, releases and other agreements specifically granted in this Confirmation Order, the Debtors and Reorganized Debtors are authorized and empowered, without action of their respective stockholders or boards of directors, to take any and all such actions as the Plan Administrator may determine are necessary or appropriate to implement, effectuate and consummate any and all documents or transactions contemplated by the Plan or this Confirmation Order.

- 20. Exemption From Certain Taxes and Recording Fees. Pursuant to section 1146 of the Bankruptcy Code, (a) the issuance, transfer or exchange of notes or equity securities under the Plan; (b) the creation of any mortgage, deed of trust, lien, pledge or other security interest; (c) the making or assignment of any contract, lease or sublease; or (d) the making or delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with, the Plan, including, without limitation, any merger agreements, agreements of consolidation, restructuring, disposition, liquidation or dissolution, deeds, bills of sale, and transfers of tangible property, will not be subject to any stamp tax, recording tax, personal property tax, real estate transfer tax, sales or use tax or other similar tax. All state or local government officials or agents are hereby directed to forgo the collection of any such tax or governmental assessment and to accept for filing and recordation any of the foregoing instruments or other documents without the payment of any such tax or governmental assessment.
- 21. Rejected Contracts and Leases: Assumption of Directors and Officers

  Liability Insurance Policies. Pursuant to Article IX of the Plan and sections 365 and 1123(b) of
  the Bankruptcy Code, all executory contracts and unexpired leases of the Debtors shall be
  deemed rejected effective on and subject to the occurrence of the Effective Date of the Plan,
  unless such contract or lease (a) previously shall have been assumed or rejected by the Debtors,
  (b) previously shall have expired or terminated pursuant to its own terms before the Effective
  Date, (c) is identified in the Plan Supplement as a contract or lease to be assumed; or (d) is one
  of the Assumed Contracts. The Debtors' assumption of the Assumed Contracts pursuant the
  provisions of Article IX of the Plan is expressly approved.
- 22. <u>Bar Date for Rejection Damage Claims</u>. Pursuant to Section 9.3 of the Plan, and notwithstanding anything to the contrary, if the rejection of an executory contract or

unexpired lease pursuant to Section 9.1 of the Plan or paragraph 21 of this Order gives rise to a Claim by the other party or parties to such contract or lease, such Claim shall be forever barred and shall not be enforceable against the applicable Debtor or its Estate, the Reorganized Debtors, the Plan Administrator or their respective successors or properties unless a proof of Claim is Filed and served on the Reorganized Debtors and counsel for the Reorganized Debtors within thirty (30) days after service of a notice of entry of the Confirmation Order or such other date as prescribed by the this Court.

Professional Fee Claims, Substantial Contribution Claims and Final Fee 23. Applications. All Professionals and other entities requesting compensation or reimbursement of Fee Claims pursuant to section 327, 328, 330, 331 or 503(b) of the Bankruptcy Code for services rendered prior to the Effective Date shall File and serve on the Reorganized Debtors, counsel for the Plan Administrator and counsel for the Plan Committee, an application for final allowance of compensation and reimbursement of expenses no later than sixty (60) days after the Effective Date, unless otherwise ordered by this Court. Objections to applications of such Professionals or other entities for compensation or reimbursement of expenses must be Filed and served on the Reorganized Debtors, counsel for the Plan Administrator, counsel for the Plan Committee and the requesting Professional or other entity no later than thirty (30) days (or such longer period as may be allowed by order of this Court) after the date on which the applicable request for compensation or reimbursement was served. Except as otherwise provided in the Plan, Professionals shall be paid pursuant to sections 328, 330, or 331 of the Bankruptcy Code and prior orders of this Court for amounts earned through the Effective Date. After the Effective Date, the Plan Administrator shall, in the ordinary course of business and without the necessity for any approval by this Court, pay the reasonable fees and expenses, incurred after the Effective

Date, of the professional persons employed by the Debtors, the Reorganized Debtors, the Creditors' Committee and Plan Committee, and of the Fee Auditor, as the case may be, in connection with the implementation and consummation of the Plan, the claims reconciliation process, and any other matters as to which such professionals may be engaged. The fees and expenses of such professionals shall be paid within ten (10) calendar days after submission of a detailed invoice therefor. If the Plan Administrator disputes the reasonableness of any such invoice, the Plan Administrator shall timely pay the undisputed portion of such invoice, and the Plan Administrator or the affected professional may submit such dispute to this Court for a determination of the reasonableness of such invoice. Upon the Effective Date, any requirement that Professionals comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date shall terminate, and the Plan Administrator, in consultation with the Plan Committee, may employ any Professional in the ordinary course of business.

Administrative Claims (including requests under section 503(b) of the Bankruptcy Code by any entity for making a substantial contribution in the Chapter 11 Cases) shall be thirty (30) days after the Confirmation Date, except with respect to Fee Claims, which shall be subject to the provisions of paragraph 23 above. Holders of asserted Administrative Claims, except for Fee Claims, not paid prior to the Confirmation Date shall submit proofs of Claim on or before such Administrative Claims Bar Date or forever be barred from doing so. The notice of Confirmation to be delivered pursuant to Fed. R. Bankr. P. 3020(c) and 2002(f) shall set forth such date and constitute notice of this Administrative Claims Bar Date. The Reorganized Debtors (in consultation with the Plan Committee) shall have sixty (60) days from the Effective Date (or

such longer period as may be allowed by order of the Bankruptcy Court upon the request of the Reorganized Debtors or the Plan Committee) following the Administrative Claims Bar Date to review and object to such Administrative Claims before a hearing for determination of allowance of such Administrative Claims.

- 25. <u>Provisions Governing Distributions</u>. The provisions in Article VIII of the Plan governing Distributions contemplated in the Plan are deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety subject to any modifications otherwise provided for in this Confirmation Order.
- 26. Southern Minnesota Construction and Weerts Construction. Prior to the Distribution Date, pursuant to Section 10.3(b) of the Plan, the Plan Administrator shall fund the Administrative Claims Reserve with Cash of the VSE Debtors in an amount equal to \$84,000, representing asserted interest, costs and counsel fees related to Claims filed by Southern Minnesota Construction (Claim No. 2258) and Weerts Construction (Claim No. 2972). Entry of this Confirmation Order shall not alter the right of the Debtors, the Creditors' Committee, the Plan Committee or the Plan Administrator, as the case may be, to object, in whole or in part, on any grounds whatsoever, to any Claim filed in these Chapter 11 Cases, including, without limitation, Claims asserted by Southern Minnesota Construction and Weerts Construction for interest, costs and counsel fees.
- 27. Setoff. Pursuant to Section 8.10 of the Plan, the Plan Administrator may, pursuant to section 553 of the Bankruptcy Code or applicable nonbankruptcy laws, but shall not be required to, set off against any Claim and the payments or other Distributions to be made pursuant to the Plan in respect of such Claim, or claims of any nature whatsoever that the Debtors or Reorganized Debtors may have against the Holder of such Claim (except to the extent

that such claims have been waived by the Debtors in the Plan or otherwise); provided, however, that neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Reorganized Debtors of any such claim that the Debtors or Reorganized Debtors may have against such Holder. In no event shall any Holder of Claims or Interests be entitled to setoff any Claim or Interest against any Claim, right, or cause of action of the Reorganized Debtors unless such Holder has filed a motion (a "Setoff Motion") with the Bankruptcy Court requesting the authority to perform such setoff on or before the Voting Deadline, provided, however, that nothing in this Confirmation Order or in Section 8.10 of the Plan shall affect any party's right to seek to offset or net out amounts due and owing in accordance with sections 362(b)(6), (7), (17), (27), 555, 556, 560 and/or 561 of the Bankruptcy Code, irrespective of whether such a party filed a Setoff Motion. Notwithstanding any provision to the contrary in the Plan, this Confirmation Order or any implementing Plan documents, nothing shall affect the rights of the United States, including the Internal Revenue Service ("IRS") and the Alcohol, Tobacco and Trade Bureau ("ATTB"), to assert setoff and recoupment. The Debtors reserve all rights and defenses in connection with any and all assertions of alleged setoff or recoupment rights.

Setoff. Notwithstanding anything to the contrary in the Disclosure Statement, Plan, any other Plan document, or this Confirmation Order (collectively the "Plan Documents"), nothing in the Plan Documents (including any provision that purports to be preemptory or supervening), shall in any way operate to, or have the effect of, impairing in any respect the legal, equitable or contractual rights, including, but not limited to, rights of setoff and recoupment, and defenses of the Debtors or Liberty Mutual Insurance Company and its affiliates ("Liberty") with respect to

any insurance policies and any agreements related to the policies (collectively, the "Policies")<sup>5</sup> entered into by Liberty and the Debtors. The rights and obligations of Liberty and the Debtors shall be determined under the Policies and any applicable law. The Policies shall remain in full force and effect in accordance with their terms, including all terms, conditions, limitations and exclusions contained therein; provided, however, notwithstanding anything to the contrary in the Plan Documents, that any Claims for monetary relief against the Debtors or Reorganized Debtors arising out of the Policies shall only be asserted through (A) draws on available collateral or on any letter of credit. (B) prosecution and enforcement of General Unsecured Claims or Administrative Claims, as applicable, under the Plan, or (C) the exercise of rights of setoff or recoupment as permitted by the Policies or applicable law. Nothing in the Plan Documents shall modify or otherwise alter Liberty's rights in the letter of credit issued by Wells Fargo, and any other collateral that secures the Debtors' obligations to Liberty, including, but not limited to, rights to draw or apply such letter of credit or collateral. In addition, nothing in the Plan Documents shall be construed to limit, diminish or otherwise affect any rights that Liberty may have to participate in or control the defense, investigation or settlement of any claims that may be covered by any of the Policies. Nothing in the Plan Documents shall impair or modify the Debtors' obligations under the Policies and applicable law to cooperate in the defense, investigation or settlement of any claims that may be covered under the Policies.

29. Resolution of Liberty Objection. Prior to the Distribution Date, pursuant to Section 10.3(b) of the Plan, the Plan Administrator shall fund the Administrative Claims

Reserve with Cash of the VSE Debtors in an amount equal to \$400,000, representing all or a

The Policies include the following policy numbers and any agreements related to the policy numbers: TB1-641-436719-016; WC1-641-436719-026; TB1-641-436719-017; WC2-641-436719-027; TB1-L45-436719-018; WC2-L45-436719-028.

portion of Liberty's potential claims and interests, including rights and interests as subrogee, that have been preserved by Liberty under this Court's Orders entered as Docket Entry Nos. 949, 1002, 1004, and 1006. In addition, notwithstanding Section 14.3 of the Plan, the Plan Documents or anything in this Confirmation Order, Liberty shall have until, and the Administrative Claims Reserve shall remain in effect for, six (6) months from the Confirmation Date (or as otherwise agreed by Liberty and the VSE Debtors) to assert the liquidated amount of its claim to funds in the Administrative Claims Reserve. Entry of this Confirmation Order shall not alter the right of the Debtors, the Creditors' Committee, the Plan Committee or the Plan Administrator, as the case may be, to object, in whole or in part, on any grounds whatsoever, to any claim asserted by Liberty. Additionally, except to the extent specifically set forth in this paragraph, nothing in this Confirmation Order shall otherwise alter, modify or limit (i) the rights reserved by Liberty in prior orders entered by this Court, (ii) the rights, remedies and defenses which Liberty has or may have with respect to claims under its bonds, or (iii) the rights or claims of Liberty under its proofs of claim numbered 364, and 955 through 966 filed in these Chapter 11 Cases.

30. <u>FNBO Accounts</u>. Notwithstanding anything to the contrary in the Plan Documents or this Confirmation Order, (a) First National Bank of Omaha ("<u>FNBO</u>"), upon confirmation of the Plan, shall retain (i) all liens, pledges or other security interests in any and all of the Debtors' or Reorganized Debtors' FNBO Accounts<sup>6</sup>, (ii) all rights of setoff,

The "FNBO Accounts" include, but are not limited to, the following accounts with FNBO: Account No. 110118507 (ASA Albion, LLC); Account No. 110118497 (ASA Bloomingburg LLC); Account No. 110118484 (ASA Linden, LLC); Account No. 110197793 (VeraSun Aurora Corporation); Account No. 110211417 (VeraSun Charles City, LLC); Account No. 110193438 (VeraSun Energy Corporation); Account No. 201651766-20 (VeraSun Energy Corporation); Account No. 110118073 (VeraSun Fort Dodge, LLC); Account No. 110210845 (VeraSun Hartley, LLC); Account No. 110203784 (VeraSun Marketing, LLC); Account No. 110203690 (VeraSun Marketing, LLC); and Account No. 110203920 (VeraSun Welcome, LLC). For the avoidance of doubt, FNBO also maintains an account in its name, Account No. 110305433, from the proceeds (cont'd)

subrogation or recoupment of any kind altogether, and (iii) all claims related thereto, pursuant to terms and conditions of the deposit agreements with respect to the FNBO Accounts and that certain Promissory Note and Continuing Letter of Credit Agreement dated June 16, 2006 (collectively, the "FNBO Instruments"); (b) the Plan Documents shall not enjoin FNBO from asserting any right to enforce any claim, lien or encumbrance, or any right of setoff, subrogation or recoupment of any kind altogether, in accordance with the FNBO Instruments or applicable law; and (c) such FNBO Instruments shall survive confirmation of the Plan and the Effective Date and shall continue to govern the parties' relationships in connection with the FNBO Accounts and the outstanding letters of credit issued by FNBO on behalf of the Debtors. The Debtors reserve all rights and defenses in connection with any and all assertions of alleged setoff or recoupment rights.

Resolution of FNBO Objection. To resolve the Protective Objection and Reservation of Rights of First National Bank of Omaha to Joint Plan of Liquidation of VeraSun Energy Corporation and Its Affiliated Debtors Dated July 31, 2009 (Docket No. 1935) (the "FNBO Objection"), the Debtors and FNBO have agreed that the Debtors will pay to FNBO, by wire transfer on or before the Confirmation Hearing, the amount of \$350,767.21, which represents the undisputed fees and expenses due and owing to FNBO (the "Undisputed Fees"). With respect to the remaining balance of fees and expenses addressed by the FNBO Objection in the amount of \$50,851.42 (the "Disputed Fees"), to the extent the Debtors and FNBO are unable to resolve, all parties fully reserve all rights and defenses each may have in connection with the

(cont'd from previous page)

of that certain Irrevocable Standby Letter of Credit issued by UBS AG Stamford Branch, as collateral for certain cash-backed letters of credit.

Disputed Fees, and FNBO does not waive any claims to recover the Disputed Fees at any time, including after confirmation of the Plan.

- 32. Resolution of A&B Process Systems Corp. Objection. Notwithstanding anything to the contrary contained in the Plan Documents, the possessory lien, if any, of A&B Process Systems Corp. ("A&B") on any and all property of the Debtors or Reorganized Debtors that is in the possession of A&B (the "Alleged Collateral") shall continue following the Effective Date and shall attach to any proceeds of the Alleged Collateral. If authorized by this Court after proper notice and hearing, A&B may liquidate the Alleged Collateral in a commercially reasonable manner. Promptly following such liquidation, A&B shall provide the Debtors an accounting of all proceeds realized by A&B as a result of such liquidation and, after deducting the dollar amount of all costs incurred by A&B in connection with such liquidation, transfer the remaining proceeds (the "Net Proceeds") to the client trust account of A&B's counsel, Michael Best & Friedrich, LLP, pending further order of the court or written agreement of A&B and the Debtors or Reorganized Debtors. Entry of this Confirmation Order shall not alter the right of the Debtors, the Creditors' Committee, the Plan Committee or the Plan Administrator, as the case may be, to object, in whole or in part, on any grounds whatsoever, to any Claim filed in these Chapter 11 Cases, including, without limitation, any Claim asserted by A&B.
- Payment of Priority Tax Claims. Notwithstanding any provision to the contrary in the Plan, this Confirmation Order and Plan Documents, nothing shall: (1) affect the ability of the IRS and the ATTB to pursue, to the extent allowed by non-bankruptcy law, any non-debtors for any liabilities that may be related to any federal tax liabilities owed by the Debtors; and (2) to the extent that the Priority Tax Claims of the IRS and ATTB are not paid in full in cash on the Effective Date,

the IRS and ATTB allowed Priority Tax Claims shall be paid in equal quarterly installments commencing on the Effective Date and interest shall accrue on such claims from the Effective Date at the rate and method set forth in 26 U.S.C. Sections 6621 and 6622.

- Authorization to Consummate Plan. The Court authorizes the Debtors to consummate the Plan after entry of this Confirmation Order. The Debtors are authorized to execute, acknowledge, and deliver such deeds, assignments, conveyances, and other assurances, documents, instruments of transfer, uniform commercial code financing statements, trust agreements, mortgages, indentures, security agreements, and bills of sale and to take such other actions as may be reasonably necessary to perform the terms and provisions of the Plan, all transactions contemplated by the Plan, and all other agreements related thereto.
- the Plan Proponents revokes or withdraws the Plan as to any or all of the Debtors, or if consummation of the Plan as to any or all of the Debtors does not occur, then, with respect to such Debtors, (a) the Plan shall be null and void in all respects, (b) any settlement or compromise embodied in the Plan (including the fixing or limiting to an amount certain any Claim or Class of Claims), assumption or rejection of executory contracts or unexpired leases affected by the Plan, and any document or agreement executed pursuant to the Plan, shall be deemed null and void, and (c) nothing contained in the Plan shall (i) constitute a waiver or release of any Claims by or against, or any Interests in, such Debtors or any other Person, (ii) prejudice in any manner the rights of such Debtors or any other Person, or (iii) constitute an admission of any sort by the Debtors or any other Person.
- 36. Retention of Jurisdiction. Pursuant to sections 105(a) and 1142 of the Bankruptcy Code, and notwithstanding the entry of this Confirmation Order or the occurrence of

the Effective Date, the Court shall retain exclusive jurisdiction as provided in the Plan over all matters arising out of, and related to, the Chapter 11 Cases and the Plan to the fullest extent permitted by law, including, among other items and matters, jurisdiction over those items and matters set forth in Article XIII of the Plan.

- On the Effective Date, the Creditors' Committee. On the Effective Date, the Creditors' Committee shall dissolve automatically and its members shall be deemed released of all their duties, responsibilities and obligations in connection with the Chapter 11 Cases or the Plan and its implementation, and the retention or employment of the Creditors' Committee's attorneys, accountants and other agents shall terminate, except with respect to (i) all matters related to Fee Claims and (ii) any appeals of the Confirmation Order. All expenses of members of the Creditors' Committee and the fees and expenses of their professionals through the Effective Date shall be paid in accordance with the terms and conditions of the applicable orders of the Bankruptcy Court. Counsel to the Creditors' Committee shall be entitled to reasonable compensation and reimbursement of expenses for post-Effective Date activities authorized hereunder upon the submission of invoices to the Reorganized Debtors.
- 38. References to Plan Provisions. The failure to include or specifically reference any particular provision of the Plan in this Confirmation Order shall not diminish or impair the effectiveness of such provision, it being the intent of the Court that the Plan be confirmed in its entirety. The provisions of the Plan and of this Confirmation Order shall be construed in a manner consistent with each other so as to effect the purposes of each; provided, however, that if there is determined to be any inconsistency between any Plan provision and any provision of this Confirmation Order that cannot be so reconciled, then, solely to the extent of such inconsistency, the provisions of this Confirmation Order shall govern and any such

provision of this Confirmation Order shall be deemed a modification of the Plan and shall control and take precedence.

- 39. <u>Separate Confirmation Orders</u>. This Confirmation Order is and shall be deemed a separate Confirmation Order with respect to each of the Debtors in each Debtors' separate Chapter 11 Case for all purposes. The Clerk of the Court is directed to file and docket this Confirmation Order in the Chapter 11 Case of each of the Debtors.
- 40. Filing and Recording. This Confirmation Order is and shall be binding upon and shall govern the acts of all entities including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state and local officials, and all other persons and entities who may be required, by operation of law, the duties of their office, or contract, to accept, file, register or otherwise record or release any document or instruments. Each and every federal, state and local government agency is hereby directed to accept any and all documents and instruments necessary, useful or appropriate (including Uniform Commercial Code financing statements) to effectuate, implement and consummate the transactions contemplated by the Plan and this Confirmation Order without payment of any recording tax, stamp tax, transfer tax or similar tax imposed by state or local law.
- Date. The Effective Date shall occur after the conditions precedent to the occurrence of the Effective Date, which are set forth in Section 11.2 of the Plan, have all taken place, or upon such later date as may be proposed by the Plan Proponents and consistent with the Plan and the exhibits thereto. On or before the fifth (5th) Business Day following the occurrence of the Effective Date, the Debtors shall serve notice of this Confirmation Order and occurrence of the

Effective Date pursuant to Bankruptcy Rules 2002(f)(7), 2002(k), and 3020(c), on all Holders of Claims, the United States Trustee and other parties-in-interest, by causing a notice of this Confirmation Order and the occurrence of the Effective Date in substantially the form of the notice annexed hereto as Exhibit C, which form is hereby approved (the "Notice of Effective Date"), to be delivered to such parties by first class mail, postage prepaid; provided, however, that notice need not be given or served under the Bankruptcy Code, the Bankruptcy Rules, or this Confirmation Order to any Person to whom the Debtors mailed the Confirmation Hearing Notice, but received such notice returned marked "undeliverable as addressed," "moved - left no forwarding address," "forwarding order expired," or similar reason, unless the Debtors have been informed in writing by such Person of that Person's new address. The Reorganized Debtors are authorized to send the Notice of Effective Date to parties-in-interest, rather than a copy of the entered Confirmation Order provided that copies of said order are made available to requesting parties at their own expense. The notice described herein is adequate under the particular circumstances of the Chapter 11 Cases, and no other or further notice is necessary. Notwithstanding the foregoing, pursuant to Bankruptcy Rule 2002(I), the Debtors may satisfy the requirements of Bankruptcy Rule 2002(f)(7), by mailing the Notice of Effective Date as described above and publishing the Notice of Effective Date in the Wall Street Journal (national edition) and the Argus Leader within fifteen (15) Business Days of the Effective Date.

42. Exhibits to the Plan Will Operate as Controlling Documents. In the event of an inconsistency between the Plan and the Exhibits to the Plan (as may be modified), the Exhibits to the Plan will control; provided, however, that any discrepancies between the Plan, Exhibits to the Plan and this Confirmation Order shall be controlled by the Confirmation Order.

- Aid and Recognition of Foreign Courts. This Court hereby requests the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in the United States or any other jurisdiction, to give effect to this Order and to assist the Debtors and their respective agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Debtors as may be necessary or desirable to give effect to this Order, or to assist the Debtors and their respective agents in carrying out the terms of this Order.
- 44. 28 U.S.C. § 157(d). Nothing in this Confirmation Order or the Plan is intended to modify or violate 28 U.S.C. § 157(d).
- 45. <u>Subordination of Claims</u>. For the avoidance of doubt, Old Equity shall also include any Claim subordinated pursuant to section 510(b) arising from the rescission of a purchase or sale of Old Equity or rights relating to such Old Equity, or any Claim for damages arising from the purchase or sale of Old Equity or any Claim for reimbursement, contribution, or indemnification arising from or relating to any such claims.

46. <u>Modifications to the Plan</u>. At the request of the Debtors, the Plan is hereby modified pursuant to section 1127(a) of the Bankruptcy Code as set forth on <u>Exhibit A</u> hereto.

Dated: Wilmington, Delaware Octobes 2009

The Honorable Brendan L. Shannon United States Bankruptcy Judge

## Exhibit A

Blackline Showing Modifications to Plan

## IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:

) Case No. 08-12606 (BLS)

VERASUN ENERGY CORPORATION, et al.,

Chapter 11

Debtors.

Jointly Administered

JOINT PLAN OF LIQUIDATION OF VERASUN ENERGY CORPORATION AND ITS AFFILIATED DEBTORS Mark S. Chehi (I.D. No. 2855) Megan E. Cleghorn (I.D. No. 4080) Davis Lee Wright (I.D. No. 4324) Skadden, Arps, Slate, Meagher & Flom LLP

One Rodney Square

P.O. Box 636

Wilmington, DE 19899

(302) 651-3000

Dennis A. Meloro (No. 4435) Donald J. Detweiler (No. 3087) Sandra G. M. Selzer (No. 4283) Greenberg Traurig, LLP The Nemours Building 1007 North Orange Street, Suite 1200 Wilmington, DE 19801 (301) 661-7000

— and —

— and —

Felicia Gerber Perlman John K. Lyons Skadden, Arps, Slate, Meagher & Flom LLP 155 North Wacker Drive Chicago, Illinois 60606 (312) 407-0700

Attorneys for VeraSun Energy Corporation, et al., Debtors and Debtors-in-Possession

Michael S. Stamer David H. Botter Alexis Freeman Akin Gump Strauss Hauer & Feld LLP One Bryant Park New York, New York 10036 (212) 872-1000

Attorneys for the Official Committee of Unsecured Creditors of VeraSun Energy Corporation, et al.

Dated: Wilmington, Delaware July 31, 2009

## TABLE OF CONTENTS

INTRO	INTRODUCTION1		
	ARTICLE I		
	DEFINED TERMS AND RULES OF INTERPRETATION		
1.1	Administrative Claim	2	
1.2	Administrative Claims Reserve	2	
1.3	Administrator Professionals	2	
1.4	AgStar Litigation	2	
1.5	AgStar Settlement	2	
1.6	Affiliate Debtors	2	
1.7	Allowed Claim	2	
1.8	Allowed Claim		
1.9	ASA Debtors	3	
1.10	Available Cash	3	
1.11	Ballot	3	
1.12	Bankruptcy Code	3	
1.13	Bankruptcy Court	3	
1.14	Bankruptcy Rules	3	
1.15	Bar Date	3	
1.16	Bar Date Order	4	
1.17	Business Day	4	
1.18	Cash	4	
1.19	Chapter 11 Case(s)		
1.20	Claim(s)		
1.21	Claims Objection Deadline		
1.22	Class	4	
1.23	Collateral		
1.24	Confirmation Date		
1.25	Confirmation Hearing		
1.26	Confirmation Order		
1.27	Creditors: Committee		
1,28	Debtor(s)	5	
1.29	Deficiency Claim	4	
1.30	Disallowed Claim.		
1.31	Disbursing Agent		
1.32	Disclosure Statement	4	
1.33	Disputed Claim		
1.34	Disputed Claim		

i DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

1.35	Disputed Claim Amount	.6
1.36	Disputed Claims Reserve	.6
1.37	Distribution	.6
1.38	Distribution Date	.6
1.39	Distribution Record Date	.6
1.40	Effective Date	.6
1.41	Estate(s)	.6
1.42	Face Amount	ه.
1.43	Fee Auditor	.7
1.44	Fee Claim	.7
1.45	File, Filed or Filing	
1.46	Final Order	
1.47	General Unsecured Claim	
1.48	Guaranty Claim	.7
1.49	Holder	
1.50	Impaired	
1.51	Intercompany Claim	
1.52	Intercompany Claims Settlement	.8
1.53	Interest	.8
1.54	Internal Revenue Code	.8
1.55	Lender Deficiency Claim	.8
1.56	Litigation Claims	.8
1.57	Litigation Reserve	.8
1.58	Non-Tax Priority Claim	8
1.59	Non-VSE Debtors	8
1.60	Old Common Stock	
1.61	Old Equity	.9
1.62	Operating Reserve	9
1.63	Other Deficiency Claim	.9
1.64	Other Secured Claim	9
1.65	Person	
1.66	Petition Date	9
1.67	Plan	9
1.68	Plan Administrator	10
1.69	Plan Administrator Agreement	
1.70	Plan Committee	10
1.71	Plan Proponents	10
1.72	Plan Supplement	10
1.73	Precluded Claims	10
1.74	Prepetition Agstar Credit Agreements	
1.75	Prepetition Albert City Credit Agreement	11
1.76	Prepetition Central City Credit Agreement	11
1.77	Prepetition Credit Agreements	11

ii DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

1.78	Prepetition Dougherty/Marion Credit Agreements			
1.79	Prepetition Dyersville Credit Agreement	11		
1.80	Prepetition First Bank Letter of Credit	11		
1.81	Prepetition Hankinson Credit Agreement			
1.82	Prepetition Janesville Credit Agreement			
1.83	Prepetition Ord Credit Agreement	11		
1.84	Prepetition Makewhole Claim12			
1.85	Prepetition Makewhole Claims Reserve			
1.86	Prepetition Secured Lender Claim(s)	12		
1.87	Prepetition WestLB/ASA Credit Agreement	12		
1.88	Prepetition Woodbury Credit Agreement	12		
1.89	Priority Tax Claim	12		
1.90	Professional			
1.91	Pro Rata	12		
1.92	Quarter			
1.93	Quarterly Distribution Date			
1.94	Reorganized			
1.95	Reserves			
1.96	Restricted Cash			
1.97	Scheduled			
1.98	Schedules			
1.99	Solicitation Order			
1.100	Subsidiary Interests			
1.101	Substantive Consolidation Order	13		
1.102	Unclaimed Distribution Reserve			
1.103	Unimpaired			
1.104	Unsecured Note Claim.			
1.105	Unsecured Note Indenture	14		
1.106	Unsecured Note Indenture Trustee			
1.107	Unsecured Note Indenture Trustee Fees and Expenses			
1.108	Unsecured Notes			
1.109	USBio Debtors			
1.110				
	VEC			
	VSE Debtors			
	VSE Secured Notes			
	Voting Deadline			
1.114	Voting Deadinic	10		
	ARTICLE II			
	ADMINISTRATIVE EXPENSES AND PRIORITY TAX CLAIMS			
2.1	Administrative Claims	16		

iii DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

2.2	Priority Tax Claims	16
	ARTICLE III	
	CLASSIFICATION OF CLAIMS AND INTERESTS	
3.1 3.2	The DebtorsClassification of Claims and Interests	17 18
	ARTICLE IV	
	IDENTIFICATION OF CLASSES OF CLAIMS AND INTERESTS IMPAIRED AND UNIMPAIRED BY THE PLAN	
4.1 4.2	Classes of Claims that are Unimpaired Impaired Classes of Claims and Interests	19 19
	ARTICLE V	
	PROVISIONS FOR TREATMENT OF CLAIMS AND INTERESTS	
5.1 5.2 5.3	Unimpaired Classes of Claims Impaired Classes of Claims Impaired Classes of Interests	22
	ARTICLE VI	
	ACCEPTANCE OR REJECTION OF THE PLAN	
6.1 6.2 6.3 6.4 6.5 6.6 6.7	Impaired Classes of Claims Entitled to Vote  Classes Deemed to Accept the Plan  Acceptance by Impaired Classes  Classes Deemed to Reject the Plan  Elimination of Classes  Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code  Confirmability and Severability of a Plan  ARTICLE VII	24 25 25
	MEANS FOR IMPLEMENTATION OF THE PLAN	
- 1		0.5
7.1 7.2 7.3	Substantive Consolidation	26
iv Per:	DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634 formed on 10/20/2009.	/26.

7.4	Directors, Managers, and Officers; Effectuating Documents; Further Transactions	27
7.5	The Plan Administrator	27
7.6	Revesting of Assets	
7.7	Preservation of Rights of Action; Settlement of Litigation Claims	
7.8	Creditors' Committee and Plan Committee	32
7.9	Cancellation of Securities, Instruments and Agreements Evidencing Claims and	
	Interests	34
7.10	Sources of Cash for Plan Distributions	35
7.11	Exemption from Transfer Taxes	35
7.12	Settlement of Certain Intercompany Claims	35
7.13	Agstar Settlement	36
	ARTICLE VIII	
	PROVISIONS GOVERNING DISTRIBUTIONS	
8.1	Distributions for Claims Allowed as of the Effective Date	
8.2	Interest on Claims	
8.3	Distributions by the Reorganized Debtors	
8.4	Delivery of Distributions and Undeliverable or Unclaimed Distributions	
8.5	Record Date for Distributions	
8.6	Unsecured Note Indenture Trustee as Claim Holder	
8.7	Allocation of Plan Distributions Between Principal and Interest	39
8.8	Means of Cash Payment	39
8.9	Withholding and Reporting Requirements	39
8.10	Setoffs	
8.11	Fractional Dollars; De Minimis Distributions	
8.12	Release of Liens	40
	ARTICLE IX	
	TREATMENT OF EXECUTORY CONTRACTS	
	AND UNEXPIRED LEASES	
9.1	Rejected Contracts and Leases	40
9.2	Assumption of Director and Officer Liability Insurance Policies	40
9.3	Bar to Rejection Damages	

## ARTICLE X

# PROCEDURES FOR RESOLVING DISPUTED, CONTINGENT AND UNLIQUIDATED CLAIMS

10.1	Objection Deadline; Prosecution of Objections	41
10.2	No Distributions Pending Allowance	41
10.3	Accounts; Reserves	
10.4	Distributions After Allowance	43
	ARTICLE XI	
	CONFIRMATION AND CONSUMMATION OF THE PLAN	
11.1	Conditions to Confirmation	43
11.2	Conditions to Effective Date	
11.3	Waiver of Conditions Precedent	
11.4	Consequences of Non-Occurrence of Effective Date	44
	ARTICLE XII	
	EFFECT OF PLAN CONFIRMATION	
12.1	Binding Effect	45
12.2	Releases	
12.3	Discharge of Claims and Termination of Interests	46
12.4	Exculpation and Limitation of Liability	46
12,5	Release by Holders of Claims	
12.6	Injunction	
12.7	Term of Bankruptcy Injunction or Stays	
	ARTICLE XIII	
	RETENTION OF JURISDICTION	
13.1	Retention of Jurisdiction	48
	ARTICLE XIV	
	MISCELLANEOUS PROVISIONS	
14.1	Effectuating Documents and Further Transactions	50
14.2	Corporate Action	
14.3	Bar Dates for Certain Claims	

vi DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

14.4	Payment of Statutory Fees	52
	Amendment or Modification of the Plan	
	Severability of Plan Provisions	
	Successors and Assigns	
	Plan Supplement	
	Revocation. Withdrawal or Non-Consummation	
14.10	Notice	54
	Governing Law	
	Schedules	
	Filing of Additional Documents	

#### INTRODUCTION

VeraSun Energy Corporation, ASA OpCo Holdings, LLC, US BioEnergy Corporation, VeraSun Albert City, LLC, VeraSun Albion, LLC, VeraSun Aurora Corporation, VeraSun BioDiesel, LLC, VeraSun Bloomingburg, LLC, VeraSun Central City, LLC, VeraSun Charles City, LLC, VeraSun Dyersville, LLC, VeraSun Fort Dodge, LLC, VeraSun Granite City, LLC, VeraSun Hankinson, LLC, VeraSun Hartley, LLC, VeraSun Janesville, LLC, VeraSun Linden, LLC, VeraSun Litchfield, LLC, VeraSun Marion, LLC, VeraSun Marketing, LLC, VeraSun Ord, LLC, VeraSun Reynolds, LLC, VeraSun Tilton, LLC, VeraSun Welcome, LLC, and VeraSun Woodbury, LLC (collectively, the "Debtors") debtors and debtors-in-possession in the above-captioned jointly administered chapter 11 reorganization cases, and the Official Committee of Unsecured Creditors appointed in the above-captioned jointly administered chapter 11 reorganization cases pursuant to section 1102 of the Bankruptcy Code (the "Creditors' Committee"), hereby propose the following joint plan of liquidation (the "Plan") for the resolution of the outstanding claims against and interests in the Debtors pursuant to section 1121(a) of the Bankruptcy Code. Reference is made to the Disclosure Statement (as that term is defined herein), distributed contemporaneously herewith, for a discussion of the Debtors' history, business, properties and operations, a summary and analysis of the Plan, and certain related matters including, among other things, the proposed substantive consolidation of certain of the Debtors' cases. Subject to certain restrictions and requirements set forth in section 1127 of the Bankruptcy Code and Fed, R. Bankr, P. 3019, each of (I) the Creditors' Committee and (II) the Debtors reserve the right to alter, amend, modify, revoke or withdraw the Plan (subject to the consent of the Creditors!' Committee or the Debtors, as applicable, which consent shall not be unreasonably withheld) prior to the Effective Date (as that term is defined herein).

As set forth herein, the Plan contemplates the substantive consolidation of the VSE Debtors as well as the substantive consolidation of the ASA Debtors. The Plan does not contemplate the substantive consolidation of the USBio Debtors, but rather constitutes an individual Plan with respect to each of the USBio Debtors. Each Debtor and the Creditors' Committee is a proponent of the Plan contained herein within the meaning of section 1129 of the Bankruptcy Code. For voting and Distribution purposes, the Plan contemplates separate classes for the VSE Debtors, the ASA Debtors and each of the USBio Debtors.

#### ARTICLE I

#### DEFINED TERMS AND RULES OF INTERPRETATION

As used herein, capitalized terms shall have the meanings set forth below. Any term that is not otherwise defined herein, but that is used in the Bankruptcy Code or the

Bankruptcy Rules, shall have the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable.

- administration of the Chapter 11 Cases Allowed under section 503(b) or 507(a)(2) of the Bankruptcy Code, including: (a) any actual and necessary costs and expenses incurred after the Petition Date of preserving the Debtors' Estates and operating the businesses of the Debtors (such as wages, salaries, commissions for services and payments for inventories, leased equipment and premises) and Claims of governmental units for taxes; (b) compensation for legal, financial, advisory, accounting and other services and reimbursement of expenses Allowed by the Bankruptcy Court under section 330, 331 or 503(b) of the Bankruptcy Code to the extent incurred prior to the Effective Date; and (c) all fees and charges assessed against the Debtors' Estates under section 1930, chapter 123 of title 28, United States Code.
- 1.2 Administrative Claims Reserve means the reserve established and maintained by the Plan Administrator, on behalf of the Reorganized Debtors, to pay Administrative Claims, Priority Tax Claims, Non-Tax Priority Claims and Other Secured Claims that become Allowed Claims after the Effective Date.
- 1.3 Administrator Professionals means the agents, financial advisors, attorneys, consultants, independent contractors, representatives and other professionals of the Plan Administrator and the Reorganized Debtors (in their capacities as such).
- 1.4 AgStar Litigation means that certain Complaint for Declaratory Judgment and Other Relief filed by AgStar Financial Services, PCA on December 23, 2008 as adversary case number 08-51897, pursuant to which AgStar Financial Services, PCA claimed that the liens it had on ethanol produced by the USBio Debtors were not released when the USBio Debtors transferred that ethanol to VeraSun Marketing, LLC.
- 1.5 AgStar Settlement means the settlement of the AgStar Litigation entered into by the Debtors and approved by the Bankruptcy Court on August 10, 2009, pursuant to the Order Pursuant to Fed. R. of Bankr. P. 9019 Authorizing the Debtors to Enter Into a Settlement Agreement With AgStar Financial Services, PCA (Docket No. 1680).
  - **1.6** Affiliate Debtors means the Debtors other than VEC.
- 1.7 Allowed . . . Claim means an Allowed Claim in the Class or category specified. Any reference herein to a particular Allowed Claim includes both the secured and unsecured portions of such Claim.
- 1.8 Allowed Claim means (a) any Claim for which a proof of claim has been timely Filed by the applicable Bar Date (or which pursuant to the Bankruptcy Code or a Final Order is not or shall not be required to be Filed); (b) any Claim that is listed in the Schedules as of the Effective Date as not disputed, not contingent, and not unliquidated, and for which no proof of
- DeltaView comparison of pedocs://chisr02a/747634/25 and pedocs://chisr02a/747634/26. Performed on 10/20/2009.

claim has been timely Filed; or (c) any Claim allowed pursuant to the Plan; provided, however, that with respect to any Claim in clauses (a) or (b) above, such Claim shall be considered Allowed only if and to the extent that (i) no objection to the allowance thereof has been interposed within the applicable period of time fixed by the Plan, the Bankruptcy Code, the Bankruptcy Rules or the Bankruptcy Court, or (ii) such an objection is so interposed and the Claim shall have been Allowed for Distribution purposes only by a Final Order. Any Claim that has been or is hereafter listed in the Schedules as disputed, contingent, or unliquidated, and for which no proof of claim has been timely Filed is not considered Allowed and shall be expunged without further action by the Reorganized Debtors and without any further notice to or action, order, or approval of the Bankruptcy Court.

- 1.9 ASA Debtors means ASA OpCo Holdings, LLC, ASA Albion, LLC, ASA Bloomingburg, LLC, and ASA Linden, LLC.
- specified Debtor(s) as of the date ten (10) Business Days prior to the date of any Distribution, other than: (a) with respect to Distributions to be made on the Distribution Date, (i) Cash to be distributed on the Distribution Date to Holders of Allowed Prepetition Secured Lender Claims, Allowed Administrative Claims, Allowed Priority Tax Claims, Allowed Other Secured Claims, and Allowed Non-Tax Priority Claims that became Allowed Claims on or before the Effective Date and (ii) Restricted Cash; and (b) with respect to Distributions to be made on any Quarterly Distribution Date, Restricted Cash.
- 1.11 Ballot means each of the ballot form or forms distributed to each Holder of an Impaired Claim, on which the Holder is to indicate acceptance or rejection of the Plan.
- **1.12** Bankruptcy Code means title 11, United States Code, as now in effect or hereafter amended.
- 1.13 Bankruptcy Court means the United States Bankruptcy Court for the District of Delaware.
- 1.14 Bankruptcy Rules means the Federal Rules of Bankruptcy Procedure and the Local Rules of Bankruptcy Practice and Procedure of the United States Bankruptcy Court for the District of Delaware, as now in effect or hereafter amended.
- 1.15 Bar Date means the applicable deadline by which a proof of claim must have been or must be Filed, as established by an order of the Bankruptcy Court, including the Bar Date Order and the Confirmation Order. The term "Bar Date" also includes the deadline for Filing Fee Claims established pursuant to Article XIV hereof, the deadline for Filing Administrative Claims established pursuant to Article XIV hereof, and the deadline for Filing Claims arising from rejection of executory contracts and unexpired leases established pursuant to Article IX hereof.
- 3 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- 1.16 Bar Date Order means the Order Under 11 U.S.C. §§ 105 and 502 and Fed. R. Bankr. P. 3003(c)(3) and 9007 (I) Establishing Deadline for Filing Proofs of Claim; (II) Approving Procedures for Filing Proofs of Claim; and (III) Approving the Form, Manner, and Sufficiency of Providing Notice Thereof (Docket No. 970), approved by the Bankruptcy Court on April 2, 2009, as the same may have been or hereafter may be amended, modified or supplemented.
- 1.17 Business Day means any day, other than a Saturday, Sunday or "legal holiday" (as defined in Bankruptcy Rule 9006(a)).
- 1.18 Cash means legal tender of the United States of America and equivalents thereof.
- 1.19 Chapter 11 Case(s) means (a) when used with reference to a particular Debtor, the chapter 11 case pending for that Debtor in the Bankruptcy Court and (b) when used with reference to all Debtors, the chapter 11 cases pending for the Debtors in the Bankruptcy Court.
- 1.20 Claim(s) means a "claim," as defined in section 101(5) of the Bankruptcy Code.
- Claims against the Debtors, which day shall be the later of (a) one hundred eighty (180) days after the Effective Date or (b) sixty (60) days after the Filing of a proof of claim for, or request for payment of, such Claim or such other date as the Bankruptcy Court may order. The Filing of a motion to extend the Claims Objection Deadline shall automatically extend the Claims Objection Deadline until a Final Order is entered on such motion. In the event that such motion to extend the Claims Objection Deadline is denied by the Bankruptcy Court, or approved by the Bankruptcy Court and reversed on appeal, the Claims Objection Deadline shall be the later of the current Claims Objection Deadline (as previously extended, as applicable) or 30 days after entry of a Final Order denying the motion to extend the Claims Objection Deadline.
- 1.22 Class means a category of Holders of Claims or Interests, as described in Article III hereof.
- 1.23 Collateral means any property or interest in property of the Debtors' Estates that is subject to a valid, perfected, and enforceable lien to secure a Claim.
- 1.24 Confirmation Date means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the docket of the Bankruptcy Court for the Chapter 11 Cases.
- 4 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- 1.25 Confirmation Hearing means the hearing held by the Bankruptcy Court pursuant to section 1128 of the Bankruptcy Code to consider confirmation of the Plan, as such hearing may be adjourned or continued from time to time.
- 1.26 Confirmation Order means the order of the Bankruptcy Court confirming the Plan pursuant to section 1129 of the Bankruptcy Code.
- 1.27 Creditors! Committee has the meaning ascribed to it in the Introduction hereto.
  - 1.28 **Debtor(s)** has the meaning ascribed to it in the Introduction hereto.
- 1.29 Deficiency Claim means the Claim of a Holder of a Prepetition Secured Lender Claim or Other Secured Claim equal to the amount by which such creditor's Allowed Claim exceeds the value of the Collateral securing such Claim, as determined pursuant to section 506 of the Bankruptcy Code.
- disallowed by Final Order, (b) is Scheduled at zero or as contingent, disputed, or unliquidated and as to which a Bar Date has been established but no proof of claim has been timely Filed or deemed timely Filed with the Bankruptcy Court pursuant to either the Bankruptcy Code or any Final Order of the Bankruptcy Court or otherwise deemed timely filed under applicable law, or (c) is not Scheduled and as to which a Bar Date has been established but no proof of claim has been timely Filed or deemed timely Filed with the Bankruptcy Court pursuant to either the Bankruptcy Code or any Final Order of the Bankruptcy Court or otherwise deemed timely filed under applicable law.
- 1.31 *Disbursing Agent* means any party designated by the Creditors' Committee in its sole discretion, after consultation with the Debtors, to serve as disbursing agent under the Plan, and any agent appointed by such Disbursing Agent for the purpose of effectuating such Distributions.
- 1.32 Disclosure Statement means the written disclosure statement (including all schedules thereto or referenced therein) in form and substance reasonably acceptable to the Plan Proponents, that relates to the Plan, as approved by the Bankruptcy Court pursuant to section 1125 of the Bankruptcy Code, as the same may be amended, modified or supplemented.
- 1.33 Disputed ... Claim means a Disputed Claim in the Class or category specified.
- 1.34 *Disputed Claim* means any Claim, or any portion thereof, that is neither an Allowed Claim nor a Disallowed Claim.
- 1.35 Disputed Claim Amount means the lesser of (a) the liquidated amount set forth in the proof of claim relating to a Disputed Claim, (b) the amount estimated by the
- 5 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Bankruptcy Court for purposes of Distributions in respect of such Disputed Claim in accordance with section 502(c) of the Bankruptcy Code pursuant to Section 8.3 hereof, and (c) the amount of such Disputed Claim Allowed by the Bankruptcy Court pursuant to section 502 of the Bankruptcy Code, or zero, if such Disputed Claim is disallowed in its entirety by the Bankruptcy Court pursuant to such section, in either case, regardless of whether the order or judgment allowing or disallowing such Claim has become a Final Order; provided, however, that in the event a Claim has been disallowed, but the order of disallowance has not yet become a Final Order, the Bankruptcy Court may require the Disbursing Agent to reserve, and hold in trust for the benefit of each Holder of such Claim, Cash in an amount equal to the Pro Rata Distributions which the Bankruptcy Court, in its discretion, determines will protect the rights of such Holder under all of the facts and circumstances relating to the order of disallowance and the appeal of such Holder from such order.

- 1.36 Disputed Claims Reserve means the reserve established and maintained by the Plan Administrator, on behalf of the Reorganized Debtors, to pay Disputed Other Deficiency Claims and Disputed General Unsecured Claims.
- 1.37 **Distribution** means the payment or delivery to any Holder of an Allowed Claim or Interest of the consideration payable to such Holder under the terms of and in accordance with the Plan.
- 1.38 Distribution Date means the date, occurring as soon as possible after the Effective Date (but not later than ten calendar days after the Effective Date, unless extended by the Plan Committee and the Reorganized Debtors), upon which initial Distributions from the Estates are made to Holders of Allowed Claims without further order of the Bankruptcy Court.
- 1.39 Distribution Record Date means (a) for all Holders of Allowed Claims other than Holders of Unsecured Note Claims, the Confirmation Date; and (b) for all Holders of Unsecured Note Claims, the close of business on the Business Day immediately preceding (i) the Distribution Date; and (ii) any other dates(s) Distributions are made to the Holders of Unsecured Note Claims.
- 1.40 Effective Date means the Business Day upon which the Plan becomes effective as provided in Article XI hereof.
- 1.41 Estate(s) means, individually, the estate of any of the Debtors and, collectively, the estates of all of the Debtors created under section 541 of the Bankruptcy Code.
- 1.42 Face Amount means (a) when used in reference to a Disputed or Disallowed Claim, the full stated amount claimed by the Claim Holder in any proof of claim timely filed with the Bankruptcy Court or otherwise deemed timely filed by any Final Order of the Bankruptcy Court or other applicable bankruptcy law, and (b) when used in reference to an Allowed Claim, the allowed amount of such claim.
- 6 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- 1.43 Fee Auditor means Warren H. Smith & Associates, P.C.
- 1.44 Fee Claim means an Administrative Claim under section 330(a), 331 or 503 of the Bankruptcy Code for compensation of a Professional or other entity for services rendered or expenses incurred in the Chapter 11 Cases on or prior to the Effective Date (including expenses of the members of the Creditors' Committee incurred as members of the Creditors' Committee in discharge of their duties as such).
- 1.45 *File, Filed or Filing* means file, filed or filing with the Bankruptcy Court or its authorized designee in the Chapter 11 Cases.
- 1.46 Final Order means an order or judgment of the Bankruptcy Court, or other court of competent jurisdiction, as entered on the docket in any of the Chapter 11 Cases, the operation or effect of which has not been stayed, reversed, or amended and as to which order or judgment (or any revision, modification, or amendment thereof) the time to appeal or seek review or rehearing has expired and as to which no appeal or petition for review or rehearing was filed or, if filed, remains pending; provided, however, that the possibility that a motion under section 502(j) of the Bankruptcy Code, Rule 59 or Rule 60 of the Bankruptcy Rules or any analogous rule may be, but has not been filed shall not cause an order not to be a Final Order.
- 1.47 General Unsecured Claim means a Claim that is not an Administrative Claim, Priority Tax Claim, Prepetition Secured Lender Claim, Other Secured Claim, Prepetition Makewhole Claim, Non-Tax Priority Claim, Lender Deficiency Claim, Other Deficiency Claim, or Guaranty Claim. For the avoidance of doubt, General Unsecured Claims include Unsecured Note Claims and Intercompany Claims (other than those Intercompany Claims described in the Intercompany Claims Settlement).
- 1.48 Guaranty Claim means a Claim arising from guaranties under the Prepetition Agstar Credit Agreements.
- 1.49 *Holder* means the Person holding the beneficial interest in a Claim or Interest.
- 1.50 *Impaired* means, when used in reference to a Claim or Interest, a Claim or Interest that is impaired within the meaning of section 1124 of the Bankruptcy Code.
- 1.51 Intercompany Claim means any Claim held by a Debtor against any other Debtor, including any intercompany book entries reflecting obligations owed by one Debtor with respect to any other Debtor.
- 1.52 Intercompany Claims Settlement means the agreement between VeraSun Marketing, LLC and the USBio Debtors to settle all intercompany claims between VeraSun Marketing, LLC and the USBio Debtors pursuant to which the VSE Debtors shall pay the Allowed USBio Intercompany Administrative Claim in full.
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- 1.53 Interest means the legal, equitable, contractual and other rights of the Holders of Old Equity, including the rights of any entity to purchase or demand the issuance of any of the foregoing, including (a) conversion, exchange, voting, participation and dividend rights; (b) liquidation preferences; (c) stock options, warrants and put rights; and (d) share-appreciation rights.
- 1.54 Internal Revenue Code means the Internal Revenue Code of 1986, as amended.
- 1.55 Lender Deficiency Claim means the Deficiency Claim of a Holder of a Prepetition Secured Lender Claim.
- 1.56 Litigation Claims means the Claims, rights of action, suits or proceedings, whether in law or in equity, whether known or unknown, that any Debtor or Estate may hold against any Person, including but not limited to the Claims, rights of action, suits and proceedings listed in the Plan Supplement, to be retained by the Estates pursuant to Section 7.7 of the Plan including, but not limited to, (i) any claims, rights of action, suits or proceedings arising under or resulting from contractual subordination or section 510(b) of the Bankruptcy Code or (ii) any claims under sections 544, 545, 547 and 548 of the Bankruptcy Code that may exist.
- 1.57 Litigation Reserve means the reserve account to be established and maintained by the Plan Administrator, on behalf of the Reorganized Debtors, into which the Reorganized Debtors shall from time to time deposit Cash (in such amounts and at such times as directed by the Plan Committee) to fund the costs and expenses associated with investigating and, as necessary, prosecuting, settling, dismissing or otherwise disposing of any or all of the Litigation Claims; provided, however, that at no time shall the amount of the Litigation Claims Reserve exceed \$100,000; provided, further, that the Plan Administrator may at any time increase or decrease the amount of the Litigation Claims Reserve with the consent of the Plan Committee.
- 1.58 Non-Tax Priority Claim means a Claim, other than an Administrative Claim or Priority Tax Claim, that is entitled to priority in payment pursuant to section 507(a) of the Bankruptcy Code.
  - 1.59 Non-VSE Debtors means the Debtors other than the VSE Debtors.
- 1.60 Old Common Stock means the common stock of VEC outstanding immediately prior to the Petition Date, including (i) treasury stock and all options, warrants, calls, rights, puts, awards, commitments or any other agreements of any character to acquire such common stock; and (ii) claims arising from rescission of a purchase or sale of the common stock of VEC, for damages arising from the purchase or sale of the common stock of VEC, or for reimbursement or contribution allowed under section 502 of the Bankruptcy Code on account of such a claim, which claims are subordinated pursuant to section 510(b) of the Bankruptcy Code.

- 1.61 Old Equity means, collectively, the Old Common Stock and the Subsidiary Interests.
- 1.62 Operating Reserve means the reserve account to be established and maintained by the Plan Administrator, on behalf of the Reorganized Debtors, into which the Reorganized Debtors shall from time to time deposit Cash (in such amounts and at such times as directed by the Plan Committee, as the case may be) to fund, among other things, the expenses of the Plan Administrator, the Reorganized Debtors and the Plan Committee and its professionals, as set forth more fully in the Plan Administrator Agreement; provided, however, that the Plan Committee shall not unreasonably refuse to fund the Operating Reserve in an amount sufficient to pay the reasonable fees and expenses of the Plan Administrator; provided, further, that if the Plan Committee should refuse to fund the Operating Reserve in an amount sufficient to pay the reasonable fees and expenses of the Plan Administrator, the Plan Administrator shall have the right to file an appropriate motion with the Bankruptcy Court requesting that the Court direct that the Operating Reserve be funded in an amount sufficient to pay the reasonable fees and expenses of the Plan Administrator.
- 1.63 Other Deficiency Claim means the Deficiency Claim of a Holder of an Other Secured Claim.
- 1.64 Other Secured Claim means a Claim (other than a Prepetition Secured Lender Claim) that is secured by a valid, perfected and enforceable lien on property in which a Debtor's Estate has an interest or that is subject to setoff under section 553 of the Bankruptcy Code, to the extent of the value of the Claim Holder's interest in the applicable Estate's interest in such property or to the extent of the amount subject to setoff, as applicable, as determined pursuant to section 506(a) of the Bankruptcy Code or, in the case of the setoff, pursuant to section 553 of the Bankruptcy Code.
- 1.65 *Person* means an individual, corporation, partnership, joint venture, association, joint stock company, limited liability company, limited liability partnership, trust, estate, unincorporated organization, or other entity.
- 1.66 Petition Date means October 31, 2008, the date on which the Debtors Filed the petitions for relief commencing the Chapter 11 Cases.
- 1.67 Plan means this chapter 11 plan of liquidation, including the Plan Supplement and all supplements, appendices and schedules thereto, either in its present form or as the same may be altered, amended or modified from time to time; provided, that each of the documents referenced in this definition shall be in form and substance reasonably acceptable to the Debtors, the Reorganized Debtors, the Creditors' Committee or the Plan Committee, as the case may be.

- 1.68 Plan Administrator means the person designated by the Creditors' Committee in its sole discretion, after consultation with the Debtors, prior to the Confirmation Date and approved by the Bankruptcy Court pursuant to the Confirmation Order to administer the Plan in accordance with the terms of the Plan and the Plan Administrator Agreement and to take such other actions as may be authorized under the Plan Administrator Agreement, and any successor thereto.
- 1.69 Plan Administrator Agreement means the agreement between and among the Debtors and the Plan Administrator, in form and substance acceptable to the Creditors' Committee in its sole discretion, specifying the rights, duties and responsibilities of and to be performed by the Plan Administrator under the Plan, in substantially the form set forth in the Plan Supplement; provided, that after the Effective Date, any modifications made to the Plan Administrator Agreement shall be in form and substance acceptable to the Plan Committee.
- 1.70 Plan Committee means the committee established pursuant to Section 7.8 of the Plan to supervise and direct the Plan Administrator, to monitor implementation of the Plan, and to take such other actions and have such other rights as are set forth in the Plan, including, without limitation, as set forth in Section 7.8 of the Plan and Article X of the Plan, or as may be approved by the Bankruptcy Court.
- 1.71 *Plan Proponents* means, individually, either the Creditors! Committee or the Debtors, as applicable, and, collectively, the Creditors! Committee and the Debtors.
- 1.72 Plan Supplement means the forms of documents specified in Section 14.8 of the Plan, which documents shall be in form and substance reasonably acceptable to the Plan Proponents, subject to sections 7.3 and 7.5 herein; provided, that after the Effective Date, any modifications made to any of the documents specified in Section 14.8 of the Plan shall be in form and substance reasonably acceptable to the Reorganized Debtors and the Plan Committee, subject to sections 7.3 and 7.5 herein.
  - 1.73 Precluded Claims means the claims defined in Section 12.4 of the Plan.
- 1.74 Prepetition Agstar Credit Agreements means the Prepetition Janesville Credit Agreement, the Prepetition Albert City Credit Agreement, the Prepetition Central City Credit Agreement, the Prepetition Dyersville Credit Agreement, the Prepetition Hankinson Credit Agreement, the Prepetition Ord Credit Agreement, and the Prepetition Woodbury Credit Agreement.
- 1.75 Prepetition Albert City Credit Agreement means the Master Loan Agreement dated as of November 15, 2005, by and among the VeraSun Albert City, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.

- 1.76 Prepetition Central City Credit Agreement means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Central City, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.77 Prepetition Credit Agreements means the Prepetition Dougherty/Marion Credit Agreement, the Prepetition First Bank Letter of Credit, the Prepetition WestLB/ASA Credit Agreement, and the Prepetition Agstar Credit Agreements.
- 1.78 Prepetition Dougherty/Marion Credit Agreements means (i) that certain Loan and Security Agreement, dated as of June 22, 2006, by and between Millennium Ethanol, LLC, a South Dakota limited liability company, and Dougherty Funding LLC, a Delaware limited liability company, as amended, restated, supplemented or otherwise modified from time to time; and (ii) that certain Revolving Credit and Security Agreement, dated as of June 29, 2006, by and between Millennium Ethanol, LLC, a South Dakota limited liability company, and Dougherty Funding LLC, a South Dakota state bank, as amended, restated, supplemented or otherwise modified from time to time.
- 1.79 Prepetition Dyersville Credit Agreement means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Dyersville, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.80 Prepetition First Bank Letter of Credit means that certain Letter of Credit Application, Reimbursement and Security Agreement, dated December 18, 2006, by and between Millennium Ethanol, LLC, a South Dakota limited liability company, and First Bank & Trust, a South Dakota state bank, as amended, restated, supplemented or otherwise modified from time to time.
- 1.81 Prepetition Hankinson Credit Agreement means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Hankinson, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.82 *Prepetition Janesville Credit Agreement* means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Janesville, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.83 Prepetition Ord Credit Agreement means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Ord, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.84 Prepetition Makewhole Claim means any Claim for any makewhole amount, prepayment premium, early termination fee, change of control premium or other similar amount asserted to be payable in respect of the VSE Secured Notes.
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- 1.85 Prepetition Makewhole Claims Reserve means the reserve maintained by the Plan Administrator, on behalf of the Reorganized Debtors, to pay Prepetition Makewhole Claims that become Allowed Claims after the Effective Date.
- 1.86 Prepetition Secured Lender Claim(s) means a Claim against a Debtor arising out of the Prepetition Credit Agreements that is secured by a valid, perfected and enforceable lien on property in which a Debtor's Estate has an interest, to the extent of the value of the Claim Holder's interest in the applicable Estate's interest in such property, as determined pursuant to Section 506(a) of the Bankruptcy Code.
- Agreement, dated as of February 6, 2006 (as amended, restated, supplemented or otherwise modified from time to time), among the ASA Debtors, as borrowers, ASA OpCo Holdings, LLC, as borrowers' agent, the lenders party thereto, WestLB AG, New York Branch, as administrative agent, First National Bank of Omaha as collateral agent and as the accounts bank, WestLB AG, New York Branch, as co-syndication agent, lead arranger, and sole lead bookrunner, First National Bank of Omaha and Standard Chartered Bank, as co-syndication agents and lead arrangers and CIT Capital USA INC. and ING Capital LLC, as co-documentation agents and lead arrangers.
- 1.88 Prepetition Woodbury Credit Agreement means the Master Loan Agreement dated as of November 15, 2005, by and among VeraSun Woodbury, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.89 *Priority Tax Claim* means a Claim of a governmental unit of the kind specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code.
- 1.90 *Professional* means (a) any professional employed in the Chapter 11 Cases pursuant to section 327 or 1103 of the Bankruptcy Code or otherwise and (b) any professional or other entity seeking compensation or reimbursement of expenses in connection with the Chapter 11 Cases pursuant to section 503(b)(4) of the Bankruptcy Code.
- 1.91 *Pro Rata* means, with respect to Claims, at any time, the proportion of the Face Amount of a Claim in a particular Class or Classes bears to the aggregate Face Amount of all Claims (including Disputed Claims, but excluding Disallowed Claims) in such Class or Classes, unless the Plan provides otherwise.
- 1.92 Quarter means the period beginning on the Effective Date and ending on the next June 30, September 30, December 31 and March 31, and each three-month period thereafter.
- 1.93 Quarterly Distribution Date means the last Business Day of the month following the end of each calendar quarter after the Effective Date; provided, however, that if the Effective Date is within 30 days of the end of a calendar quarter, the first Quarterly Distribution
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Date shall be the last Business Day of the month following the end of the first calendar quarter after the calendar quarter in which the Effective Date falls.

- **1.94** Reorganized means the applicable Debtor from and after the Effective Date.
- 1.95 Reserves means, collectively, the Administrative Claims Reserve, Disputed Claims Reserve, the Litigation Claims Reserve, the Operating Reserve, the Prepetition Makewhole Claims Reserve, and the Unclaimed Distribution Reserve.
- 1.96 Restricted Cash means the Cash segregated (whether physically or merely on the books and records of the Reorganized Debtors) by the Plan Administrator, on behalf of the Reorganized Debtors, to fund the Reserves.
- 1.97 Scheduled means with respect to any Claim or Interest, the status and amount, if any, of such Claim or Interest as set forth in the Schedules.
- 1.98 Schedules means the schedules of assets and liabilities, the list of Holders of Interests and the statements of financial affairs Filed by the Debtors on January 14, 2009, as such schedules have been or may be further modified, amended or supplemented in accordance with Fed. R. Bankr. P. 1009 or orders of the Bankruptcy Court.
- 1.99 Solicitation Order means the Order Under Bankruptcy Code Sections 105, 502, 1125 and 1126, Bankruptcy Rules 2002, 3003, 3017, 3018, 3020 and 9007 and Local Bankruptcy Rule 3017-1 (I) Approving the Form and Manner of Notice of the Disclosure Statement Hearing; (II) Determining Adequacy of the Disclosure Statement; (III) Establishing a Record Date and Voting, Objection and Other Deadlines With Respect to Confirmation of Plan; (IV) Scheduling Hearing on Confirmation of Plan; (V) Approving (A) the Form and Manner of Notice of the Hearing on Confirmation and Related Matters and (B) the Deadline for Filing Confirmation Objections; (VI) Approving the Solicitation Procedures; and (VII) Granting Certain Other Relief (Docket No. 1797), approved by the Bankruptcy Court on September 10, 2009, as the same may have been or hereafter may be amended, modified or supplemented; provided, that the Solicitation Order shall be in form and substance reasonably acceptable to the Plan Proponents.
- 1.100 Subsidiary Interests means, collectively, the issued and outstanding shares of stock of the Affiliate Debtors, as of the Petition Date.
- 1.101 Substantive Consolidation Order means the order, or provision of the Confirmation Order, substantively consolidating (a) the Chapter 11 Cases of the VSE Debtors as one consolidated group of Debtors and (b) the Chapter 11 Cases of the ASA Debtors as another consolidated group of Debtors, as provided in Sections 3.1 and 7.1 of the Plan; provided, that the Substantive Consolidation Order shall be in form and substance reasonably acceptable to the Plan Proponents.
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- 1.102 Unclaimed Distribution Reserve means the reserve established pursuant to Section 8.4(b)(i) of the Plan.
- 1.103 *Unimpaired* means, when used in reference to a Claim or Interest, a Claim or Interest that is not impaired within the meaning of section 1124 of the Bankruptcy Code.
- 1.104 Unsecured Note Claim means individually, a Claim of a Holder of the Unsecured Notes, and collectively, the Claims of all such Holders arising under or as a result of such Unsecured Notes.
- 1.105 Unsecured Note Indenture means that certain Indenture, dated as of May 16, 2007, as supplemented by the First Supplemental Indenture, dated as of December 31, 2007, among VEC, as Issuer, each of the Subsidiary Guarantors named therein, and Wells Fargo Bank, N.A., as the original Trustee, pursuant to which \$437,000,000 aggregate principal amount of 9\% Senior Exchange Notes due 2017 were issued by VEC.
- 1.106 Unsecured Note Indenture Trustee means HSBC Bank USA, National Association, as successor in interest to Wells Fargo Bank, N.A., in its capacity as indenture trustee under the Unsecured Note Indenture. Any reference to a Distribution or other payment being made "to the Unsecured Note Indenture Trustee" shall mean a Distribution or payment "to the Unsecured Note Indenture Trustee, for the benefit of Holders of Unsecured Notes (except to the extent the Distribution or other payment is in payment of the Unsecured Note Indenture Trustee Fees and Expenses)."
- 1.107 Unsecured Note Indenture Trustee Fees and Expenses means any and all fees, expenses, disbursements and advances of the Unsecured Note Indenture Trustee (and its counsel, agents and advisors) that are provided for under the Unsecured Note Indenture (including, without limitation, in connection with service on the Creditors' Committee, the Plan Committee (if applicable) and in connection with Distributions under the Plan), which are incurred at any time prior to or after the Effective Date.
- 1.108 Unsecured Notes means the 9\%% Senior Exchange Notes issued by VEC, under the Unsecured Note Indenture.
- 1.109 USBio Debtors means US BioEnergy Corporation, VeraSun Albert City, LLC, VeraSun Central City, LLC, VeraSun Dyersville, LLC, VeraSun Hankinson, LLC, VeraSun Janesville, LLC, VeraSun Ord, LLC, VeraSun Woodbury, LLC, and US Bio Marion, LLC.
- 1.110 USBio Intercompany Administrative Claim means the aggregate Allowed Administrative Claim of all the USBio Debtors against the VSE Debtors in the amount of \$1.5 million.
  - 1.111 *VEC* means VeraSun Energy Corporation.
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- 1.112 *VSE Debtors* means VEC, VeraSun Granite City, LLC, VeraSun Reynolds, LLC, VeraSun BioDiesel, LLC, VeraSun Litchfield, LLC, VeraSun Tilton, LLC, VeraSun Aurora Corporation, VeraSun Charles City, LLC, VeraSun Marketing, LLC, VeraSun Welcome, LLC, VeraSun Fort Dodge, LLC, and VeraSun Hartley, LLC.
- 1.113 *VSE Secured Notes* means the 9 7/8% million Secured Notes Due 2012 issued by VEC and guaranteed by the VSE Debtors other than VEC.
- 1.114 Voting Deadline means the date set by the Bankruptcy Court pursuant to the Solicitation Order by which all Ballots to accept or reject the Plan in accordance with section 1126 of the Bankruptcy Code must be received by the Debtors or their Bankruptcy Court appointed agent.

Rules of Interpretation and Computation of Time. For purposes of the Plan, unless otherwise provided herein: (a) whenever from the context it is appropriate, each term, whether stated in the singular or the plural, will include both the singular and the plural; (b) unless otherwise provided in the Plan, any reference in the Plan to a contract, instrument, release or other agreement or document being in a particular form or on particular terms and conditions means that such document will be substantially in such form or substantially on such terms and conditions; (c) any reference in the Plan to an existing document or schedule Filed or to be Filed means such document or schedule, as it may have been or may be amended, modified or supplemented pursuant to the Plan; (d) any reference to an entity as a Holder of a Claim or Interest includes that entity's successors and assigns; (e) all references in the Plan to Sections, Articles and Schedules are references to Sections, Articles and Schedules of or to the Plan; (f) the words "herein." "hereunder" and "hereto" refer to the Plan in its entirety rather than to a particular portion of the Plan; (g) captions and headings to Articles and Sections are inserted for convenience of reference only and are not intended to be a part of or to affect the interpretation of the Plan; (h) subject to the provisions of any contract, certificates of incorporation, by-laws, instrument, release or other agreement or document entered into in connection with the Plan, the rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, federal law, including the Bankruptcy Code and Bankruptcy Rules; (i) the rules of construction set forth in section 102 of the Bankruptcy Code will apply; and (j) in computing any period of time prescribed or allowed by the Plan, the provisions of Bankruptcy Rule 9006(a) will apply.

#### ARTICLE II

#### ADMINISTRATIVE EXPENSES AND PRIORITY TAX CLAIMS

2.1 Administrative Claims Except as otherwise provided for herein, and subject to the requirements of Article XII and Section 14.3 hereof, on, or as soon as reasonably practicable after (i) the Distribution Date, if such Administrative Claim is an Allowed Administrative Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

date such Administrative Claim becomes an Allowed Administrative Claim, each Holder of an Allowed Administrative Claim shall receive in full satisfaction, settlement and release of and in exchange for such Allowed Administrative Claim (A) Cash of the Debtor or consolidated group of Debtors against which such Administrative Claim is Allowed equal to the unpaid portion of such Allowed Administrative Claim or (B) such less favorable treatment as to which such Debtor, Reorganized Debtor, or consolidated group of Debtors or Reorganized Debtors (subject to the consent of the Plan Committee, which consent shall not be unreasonably withheld) and the Holder of such Allowed Administrative Claims shall have agreed upon in writing; provided, however, that Allowed Administrative Claims with respect to liabilities incurred by the Debtors or consolidated group of Debtors in the ordinary course of business during the Chapter 11 Cases shall be paid in the ordinary course of business in accordance with the terms and conditions of any agreements relating thereto. For the avoidance of doubt, the payment of Professional Fee Claims shall be governed by Section 14.3 herein.

Priority Tax Claims. On, or as soon as reasonably practicable after (i) the 2.2 Distribution Date, if such Priority Tax Claim is an Allowed Priority Tax Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Priority Tax Claim becomes an Allowed Priority Tax Claim, each Holder of an Allowed Priority Tax Claim shall receive in full satisfaction, settlement and release of and in exchange for such Allowed Priority Tax Claim (A) Cash of the Debtor or consolidated group of Debtors against which such Priority Tax Claim is Allowed equal to the amount of such Allowed Priority Tax Claim, (B) such less favorable treatment as to which such Debtor, Reorganized Debtor, or consolidated group of Debtors or Reorganized Debtors (subject to the consent of the Plan Committee, which consent shall not be unreasonably withheld) and the Holder of such Allowed Priority Tax Claims shall have agreed upon in writing; provided, however, that any Priority Tax Claim not due and owing on the Effective Date will be paid in accordance with this Section when such Claim becomes due and owing; or (C) at the option of the Reorganized Debtors and the Plan Committee, Cash of the Debtor or consolidated group of Debtors in an aggregate amount of such Allowed Priority Tax Claim payable in installment payments over a period of not more than five years after the Petition Date, pursuant to section 1129(a)(9)(c) of the Bankruptcy Code.

#### ARTICLE III

#### CLASSIFICATION OF CLAIMS AND INTERESTS

ASA Debtors shall be substantively consolidated for Plan voting and Distribution purposes as described in Article VII. The USBio Debtors shall not be substantively consolidated and this Plan is a separate Plan for each USBio Debtor. Each Debtor or group of consolidated Debtors has been assigned a number below for the purposes of classifying and treating Claims against and Interests in each Debtor or consolidated group of Debtors for balloting purposes. The Claims against and Interests in each Debtor or consolidated group of Debtors, in turn, have been assigned to separate lettered Classes with respect to each Debtor or consolidated group of Debtors, based on the type of

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Claim involved. Accordingly, the classification of any particular Claim against or Interest in any Debtor or consolidated group of Debtors depends on the particular Debtor against which such Claim is asserted (or in which such Interest is held) and the type of Claim or Interest in question. The number will denote which Debtor or consolidated group of Debtors against which a Claim or Interest has been asserted, and the letter will denote the Class of such Claim or Interest. The numbers applicable to the various Debtor groups are as follows:

	Number	Consolidated Debtor Group or Debtor Name
1		VSE Debtors
2		ASA Debtors
3		US BioEnergy Corporation
4		VeraSun Albert City, LLC
5		VeraSun Central City, LLC
6		VeraSun Dyersville, LLC
7		VeraSun Hankinson, LLC
8		VeraSun Janesville, LLC
9		VeraSun Ord, LLC
10		VeraSun Woodbury, LLC
11		US Bio Marion, LLC

#### 3.2 Classification of Claims and Interests.

(a) Pursuant to section 1122 of the Bankruptcy Code, set forth below is a designation of classes of Claims against and Interests in each of the Debtors. A Claim or Interest is placed in a particular Class for the purposes of voting on the Plan and of receiving Distributions pursuant to the Plan only to the extent that such Claim or Interest is an Allowed Claim or an Allowed Interest in that Class and such Claim or Interest has not been paid, released, or otherwise

<sup>17</sup> DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

settled prior to the Effective Date. In accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Claims and Priority Tax Claims have not been classified and their treatment is set forth in Article II above.

(b) Claims against and Interests in each of the Debtors are divided into lettered Classes. Not all of the Classes apply to every Debtor, and consequently not all of the lettered Classes appear in the case of each Debtor or group of consolidated Debtors. For purposes of voting, Claims within the Class shall be counted for each applicable Debtor or group of consolidated Debtors. Whenever such a Class of Claims or Equity Interests is relevant to a particular Debtor, that class of Claims or Interests shall be grouped under the appropriate lettered Class from the following list:

Class A	Class A consists of Prepetition Secured Lender Claims against the applicable Debtor or consolidated group of Debtors
Class B	Class B consists of Other Secured Claims against the applicable Debtor or consolidated group of Debtors
Class C	Class C consists of Prepetition Makewhole Claims against the applicable Debtor or consolidated group of Debtors
Class D	Class D consists of all Non-Tax Priority Claims against the applicable Debtor or consolidated group of Debtors
Class E	Class E consists of all Lender Deficiency Claims against the applicable Debtor or consolidated group of Debtors
Class F	Class F consists of all Other Deficiency Claims against the applicable Debtor or consolidated group of Debtors
Class G	Class G consists of all General Unsecured Claims against the applicable Debtor or consolidated group of Debtors
Class H	Class H consists of all Guaranty Claims against the applicable Debtor or consolidated group of Debtors
Class I	Class I consists of all Interests in the applicable Debtor or consolidated group of Debtors

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

#### ARTICLE IV

## IDENTIFICATION OF CLASSES OF CLAIMS AND INTERESTS IMPAIRED AND UNIMPAIRED BY THE PLAN

4.1 Classes of Claims that are Unimpaired. The following Classes of Claims are Unimpaired by the Plan:

Classes 2A and 4A through (Prepetition Secured Lender Claims

11A against Non-VSE Debtors other

than US Bio Energy Corporation)<sup>1</sup>

Class 1B through 11B (Other Secured Claims)

Class 1C (Prepetition Makewhole Claims

against the VSE Debtors)

Class 1D through 11D (Non-Tax Priority Claims)

**4.2** Impaired Classes of Claims and Interests. The following Classes of Claims and Interests are Impaired by the Plan:

Classes 2E and 4E through 11E (Lender Deficiency Claims against

Non-VSE Debtors other than US Bio

Energy Corporation)<sup>2</sup>

Classes 1F through 11F (Other Deficiency Claims)

Classes 1G through 11G (General Unsecured Claims)

Class 3H (Guaranty Claims against US Bio

Energy Corporation)

Classes 1I through 11I (Interests)

#### ARTICLE V

#### PROVISIONS FOR TREATMENT OF CLAIMS AND INTERESTS

#### 5.1 Unimpaired Classes of Claims.

<sup>&</sup>lt;sup>1</sup> There is no Class 3A because there are no Prepetition Secured Lender Claims against US BioEnergy Corporation.

<sup>&</sup>lt;sup>2</sup> There is no Class 3E because there are no Lender Deficiency Claims against US BioEnergy Corporation.

<sup>19</sup> DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- (a) <u>Class 2A (Prepetition Secured Lender Claims against the ASA</u>

  <u>Debtors)</u>. On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against the ASA Debtors shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing the obligations of the ASA Debtors under the Prepetition WestLB/ASA Credit Agreement. Any Deficiency Claim against the ASA Debtors that is ultimately Allowed shall be a Class 2E Claim.
- (b) <u>Class 4A (Prepetition Secured Lender Claims against VeraSun Albert City, LLC)</u>. On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Albert City, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing the obligations under the Prepetition Albert City Credit Agreement. Any Deficiency Claim against VeraSun Albert City, LLC that is ultimately Allowed shall be a Class 4E Claim.
- (c) <u>Class 5A (Prepetition Secured Lender Claims against VeraSun Central City, LLC)</u>. On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Central City, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Central City, LLC's obligations under the Prepetition Central City Credit Agreement. Any Deficiency Claim against VeraSun Central City, LLC that is ultimately Allowed shall be a Class 5E Claim.
- (d) Class 6A (Prepetition Secured Lender Claims against VeraSun Dyersville, LLC). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Dyersville, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Dyersville, LLC's obligations under the Prepetition Dyersville Credit Agreement. Any Deficiency Claim against Verasun Dyersville, LLC that is ultimately Allowed shall be a Class 6E Claim.
- (e) <u>Class 7A (Prepetition Secured Lender Claims against VeraSun Hankinson, LLC)</u>. On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Hankinson, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Hankinson, LLC's obligations

under the Prepetition Hankinson Credit Agreement. Any Deficiency Claim against VeraSun Hanksinson, LLC that is ultimately Allowed shall be a Class 7E Claim.

- (f) <u>Class 8A (Prepetition Secured Lender Claims against VeraSun Janesville, LLC</u>). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Janesville, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Janesville, LLC's obligations under the Prepetition Janesville Credit Agreement. Any Deficiency Claim against VeraSun Janesville, LLC that is ultimately Allowed shall be a Class 8E Claim.
- (g) Class 9A (Prepetition Secured Lender Claims against VeraSun Ord, LLC). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Ord, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Ord, LLC's obligations under the Prepetition Ord Credit Agreement. Any Deficiency Claim against VeraSun Ord, LLC that is ultimately Allowed shall be a Class 9E Claim.
- (h) Class 10A (Prepetition Secured Lender Claims against VeraSun Woodbury, LLC). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Woodbury, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Woodbury, LLC's obligations under the Prepetition Woodbury Credit Agreement. Any Deficiency Claim against VeraSun Woodbury, LLC that is ultimately Allowed shall be a Class 10E Claim.
- (i) <u>Class 11A (Prepetition Secured Lender Claims against US Bio</u>
  <u>Marion, LLC</u>). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against US Bio Marion, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing the obligations of US Bio Marion, LLC under the Prepetition Dougherty/Marion Credit Agreements. Any Deficiency Claim US Bio Marion, LLC that is ultimately Allowed shall be a Class 11E Claim.
- (j) <u>Class 1B through 11B (Other Secured Claims)</u>. On, or as soon as reasonably practicable after (i) the Distribution Date, if such Other Secured Claim is an Allowed Other Secured Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Other Secured Claim becomes an Allowed Other Secured Claim, the Holder of an

<sup>21</sup> DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Allowed Other Secured Claim shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Other Secured Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing such Other Secured Claim in an amount up to the amount of such Allowed Other Secured Claim.

- (k) Class 1C (Prepetition Makewhole Claims against the VSE Debtors). On, or as soon as reasonably practicable after (i) the Distribution Date, if such Prepetition Makewhole Claim is an Allowed Prepetition Makewhole Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Prepetition Makewhole Claim becomes an Allowed Prepetition Makewhole Claim, each Holder of an Allowed Prepetition Makewhole Claim against the VSE Debtors shall receive in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Makewhole Claim its Pro Rata share of the proceeds of the Collateral, if any, securing the obligations of the VSE Debtors under the VSE Secured Notes.
- (l) Class 1D through 11D (Non-Tax Priority Claims). On, or as soon as reasonably practicable after (i) the Distribution Date, if such Non-Tax Priority Claim is an Allowed Non-Tax Priority Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Non-Tax Priority Claim becomes an Allowed Non-Tax Priority Claim, each Holder of an Allowed Non-Tax Priority Claim shall receive, to the extent not previously received, in full satisfaction, settlement, compromise and release of and in exchange for such Allowed Non-Tax Priority Claim (A) Cash equal to the amount of such Allowed Non-Tax Priority Claim or (B) such less favorable treatment as to which the Debtors or Reorganized Debtors (subject to the consent of the Plan Committee) and the Holder of such Allowed Non-Tax Priority Claim have agreed upon in writing.

#### 5.2 Impaired Classes of Claims.

- Non-VSE Debtors other than US Bio Energy Corporation). On, or as soon as reasonably practicable after, (i) the Distribution Date if such Lender Deficiency Claim is an Allowed Lender Deficiency Claim as of the Effective Date or (ii) the first Quarterly Distribution Date after the date a Lender Deficiency Claim becomes an Allowed Lender Deficiency Claim, each Holder of an Allowed Lender Deficiency Claim, together with Holders of Allowed Other Deficiency Claims and Allowed General Unsecured Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed. On each ensuing Quarterly Distribution Date, each Holder of an Allowed General Unsecured Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed.
- (b) <u>Classes 1F through 11F (Other Deficiency Claims)</u>. On, or as soon as reasonably practicable after, (i) the Distribution Date if such Other Deficiency Claim is an
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Allowed Other Deficiency Claim as of the Effective Date or (ii) the first Quarterly Distribution Date after the date that an Other Deficiency Claim becomes an Allowed Other Deficiency Claim, each Holder of an Allowed Other Deficiency Claim, together with Holders of Allowed Lender Deficiency Claims, Allowed General Unsecured Claims, and Guaranty Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed. On each ensuing Quarterly Distribution Date, each Holder of an Allowed Other Deficiency Claims, along with Holders of Allowed Lender Deficiency Claims, Allowed General Unsecured Claims, and Allowed Guaranty Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed.

c) Classes 1G-11G (General Unsecured Claims). On, or as soon as reasonably practicable after, (i) the Distribution Date if such General Unsecured Claim is an Allowed General Unsecured Claim as of the Effective Date or (ii) the first Quarterly Distribution Date after the date a General Unsecured Claim becomes an Allowed General Unsecured Claim, each Holder of an Allowed General Unsecured Claim, together with Holders of Allowed Lender Deficiency Claims, Allowed Other Deficiency Claims, and Allowed Guaranty Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed. On each ensuing Quarterly Distribution Date, each Holder of an Allowed General Unsecured Claims, along with Holders of Allowed Lender Deficiency Claims, Allowed General Unsecured Claims, and Allowed Guaranty Claims shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed.

Notwithstanding the foregoing, the Reorganized Debtor shall pay, on or as soon as reasonably practicable after the Effective Date, all reasonable Unsecured Note Indenture Trustee Fees and Expenses, in full in Cash, without application to or approval of the Bankruptcy Court and without a reduction to the recoveries of the Holders of Unsecured Notes. Notwithstanding the foregoing, to the extent any Unsecured Note Indenture Trustee Fees and Expenses are not paid (including, without limitation, any fees or expenses incurred in connection with any unresolved litigation relating to Disputed Claims), the Unsecured Note Indenture Trustee may assert its charging lien against any recoveries received on behalf of its Holders for payment of such unpaid amounts.

(d) Class 3H (Guaranty Claims against US Bio Energy Corporation). On, or as soon as reasonably practicable after, (i) the Distribution Date if such Guaranty Claim is an Allowed Guaranty Claim as of the Effective Date or (ii) the first Quarterly Distribution Date after the date a Guaranty Claim becomes an Allowed Guaranty Claim, each Holder of an Allowed Guaranty Claim, together with Holders of Allowed Other Deficiency Claims and Allowed General Unsecured Claims against US Bio Energy Corporation, shall receive its Pro Rata share of Available Cash, if any, of US Bio Energy Corporation. On each ensuing Quarterly Distribution Date, each Holder of an Allowed Guaranty Claim, along with Holders of Other Deficiency Claims and General Unsecured Claims against US Bio Energy Corporation, shall receive its Pro Rata share of Available Cash of US Bio Energy Corporation.

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

5.3 Classes of Interests Impaired. Class 11-11 (Interests). All Interests, and all stock certificates, instruments, and other documents evidencing such Interests, shall be cancelled as of the Effective Date. The Holders of Interests in Class 11-111 shall not receive or retain any Distribution or other property on account of such Interests.

#### ARTICLE VI

#### ACCEPTANCE OR REJECTION OF THE PLAN

- 6.1 Impaired Classes of Claims Entitled to Vote. Each Impaired Class of Claims that will (or may) receive or retain property or any interest in property under the Plan shall be entitled to vote to accept or reject the Plan; provided, however, that the votes of Holders of Intercompany Claims shall not be solicited, and any Ballots submitted by such Holders shall not be counted.
- 6.2 Classes Deemed to Accept the Plan. Class 2A, Classes 4A through 11A, Classes 1B through 11B, Class 1C and Classes 1D through 11D are Unimpaired under the Plan. Pursuant to section 1126(f) of the Bankruptcy Code, such Classes are conclusively presumed to have accepted the Plan, and the votes of Holders of Claims in such Classes therefore shall not be solicited.
- 6.3 Acceptance by Impaired Classes. Class 2E, Classes 4E through 11E, Classes 1F through 11F, Classes 1G through 11G, and Class 3H are Impaired under the Plan and thus the votes of Holders of Claims in such Classes shall be solicited. Pursuant to section 1126(c) of the Bankruptcy Code and except as otherwise provided in section 1126(e) of the Bankruptcy Code, an Impaired Class of Claims has accepted the Plan if the Plan is accepted by the Holders of at least two-thirds in dollar amount and more than one-half in number of the Allowed Claims of such Class that have timely and properly voted to accept or reject the Plan.
- 6.4 Classes Deemed to Reject the Plan. Holders of Claims and Interests in Classes 1I-11I are not entitled to receive any Distribution under the Plan on account of their Interests and, therefore, votes to accept or reject the Plan shall not be solicited from Holders of Claims or Interests in such Classes.
- 6.5 Elimination of Classes. Any Class that does not contain any Allowed Claims or any Claims temporarily allowed for voting purposes under Bankruptcy Rule 3018, as of the date of the commencement of the Confirmation Hearing, shall be deemed to have been deleted from the Plan for purposes of (a) voting to accept or reject the Plan and (b) determining whether it has accepted or rejected the Plan under section 1129(a)(8) of the Bankruptcy Code.
- 6.6 Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code. To the extent that any Impaired Class votes to reject the Plan or is deemed to have rejected it, Debtors shall request confirmation of the Plan under section 1129(b) of the Bankruptcy Code.
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

requirements of section 1129 of the Bankruptcy Code must be satisfied separately with respect to each Debtor or consolidated group of Debtors. The Plan Proponents reserve the right to alter, amend, modify, revoke or withdraw the Plan (subject to the consent of the other Plan Proponent, as applicable, which consent shall not be unreasonably withheld) as it applies to any particular Debtor or consolidated group of Debtors. A determination by the Bankruptcy Court that the Plan, as it applies to a particular Debtor, is not confirmable pursuant to section 1129 of the Bankruptcy Code shall not limit or affect: (i) the confirmability of the Plan as it applies to any other Debtor; or (ii) the Debtors' ability to modify the Plan, subject to the limitations in this Section 6.7, as it applies to any particular Debtor, to satisfy the confirmation requirements of section 1129 of the Bankruptcy Code.

#### ARTICLE VII

#### MEANS FOR IMPLEMENTATION OF THE PLAN

#### 7.1 Substantive Consolidation.

- (a) The Plan contemplates and is predicated upon the substantive consolidation of certain of the Debtors' Estates, but only for purposes of voting on the Plan and making Distributions to Holders of Claims under the Plan. For purposes of the Plan, the VSE Debtors shall be substantively consolidated; the ASA Debtors shall be substantively consolidated; and the remaining Debtors shall not be substantively consolidated. Notwithstanding the foregoing, the Debtors reserve all rights (subject to the consent of the Creditors' Committee and, as applicable, the Plan Committee, which consent shall not be unreasonably withheld) with respect to substantive consolidation of any and all of the Debtors.
- With respect to the consolidated Debtor entities, on the Effective Date, and only as to the consolidated Debtor entities, (i) all assets and liabilities of the VSE Debtors and the ASA Debtors, respectively, will, for voting and Distribution purposes only, be merged or treated as if they were merged, (ii) each Claim against the VSE Debtors or the ASA Debtors, respectively, will be deemed a single Claim against and a single obligation of the VSE Debtors and the ASA Debtors, respectively, (iii) all Intercompany Claims by, between, and among the VSE Debtors and the ASA Debtors, respectively, will, for voting and Distribution purposes only, be eliminated, and (iv) any obligation of the VSE Debtors and the ASA Debtors, respectively, and all guaranties thereof by one or more of the other VSE Debtors and ASA Debtors, respectively, will be deemed to be one obligation of all of the VSE Debtors and the ASA Debtors, respectively. Except as set forth in this Article, such substantive consolidation shall not (other than for purposes related to the Plan) (x) affect the legal and corporate structures of the Debtors or Reorganized Debtors, (y) cause any Debtor to be liable for any Claim or Interest under the Plan which it otherwise is not liable, and the liability of any Debtor for such Claim or Interest shall not be affected by such substantive consolidation, and (z) except as otherwise stated in this Article VII, affect Intercompany Claims of Debtors against Debtors.

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

(c) Unless the Bankruptcy Court has approved by a prior order the substantive consolidation of certain of the Debtors' Estates, which order shall be in form and substance reasonably acceptable to the Plan Proponents, the Plan shall serve as, and shall be deemed to be, a request for entry of an order substantively consolidating certain of the Debtors' Estates, but only for purposes of voting on the Plan and making Distributions to Holders of Claims and Interests under the Plan. If no objection to substantive consolidation of certain of the Debtors' Estates is timely filed and served by any Holder of an Impaired Claim affected by the Plan as provided herein on or before the Voting Deadline or such other date as may be established by the Bankruptcy Court, the Confirmation Order shall serve as the order approving the substantive consolidation of certain of the Debtors' Estates, but only for purposes of voting on the Plan and making Distributions to Holders of Claims and Interests under the Plan. If any such objections are timely filed and served, a hearing with respect to the substantive consolidation of certain of the Debtors' Estates, but only for purposes of voting on the Plan and making Distributions to Holders of Claims and Interests under the Plan, and any objections thereto shall be part of the Confirmation Hearing.

# 7.2 Continued Corporate Existence; Dissolution of Reorganized Debtors.

- (a) The Debtors shall continue to exist as the Reorganized Debtors after the Effective Date in accordance with the laws of their respective states of incorporation or organization and pursuant to their respective certificates of incorporation, by-laws, articles of formation, operating agreements, and other organizational documents in effect prior to the Effective Date, except to the extent such certificate of incorporation and by-laws are amended under the Plan, for the limited purposes of liquidating all of the assets of the Estates and making Distributions in accordance with the Plan.
- (b) As soon as practicable after the Plan Administrator exhausts the assets of the Debtors' Estates by making the final Distribution of Cash under the Plan and the Plan Administrator Agreement, the Plan Administrator shall at the expense of the Debtors' Estates (i) provide for the retention and storage of the books, records and files that shall have been delivered to or created by the Plan Administrator until such time as all such books, records and files are no longer required to be retained under applicable law, and File a certificate informing the Bankruptcy Court of the location at which such books, records and files are being stored; (ii) File a certification stating that the assets of the Debtors' Estates have been exhausted and final Distributions of Cash have been made under the Plan; (iii) File the necessary paperwork in the respective jurisdictions in which they are incorporated or otherwise organized to effectuate the dissolution of the Reorganized Debtors in accordance with the laws of such jurisdiction or jurisdictions; and (iv) resign as the sole officer, manager, or director, as applicable, of the Reorganized Debtors. Upon the Filing of the certificate described in section (ii) of the preceding sentence, the Reorganized Debtors shall be deemed dissolved for all purposes without the necessity for any other or further actions to be taken by or on behalf of the Reorganized Debtors or payments to be made in connection therewith other than the filing of a motion for final decree with respect to a particular Debtor.

- 7.3 Certificate of Incorporation and By-laws. The certificate and articles of incorporation and by-laws of each Debtor shall be amended as necessary to satisfy the provisions of the Plan and the Bankruptcy Code and shall include, among other things, a provision (a) prohibiting the issuance of non-voting equity securities under Section 1123(a)(6) of the Bankruptcy Code and (b) limiting the activities of each Reorganized Debtor to matters authorized under the Plan. The amended certificate of incorporation and by-laws of each Debtor, in form and substance acceptable to the Creditors' Committee in its sole discretion and subject to the reasonable consent of the Debtors, shall be Filed on or before the date of the Confirmation Hearing.
- 7.4 Directors, Managers, and Officers; Effectuating Documents; Further Transactions. From and after the Effective Date, the Plan Administrator shall serve as the sole officer and director or manager, as applicable, of the Reorganized Debtors. Subject to the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan, the Plan Administrator shall be authorized to execute, deliver, file or record such documents, instruments, releases and other agreements and to take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.

### 7.5 The Plan Administrator.

- (a) Appointment. From and after the Effective Date, an entity to be designated by the Creditors' Committee in its sole discretion, in consultation with the Debtors, prior to the Confirmation Date shall serve as the Plan Administrator pursuant to the Plan Administrator Agreement and the Plan, until the resignation or discharge and the appointment of a successor Plan Administrator in accordance with the Plan Administrator Agreement and the Plan.
- (b) Plan Administrator Agreement. Prior to or on the Effective Date, the Debtors shall execute a Plan Administrator Agreement in substantially the same form as set forth in the Plan Supplement which Plan Administrator Agreement shall be in form and substance acceptable to the Creditors' Committee in its sole discretion. Any nonmaterial modifications to the Plan Administrator Agreement made by the Debtors, with the consent of the Creditors' Committee, prior to the Effective Date are hereby ratified. The Plan Administrator Agreement will contain provisions permitting the amendment or modification of the Plan Administrator Agreement necessary to implement the provisions of the Plan.
- (c) Rights, Powers and Duties of the Reorganized Debtors and the Plan Administrator. The Reorganized Debtors shall retain and have all the rights, powers and duties necessary to carry out their responsibilities under the Plan. Subject to the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan, such rights, powers and duties, which shall be exercisable by the Plan Administrator on behalf of the Reorganized Debtors and the Estates pursuant to the Plan and the Plan Administrator Agreement, shall include, among others:

<sup>27</sup> DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

- (i) liquidating the Reorganized Debtors' assets;
- (ii) investing the Cash of the Estates of the Reorganized Debtors, including, but not limited to, the Cash held in the Reserves in (A) direct obligations of the United States of America or obligations of any agency or instrumentality thereof that are backed by the full faith and credit of the United States of America, including funds consisting solely or predominantly of such securities; (B) money market deposit accounts, checking accounts, savings accounts or certificates of deposit, or other time deposit accounts that are issued by a commercial bank or savings institution organized under the laws of the United States of America or any state thereof; or (C) any other investments that may be permissible under section 345 of the Bankruptcy Code or as otherwise ordered by the Bankruptcy Court;
- (iii) calculating and paying all Distributions to be made under the Plan, the Plan Administrator Agreement and other orders of the Bankruptcy Court to Holders of Allowed Claims;
- (iv) employing, supervising and compensating professionals (including, for the avoidance of doubt, professionals retained by the Creditors' Committee and the Plan Committee) retained to represent the interests of and serve on behalf of the Reorganized Debtors and the Estates;
- (v) making and filing tax returns for any of the Debtors or the Reorganized Debtors;
- (vi) as provided in Section 7.5(g) hereof, objecting to Claims or Interests Filed against any of the Debtors' Estates on any basis;
- (vii) seeking estimation of contingent or unliquidated claims under section 502(c) of the Bankruptcy Code;
- (viii) seeking determination of tax liability under section 505 of the Bankruptcy Code;
- (ix) prosecuting avoidance actions under sections 544, 545, 547, 548, 549 and 553 of the Bankruptcy Code;
- (x) prosecuting turnover actions under sections 542 and 543 of the Bankruptcy Code;
- (xi) prosecuting, settling, dismissing or otherwise disposing of the Litigation Claims;

- (xii) closing the Chapter 11 Cases;
- (xiii) dissolving and winding up the Reorganized Debtors;
- (xiv) exercising all powers and rights, and taking all actions, contemplated by or provided for in the Plan Administrator Agreement; and
- (xv) taking any and all other actions necessary or appropriate to implement or consummate the Plan and the provisions of the Plan Administrator Agreement.
- (d) Compensation of the Plan Administrator. The Plan Administrator shall be compensated from the Operating Reserve pursuant to the terms of the Plan Administrator Agreement. Any professionals retained by the Plan Administrator shall be entitled to reasonable compensation for services rendered and reimbursement of expenses incurred from the Operating Reserve. The payment of the fees and expenses of the Plan Administrator and its retained professionals shall be made in the ordinary course of business and shall not be subject to the approval of the Bankruptcy Court; provided, however, that any disputes related to such fees and expenses shall be brought before the Bankruptcy Court.
- *Indemnification.* The Reorganized Debtors shall indemnify and hold harmless (i) the Plan Administrator (in its capacity as such and as officer, director, or manager, as applicable, of the Reorganized Debtors), (ii) such individuals that may serve as officers and directors of the Reorganized Debtors, if any, and (iii) the Administrator Professionals (collectively, the "Indemnified Parties"), from and against and with respect to any and all liabilities, losses, damages, claims, costs and expenses, including but not limited to attorneys' fees arising out of or due to their actions or omissions, or consequences of such actions or omissions, other than acts or omissions resulting from such Indemnified Party's willful misconduct or gross negligence, with respect to the Reorganized Debtors or the implementation or administration of the Plan or Plan Administrator Agreement. To the extent an Indemnified Party asserts a claim for indemnification as provided above, the legal fees and related costs incurred by counsel to the Plan Administrator in monitoring and participating in the defense of such claims giving rise to the asserted right of indemnification shall be advanced to such Indemnified Party (and such Indemnified Party undertakes to repay such amounts if it ultimately shall be determined that such Indemnified Party is not entitled to be indemnified therefore) out of the Operating Reserve or any insurance purchased using the Operating Reserve. The indemnification provisions of the Plan Administrator Agreement shall remain available to and be binding upon any former Plan Administrator or the estate of any decedent of the Plan Administrator and shall survive the termination of the Plan Administrator Agreement.
- (f) Insurance. The Plan Administrator shall be authorized to obtain and pay for out of the Operating Reserve all reasonably necessary insurance coverage for itself, its agents, representatives, employees or independent contractors, and the Reorganized Debtors,
- 29 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

including, but not limited to, coverage with respect to (i) any property that is or may in the future become the property of the Reorganized Debtors or their Estates and (ii) the liabilities, duties and obligations of the Plan Administrator and its agents, representatives, employees or independent contractors under the Plan Administrator Agreement (in the form of an errors and omissions policy or otherwise), the latter of which insurance coverage may, at the sole option of the Plan Administrator, remain in effect for a reasonable period (not to exceed seven years) after the termination of the Plan Administrator Agreement.

- (g) Authority to Object to Claims and Interests and to Settle Disputed Claims. From and after the Effective Date, the Plan Administrator, on behalf of the Reorganized Debtors, shall be authorized, with respect to those Claims or Interests which are not Allowed hereunder or by Court order, (i) to object to any Claims or Interests Filed against any of the Debtors' Estates and (ii) subject to Section 7.5(h) hereof, pursuant to Fed. R. Bankr. P. 9019(b) and section 105(a) of the Bankruptcy Code, to compromise and settle Disputed Claims, in accordance with the following procedures, which shall constitute sufficient notice in accordance with the Bankruptcy Code and the Bankruptcy Rules for compromises and settlements of claims:
  - (i) If the proposed amount at which the Disputed Claim is to be Allowed is less than or equal to \$500,000 or if the difference between the Scheduled amount of a Disputed Claim and the proposed amount at which the Disputed Claim is to be Allowed is less than \$100,000, the Reorganized Debtors and the Plan Administrator shall be authorized and empowered to settle the Disputed Claim and execute necessary documents, including a stipulation of settlement or release, in their sole discretion and without notice to any party, and the Plan Administrator shall have no liability to any party for the reasonableness of such settlement.
  - (ii) If the proposed amount at which the Disputed Claim is to be allowed is greater than \$500,000 but less than or equal to \$10 million or if the difference between the Scheduled amount of a Disputed Claim and the proposed amount at which the Disputed Claim is to be Allowed is equal to or greater than \$100,000 but less than \$500,000, the Reorganized Debtors and the Plan Administrator shall be authorized and empowered to settle such Disputed Claim and execute necessary documents, including a stipulation of settlement or release, only if the Plan Committee after ten (10) days notice does not object to such proposed settlement. If the Plan Committee objects to the settlement, the Plan Administrator may seek Bankruptcy Court approval of such settlement.
  - (iii) If the proposed amount at which the Disputed Claim is to be allowed is greater than \$10 million, the Reorganized Debtors and the Plan Administrator shall be authorized and empowered to settle the Disputed Claim and execute necessary documents, including a stipulation of settlement or release, only upon receipt of Bankruptcy Court approval of such settlement.

- (h) Authority to Settle Disputed Prepetition Makewhole Claims.

  Notwithstanding anything contained herein, the procedures set forth in Section 7.5(g) shall not apply to Disputed Prepetition Makewhole Claims. With respect to any such Disputed Prepetition Makewhole Claims, from and after the Effective Date, the Plan Administrator, on behalf of the Reorganized Debtors, shall be authorized to litigate, compromise and settle, pursuant to Fed. R. Bankr. P. 9019(b) and section 105(a) of the Bankruptcy Code, any Disputed Prepetition Makewhole Claims, subject to the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan. If the Plan Committee does not consent or objects to the settlement, the Plan Administrator may seek Bankruptcy Court approval of such settlement.
- 7.6 Revesting of Assets. Except as expressly provided elsewhere in this Plan, on the Effective Date, the property of each Debtor's Estate shall revest in the applicable Reorganized Debtor.

# 7.7 Preservation of Rights of Action; Settlement of Litigation Claims.

- (a) Preservation of Rights of Action. Except as otherwise provided in the Plan, the Confirmation Order, or in any document, instrument, release or other agreement entered into in connection with the Plan, in accordance with section 1123(b) of the Bankruptcy Code, the Debtors and their Estates shall retain the Litigation Claims. The Plan Administrator, on behalf of the Reorganized Debtors, may, subject to the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan, enforce, sue on, settle or compromise (or decline to do any of the foregoing) any or all of the Litigation Claims. Entry of the Confirmation Order shall not constitute a waiver or release by the Debtors or their Estates of any cause of action except as expressly provided for by the Plan. On and after the Effective Date, the Plan Administrator shall be assigned all Causes of Action arising under Sections 542, 543, 544, 547 through 551, and 553 of the Bankruptcy Code including Causes of Action arising under fraudulent transfer laws under federal and state statutes and common law and all Causes of Action against the Debtors' current and former officers, directors, employees, shareholders or non-legal professionals under section 1123(b)(3) of the Bankruptcy Code related to conduct in connection with the Debtors prior to the Petition Date, and any and all proceeds from insurance policies related thereto. The Plan Administrator shall be authorized to enforce, prosecute, settle or compromise the Causes of Action. The Plan Administrator may pursue such Causes of Action, in accordance with the obligations of the Plan and the best interests of all of the beneficiaries of the Plan. The method of distribution of the Estates' assets pursuant to the Plan shall not, and shall not be deemed to, prejudice the Causes of Action, which shall survive entry of the Confirmation Order for the beneficiaries of the Plan.
- (b) Settlement of Litigation Claims. At any time after the Confirmation Date and before the Effective Date, notwithstanding anything in the Plan to the contrary, but subject to the ultimate supervisory authority of the Creditors' Committee, as set forth in section 7.8 of the Plan, the Debtors may settle any or all of the Litigation Claims with the approval of the Bankruptcy Court pursuant to Fed. R. Bankr. P. 9019.

<sup>31</sup> DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

#### 7.8 Creditors' Committee and Plan Committee.

(a) Dissolution of Creditors' Committee. On the Effective Date, the Creditors' Committee shall dissolve automatically and its members shall be deemed released of all their duties, responsibilities and obligations in connection with the Chapter 11 Cases or the Plan and its implementation, and the retention or employment of the Creditors' Committee's attorneys, accountants and other agents shall terminate, except with respect to (i) all matters related to Fee Claims and (ii) any appeals of the Confirmation Order. All expenses of members of the Creditors' Committee and the fees and expenses of their professionals through the Effective Date shall be paid in accordance with the terms and conditions of the applicable orders of the Bankruptcy Court. Counsel to the Creditors' Committee shall be entitled to reasonable compensation and reimbursement of expenses for post-Effective Date activities authorized hereunder upon the submission of invoices to the Reorganized Debtors.

### (b) Creation of Plan Committee; Procedures.

- (i) On the Effective Date, the Plan Committee shall be formed and constituted unless otherwise determined in the sole discretion of the Creditors' Committee. The Plan Committee shall consist of three (3) members who shall be appointed by the Creditors' Committee and whose identities shall be disclosed to the Bankruptcy Court at the Confirmation Hearing. Membership on the Plan Committee shall be on an institutional and not on an individual basis. In the event that a member of the Plan Committee resigns from its position on the Plan Committee, such member shall have the right to designate its successor on the Plan Committee does not exercise its right to designate its successor on the Plan Committee within ten (10) days after such member's resignation from the Plan Committee, the non-resigning members of the Plan Committee shall have the right to designate a successor to the resigning member of the Plan Committee.
- (ii) In the event that there have been no Plan Committee members for a period of thirty (30) consecutive days, then the Plan Administrator may, during such vacancy and thereafter, in its sole discretion, ignore any reference in the Plan, the Plan Administrator Agreement or the Confirmation Order to a Plan Committee, and all references to the Plan Committee's ongoing duties and rights in the Plan, the Plan Administrator Agreement and the Confirmation Order shall be null and void.
- (c) Standing of Plan Committee. The Plan Committee shall have independent standing to appear and be heard in the Bankruptcy Court as to any matter relating to the Plan, the Plan Administrator Agreement, the Plan Administrator, the Estates or the
- 32 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Reorganized Debtors, including any matter as to which the Bankruptcy Court has retained jurisdiction pursuant to Article XIII of the Plan.

- (d) Function and Duration; Compensation and Expenses. The Plan Committee shall have ultimate supervisory authority over the Plan Administrator. The Plan Administrator shall report to the Plan Committee, and the Plan Committee shall have the power to remove the Plan Administrator. The Plan Committee (i) shall be responsible for (A) instructing and supervising the Reorganized Debtors and the Plan Administrator with respect to their responsibilities under the Plan and the Plan Administrator Agreement, (B) reviewing and approving the prosecution of adversary and other proceedings, if any, including approving proposed settlements thereof, (C) reviewing and approving objections to and proposed settlements of Disputed Claims, (D) performing such other duties that may be necessary and proper to assist the Plan Administrator and its retained professionals, and (ii) shall remain in existence until such time as the final Distributions under the Plan have been made by the Reorganized Debtors. The members of the Plan Committee shall serve without compensation for their performance of services as members of the Plan Committee, except that they shall be entitled to reimbursement of reasonable expenses by the Reorganized Debtors from the Operating Reserve, provided, however, that if the Unsecured Note Indenture Trustee shall be a member of the Plan Committee, it shall be entitled to be paid its Unsecured Note Indenture Trustee Fees and Expenses. The Plan Committee may retain counsel or other professionals who shall be entitled to reasonable compensation and reimbursement of actual, necessary expenses upon the submission of invoices to the Reorganized Debtors; provided, however, that any disputes related to such fees and expenses may be brought before the Bankruptcy Court.
- Liability; Indemnification. Neither the Plan Committee, nor any of its members or designees, nor any duly designated agent or representative of the Plan Committee, or their respective employees, shall be liable for the act or omission of any other member, designee, agent or representative of the Plan Committee, nor shall any member be liable for any act or omission taken or omitted to be taken in its capacity as a member of the Plan Committee, other than acts or omissions resulting from such member's willful misconduct or gross negligence. The Reorganized Debtors shall indemnify and hold harmless the Plan Committee and its members and designees, and any duly designated agent or representative thereof (in their capacity as such), from and against and with respect to any and all liabilities, losses, damages, claims, costs and expenses, including but not limited to attorneys' fees arising out of or due to their actions or omissions, or consequences of such actions or omissions, other than as a result of their willful misconduct or gross negligence, with respect to the Reorganized Debtors or the implementation or administration of the Plan. To the extent the Reorganized Debtors indemnify and hold harmless the Plan Committee and its members and designees, or any duly designated agent or representative thereof (in their capacity as such), as provided above, the legal fees and related costs incurred by counsel to the Plan Committee in monitoring and participating in the defense of such claims giving rise to the right of indemnification shall be paid out of the Operating Reserve.

### 7.9 Cancellation of Securities, Instruments and Agreements Evidencing

Claims and Interests. Except as otherwise provided in the Plan, on the Effective Date and concurrently with the applicable Distributions made pursuant to this Article VII, the promissory notes, share certificates (including treasury stock), the Unsecured Notes, the Unsecured Note Indenture, other instruments evidencing any Claims or Interests, and all options, warrants, calls, rights, puts, awards, commitments or any other agreements of any character to acquire such Interests shall be deemed automatically extinguished, canceled and of no further force and effect, without any further act or action under any applicable agreement, law, regulation, order or rule, and the obligations of the Debtors under the notes, share certificates, Unsecured Notes, the Unsecured Note Indenture and other agreements and instruments governing such Claims and Interests shall be automatically discharged. The Holders of or parties to such canceled notes, share certificates, Unsecured Notes, the Unsecured Note Indenture and other agreements and instruments shall have no rights arising from or relating to such notes, share certificates, Unsecured Notes, the Unsecured Note Indenture, and other agreements and instruments or the cancellation thereof, except the rights provided pursuant to the Plan; provided, however, that the Unsecured Notes and the Unsecured Note Indenture shall continue in effect solely for the purposes of (i) allowing the Unsecured Note Indenture Trustee or its agents to make Distributions to Holders of Unsecured Notes; (ii) allowing Holders of the Unsecured Notes to receive Distributions hereunder; and (iii) preserving the rights and liens of the Unsecured Note Indenture Trustee with respect to the Unsecured Note Indenture Trustee Fees and Expenses to the extent not otherwise paid. The Unsecured Note Indenture shall terminate completely upon the completion of all Distributions to the Holders of the Unsecured Notes and the payment in full of the Unsecured Note Indenture Trustee Fees and Expenses. Notwithstanding the foregoing, the indenture governing the Prepetition Makewhole Claims and any related documents thereto governing the rights and obligations of the Debtors and non-Debtor parties with respect to the Prepetition Makewhole Claims shall survive for purposes of the Prepetition Makewhole Claim litigation, until such time as the Prepetition Makewhole Claims have been Allowed by a Final Order and the obligations are satisfied in accordance with the terms of the Plan or the Prepetition Makewhole Claim have been disallowed by Final Order or expunged.

After the performance by the Unsecured Note Indenture Trustee or its agents of all duties that are required under the Plan, the Confirmation Order and/or under the terms of the Unsecured Note Indenture, the Unsecured Note Indenture Trustee and its agents and advisors shall be relieved of, and released from, all obligations associated with the Unsecured Notes arising under the Unsecured Note Indenture or under other applicable agreements or law, and the Unsecured Note Indenture Trustee shall be deemed fully released and discharged.

7.10 Sources of Cash for Plan Distributions. Except as otherwise provided in the Plan or the Confirmation Order, all Cash necessary for the Reorganized Debtors and the Plan Administrator to make payments pursuant to the Plan shall be obtained from the Debtors' Cash balances and the liquidation of the Debtors' and the Estates' remaining non-Cash assets, if any. Cash payments to be made pursuant to the Plan shall be made by the Reorganized Debtors (or any

successor thereto) or, if the Disbursing Agent is an entity other than the Reorganized Debtors, the Disbursing Agent.

- 7.11 Exemption from Transfer Taxes. Pursuant to section 1146 of the Bankruptcy Code, (a) the issuance, transfer or exchange of notes or equity securities under the Plan; (b) the creation of any mortgage, deed of trust, lien, pledge or other security interest; (c) the making or assignment of any contract, lease or sublease; or (d) the making or delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with, the Plan, including, without limitation, any merger agreements, agreements of consolidation, restructuring, disposition, liquidation or dissolution, deeds, bills of sale, and transfers of tangible property, will not be subject to any stamp tax, recording tax, personal property tax, real estate transfer tax, sales or use tax or other similar tax. The Confirmation Order shall direct the appropriate state or local government officials or agents to forgo the collection of any such tax or governmental assessment and to accept for filing and recordation any of the foregoing instruments or other documents without the payment of any such tax or governmental assessment.
- 7.12 Settlement of Certain Intercompany Claims. The Intercompany Claims Settlement shall be in full satisfaction of all of the USBio Debtors: claims against VeraSun Marketing, LLC and all of VeraSun Marketing, LLC: claims against the USBio Debtors. VeraSun Marketing, LLC and the USBio Debtors hereby release each other of and from any and all claims, causes of action or demands, whether known or unknown, that either now has or may ever have had against each other; provided, however, that VeraSun Marketing shall not be released from its obligation to transfer funds pursuant to the Intercompany Claims Settlement.
- 7.13 Agstar Settlement. Nothing set forth in this Plan shall be deemed to modify or amend the AgStar Settlement.

#### ARTICLE VIII

#### PROVISIONS GOVERNING DISTRIBUTIONS

- otherwise provided herein or as ordered by the Bankruptcy Court, initial Distributions to be made on account of Claims that are Allowed Claims as of the Effective Date shall be made on the Distribution Date. Any payment or Distribution required to be made under the Plan on a day other than a Business Day shall be made on the next Business Day. Distributions on account of Claims that first become Allowed Claims after the Effective Date shall be made pursuant to the terms and conditions of the Plan Administrator Agreement and Articles V, VIII and X of the Plan.
- 8.2 Interest on Claims. Unless otherwise specifically provided for in the Plan, the Confirmation Order or another order of the Bankruptcy Court, or if required by applicable bankruptcy law, interest, fees, costs and other charges accruing or incurred on or after the Petition Date shall not accrue or be paid on any Claims, and no Holder of a Claim shall be entitled to
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

interest, fees, costs or other charges accruing or incurred on or after the Petition Date through the date such Claim is satisfied in accordance with the terms of the Plan. In no event shall interest accrue after the Effective Date on account of any Allowed Claim unless specifically provided for in the Plan, applicable bankruptcy law, the Confirmation Order or another Final Order of the Bankruptcy Court.

8.3 Distributions by the Reorganized Debtors. The Reorganized Debtors shall make all Distributions of Cash required to be distributed under the applicable provisions of the Plan and the Plan Administrator Agreement. The Reorganized Debtors may employ or contract with other entities to assist in or make the Distributions required by the Plan and the Plan Administrator Agreement.

# 8.4 Delivery of Distributions and Undeliverable or Unclaimed Distributions.

- (a) Delivery of Distributions in General. Distributions to Holders of Allowed Claims shall be made at the addresses set forth in the Schedules unless such addresses are superseded by proofs of claim or transfers of claim Filed pursuant to Bankruptcy Rule 3001 (or at the last known addresses of such Holders if the Debtors or the Reorganized Debtors have been notified in writing of a change of address); provided, however, that in the case of (i) the Holders of Unsecured Notes, Distributions shall be made as set forth in clause (b) below; and (ii) a Claim Holder whose Claim is governed by an indenture or other agreement and is administered by an indenture trustee other than the Unsecured Note Indenture Trustee, Distributions shall be made at the addresses contained in the official records of the indenture trustee, including as set forth in any Ballots cast with respect to such Claims.
- (b) The Reorganized Debtors or any entities employed or contracted with to make Distributions pursuant to Section 8.3 above shall make all Distributions of Cash required to be made to Holders of Unsecured Notes to the Unsecured Note Indenture Trustee or its designated agent. The Unsecured Note Indenture Trustee or its designated agent shall make all Distributions to Holders of Unsecured Notes in accordance with the terms of the Unsecured Note Indenture, subject to the terms of the Plan; provided, however, that under all circumstances, the fees and expenses of the Unsecured Note Indenture or its designated agent incurred in connection with such Distributions shall be paid by the Reorganized Debtors without further approval from the Bankruptcy Court.

### (c) Surrender of Canceled Notes

As a condition precedent to receiving any Distribution hereunder on account of an Allowed Claim evidenced by an Unsecured Note canceled pursuant to Section 7.9 hereof, the Holder of such Allowed Claim shall surrender such Unsecured Note to the Unsecured Note Indenture Trustee for cancellation. All payments to Holders of Unsecured Notes shall only

be made after such surrender. Upon surrender of such Unsecured Notes certificates, the Unsecured Note Indenture Trustee shall cancel and destroy such Unsecured Notes. As soon as practicable after surrender of the Unsecured Notes certificates, the Unsecured Note Indenture Trustee shall distribute to the Holder thereof such Holder's pro rata share of the Distribution, but subject to the rights of the Unsecured Note Indenture Trustee to assert its charging lien to the extent the Unsecured Note Indenture Trustee Fees and Expenses are not paid pursuant to the Plan.

- (d) Undeliverable and Unclaimed Distributions.
- (i) Holding and Investment of Undeliverable and Unclaimed Distributions. If the Distribution to any Holder of an Allowed Claim is returned to the Disbursing Agent as undeliverable or is otherwise unclaimed, no further Distributions shall be made to such Holder unless and until the Disbursing Agent is notified in writing of such Holder's then-current address. Undeliverable and unclaimed Distributions shall be deposited in a segregated account, designated as an "unclaimed distribution reserve" (the "Unclaimed Distribution Reserve"), for the benefit of all such similarly situated Persons until such time as a Distribution becomes deliverable or is claimed. The account for the Unclaimed Distribution Reserve may be an interest-bearing account, provided that any interest accruing on funds in the Unclaimed Distribution Reserve shall be transferred to the Reorganized Debtors for Distribution in accordance with the terms of the Plan.
- (ii) After Distributions Become Deliverable. The Disbursing Agent and the Unsecured Note Indenture Trustee, as applicable, shall make all Distributions that have become deliverable or have been claimed since the Distribution Date on the next Quarterly Distribution Date, or as soon as practicable after such Distribution has become deliverable.
- Holder of an Allowed Claim that does not assert a claim pursuant to the Plan for an undeliverable or unclaimed Distribution within six months after the applicable date of Distribution shall be deemed to have forfeited its claim for such undeliverable or unclaimed Distribution and shall be forever barred and enjoined, notwithstanding any federal or state escheat laws to the contrary, from asserting any such claim for an undeliverable or unclaimed Distribution against the Debtors and the Estates, the Reorganized Debtors, the Plan Administrator, the Unsecured Note Indenture Trustee or their property. In such cases, any Cash in the Unclaimed Distribution Reserve, or held in trust by the Unsecured Note Indenture Trustee, for Distribution on account of such claims for undeliverable or unclaimed Distributions shall be Distributed to the other Holders in the same Class, free of any restrictions thereon. Nothing contained in the Plan or the Plan Administrator Agreement shall require the Unsecured Note Indenture Trustee and/or any Disbursing Agent, including, but not limited to, the Plan Administrator or the Reorganized Debtors, to attempt to

locate any Holder of an Allowed Claim for purposes of making any Distribution under the Plan.

- Debtors or the Unsecured Note Indenture Trustee, as the case may be, will have no obligation to recognize the transfer of, or the sale of any participation in, any Allowed Claim that occurs after the close of business on the Distribution Record Date, and will be entitled for all purposes herein to recognize and distribute only to those Holders of Allowed Claims who are Holders of such Claims, or participants therein, as of the close of business on the Distribution Record Date. The Disbursing Agent, the Reorganized Debtors, and the Unsecured Note Indenture Trustee shall instead be entitled to recognize and deal for all purposes under the Plan with only those record Holders stated on the official claims register or the transfer books and records as maintained by the Unsecured Note Indenture Trustee or its agents, as the case may be, as of the close of business on the Distribution Record Date.
- 8.6 Unsecured Note Indenture Trustee as Claim Holder. Consistent with Bankruptcy Rule 3003(c), the Debtors or the Reorganized Debtors, as the case may be, shall recognize proofs of claim filed by the Unsecured Note Indenture Trustee in respect of the Unsecured Note Claims. Accordingly, any Unsecured Note Claim, proof of which is filed by the registered or beneficial Holder of an Unsecured Note Claim, may be disallowed as duplicative of the Unsecured Note Claims of the Unsecured Note Indenture Trustee, without need for any further action or Bankruptcy Court order.
- 8.7 Allocation of Plan Distributions Between Principal and Interest. To the extent that any Allowed Claim entitled to a Distribution under the Plan is comprised of indebtedness and accrued but unpaid interest thereon, such Distribution shall, for federal and all other income tax purposes, be allocated to the principal amount of the Claim first and then, to the extent the consideration exceeds the principal amount of the claim, to the portion of such Claim representing unpaid interest accrued through the Effective Date.
- 8.8 Means of Cash Payment. Payments of Cash made pursuant to the Plan shall be in U.S. dollars and shall be made, at the option and in the sole discretion of the Plan Administrator, by (a) checks drawn on or (b) wire transfer from a domestic bank selected by the Plan Administrator. Checks issued by the Disbursing Agent shall be null and void if not negotiated within ninety days after issuance, but may be requested to be reissued until the distribution revests to the Reorganized Debtors pursuant to Section 8.4(b)(iii) hereof. Cash payments to foreign creditors may be made, at the option of the Plan Administrator, in such funds and by such means as are necessary or customary in a particular foreign jurisdiction.
- 8.9 Withholding and Reporting Requirements. In connection with the Plan and all Distributions thereunder, the Disbursing Agent or the Plan Administrator, as applicable, on behalf of the Reorganized Debtors, is authorized to take any and all actions that may be necessary or appropriate to comply with all withholding, payment and reporting requirements imposed by

<sup>38</sup> DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

any federal, state, local or foreign taxing authority, and all Allowed Claims and Distributions hereunder shall be subject to any such withholding and reporting requirements. The Disbursing Agent or the Plan Administrator, as applicable, on behalf of the Reorganized Debtors, shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding, payment and reporting requirements. All Holders of Claims shall be required to provide any information necessary to effect information reporting, payment and withholding with respect to such taxes.

- Bankruptcy Code or applicable nonbankruptcy laws, but shall not be required to, set off against any Claim and the payments or other Distributions to be made pursuant to the Plan in respect of such Claim, or claims of any nature whatsoever that the Debtors or Reorganized Debtors may have against the Holder of such Claim (except to the extent that such claims have been waived by the Debtors in the Plan or otherwise); provided, however, that neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Reorganized Debtors of any such claim that the Debtors or Reorganized Debtors may have against such Holder. In no event shall any Holder of Claims or Interests be entitled to setoff any Claim or Interest against any Claim, right, or cause of action of the Reorganized Debtors, unless such Holder has filed a motion with the Bankruptcy Court requesting the authority to perform such setoff on or before the Voting Deadline.
- 8.11 Fractional Dollars; De Minimis Distributions. Notwithstanding any other provision of the Plan or the Plan Administrator Agreement, (a) neither the Plan Administrator nor the Reorganized Debtors shall be required to make Distributions or payments of fractions of dollars, and whenever any payment of a fraction of a dollar under the Plan would otherwise be called for, the actual payment made shall reflect a rounding of such fraction to the nearest whole dollar (up or down), with half dollars being rounded down, and (b) the Plan Administrator shall have no obligation to make a Distribution on account of an Allowed Claim from any Reserve or account (i) to any Holder of an Allowed Claim if the aggregate amount of all Distributions authorized to be made from all such Reserves or accounts on the Quarterly Distribution Date in question is less than \$250,000, in which case such Distributions shall be deferred to the next Quarterly Distribution Date, or (ii) to a specific Holder of an Allowed Claim if the amount to be distributed to that Holder on the particular Distribution Date is less than \$50.00, unless such Distribution constitutes the final Distribution to such Holder.
- 8.12 Release of Liens. Except as otherwise provided in the Plan or in any contract, instrument, release or other agreement or document created or assumed in connection with the Plan, on the Effective Date and concurrently with the applicable Distributions made pursuant to the Plan, all mortgages, deeds of trust, liens, pledges or other security interests against the property of any Debtor's Estate shall be fully released and discharged (except for charging Liens of the Unsecured Note Indenture Trustee Fees and Expenses are not paid pursuant to the Plan), and all of the right, title and interest of any

<sup>39</sup> DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

holder of such mortgages, deeds of trust, liens, pledges or other security interests shall revert to the applicable Reorganized Debtor.

#### ARTICLE IX

# TREATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES

- 9.1 Rejected Contracts and Leases. Except as otherwise provided in the Plan, the Confirmation Order shall constitute an order under section 365 of the Bankruptcy Code rejecting all prepetition executory contracts and unexpired leases to which any Debtor is a party, on and subject to the occurrence of the Effective Date, unless such contract or lease (a) previously shall have been assumed or rejected by the Debtors, (b) previously shall have expired or terminated pursuant to its own terms before the Effective Date, or (c) is identified in the Plan Supplement as a contract or lease to be assumed; provided, however, that the Debtors may, subject to the consent of the Creditors' Committee (which consent shall not be unreasonably withheld), amend such list of contracts or leases to be assumed at any time prior to the Confirmation Date.
- Assumption of Director and Officer Liability Insurance Policies. 9.2 Notwithstanding anything herein to the contrary, as of the Effective Date, the Debtors shall assume (and assign to the Reorganized Debtors if necessary to continue the Debtors director and officer liability insurance policies in full force) all of the director and officer liability insurance policies pursuant to section 365(a) of the Bankruptcy Code. Entry of the Confirmation Order shall constitute the Bankruptcy Court's approval of the Debtors' foregoing assumption of each of the director and officer liability insurance policies. Notwithstanding anything to the contrary contained herein, Confirmation of the Plan shall not discharge, impair, or otherwise modify any obligations assumed by the foregoing assumption of the director and officer liability insurance policies, and each such obligation shall be deemed and treated as an Executory Contract that has been assumed by the Debtors under the Plan as to which no Proof of Claim need be Filed. On or before the Effective Date, the Debtors may obtain reasonably sufficient tail coverage (i.e., director and officer insurance coverage that extends beyond the end of the policy period) under a directors and officers' liability insurance policy for the current and former directors, officers, and managers for such terms or periods of time, and placed with such insurers as are determined by the Debtors and the Creditors!' Committee to be reasonable under the circumstances or as otherwise specified and ordered by the Bankruptcy Court in the Confirmation Order.
- 9.3 Bar to Rejection Damages. If the rejection of an executory contract or unexpired lease pursuant to Section 9.1 above gives rise to a Claim by the other party or parties to such contract or lease, such Claim shall be forever barred and shall not be enforceable against the applicable Debtor or its Estate, the Reorganized Debtors, the Plan Administrator or their respective successors or properties unless a proof of Claim is Filed and served on the Reorganized Debtors and counsel for the Reorganized Debtors within thirty (30) days after service of a notice of entry of the Confirmation Order or such other date as prescribed by the Bankruptcy Court.

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

#### ARTICLE X

# PROCEDURES FOR RESOLVING DISPUTED, CONTINGENT AND UNLIQUIDATED CLAIMS

- Objection Deadline; Prosecution of Objections. No later than the Claims Objection Deadline (unless extended by an order of the Bankruptcy Court), the Debtors, the Creditors' Committee, the Plan Committee or the Plan Administrator, on behalf of the Reorganized Debtors, as the case may be, shall File objections to Claims with the Bankruptcy Court and serve such objections upon the Holders of each of the Claims to which objections are made. Nothing contained herein, however, shall limit the right to object to Claims, if any, Filed or amended after the Claims Objection Deadline. Subject to the limitations set forth in the Plan Administrator Agreement and Section 7.5 of the Plan, and the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan, the Plan Administrator, on behalf of the Reorganized Debtors, shall be authorized to, and shall, resolve all Disputed Claims by withdrawing or settling such objections thereto, or by litigating to judgment in the Bankruptcy Court or such other court having jurisdiction over the validity, nature and/or amount thereof.
- 10.2 No Distributions Pending Allowance. Notwithstanding any other provision of the Plan, no payments or Distributions shall be made with respect to all or any portion of a Disputed Claim unless and until all objections to such Disputed Claim have been settled or withdrawn or have been determined by Final Order, and the Disputed Claim, or some portion thereof, has become an Allowed Claim.
- Reorganized Debtors, shall, subject to and in accordance with the provisions of the Plan Administrator Agreement, (a) establish one or more general accounts into which shall be deposited all funds not required to be deposited into any other account or Reserve and (b) create, fund and withdraw funds from, as appropriate, the Operating Reserve, Administrative Claims Reserve, Disputed Claims Reserve, the Prepetition Makewhole Claims Reserve, the Litigation Claims Reserve and Unclaimed Distributions Reserve. The Reorganized Debtors shall sell non-Cash assets of the Estates, if any, in accordance with the provisions of the Plan and the Plan Administrator Agreement. The net proceeds of any such sales shall be deposited in an account or Reserve pursuant to the terms of the Plan Administrator Agreement.
- (a) Disputed Claims Reserve. On Prior to the Effective Date (or as soon thereafter as is practicable) Distribution Date, subject to and in accordance with the provisions of the Plan Administrator Agreement, the Plan Administrator, in consultation with the Plan Committee, shall create and fund the Disputed Claims Reserve with an amount of the Estates' Cash equal to one hundred percent (100%) of Distributions to which Holders of Disputed General Unsecured Claims would be entitled under the Plan as of such date if such Disputed General Unsecured Claims were Allowed Claims in their Disputed Claims Amount; provided, however, that the Debtors, the Reorganized Debtors, the Creditors' Committee or the Plan Committee, as
- 41 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

applicable, may File motion(s) pursuant to section 502(c) of the Bankruptcy Code for order(s) estimating and limiting the amount of Cash which shall be deposited in the Disputed Claims Reserve in respect of any Disputed General Unsecured Claims, with notice and an opportunity to be heard to the affected Holders of such Disputed General Unsecured Claims and the Plan Committee; provided, further, that the Reorganized Debtors must within sixty (60) days after the Effective Date (or such other date as the Bankruptcy Court may order) File a motion(s) seeking to estimate any contingent or unliquidated Claims, with a notice and an opportunity to be heard to the affected Holders of such Disputed Claims and the Plan Committee. The Disputed Claims Reserve shall be fully funded no later than ten (10) days after the Bankruptcy Court has approved an order resolving such motion(s) and such order has been entered by the Clerk of the Bankruptcy Court.

- (b) Administrative Claims Reserve. On Prior to the Effective Date (or as soon thereafter as is practicable) Distribution Date and on the first Quarterly Distribution Date after the Effective Date, subject to and in accordance with the provisions of the Plan Administrator Agreement, the Plan Administrator, in consultation with the Plan Committee, shall create and fund the Administrative Claims Reserve with an amount of the Estates' Cash equal to the aggregate Disputed Claim Amount of all Disputed Administrative Claims, Disputed Priority Tax Claims, Disputed Non-Tax Priority Claims, and Disputed Other Secured Claims.
- (c) Prepetition Makewhole Claims Reserve. On Prior to the Effective Date (or as soon thereafter as is practicable) Distribution Date, subject to and in accordance with the provisions of the Plan Administrator Agreement, the Plan Administrator, in consultation with the Plan Committee, shall create and fund the Prepetition Makewhole Claims Reserve with an amount of the Estates' Cash equal to the aggregate Disputed Claim Amount of the Disputed Prepetition Makewhole Claims.
- (d) Litigation Claims Reserve. On Prior to the Effective Date (or as soon thereafter as is practicable) Distribution Date, subject to and in accordance with the Plan Administrator Agreement, the Plan Administrator shall create and fund the Litigation Claims Reserve with Cash of the Estates to fund the costs and expenses associated with investigating and, as necessary, prosecuting, settling, dismissing or otherwise disposing of any or all of the Litigation Claims; provided, however, that at no time shall the amount of the Litigation Claims Reserve exceed \$100,000; provided, further, that the Plan Administrator may at any time increase or decrease the amount of the Litigation Claims Reserve with the consent of the Plan Committee.
- payments and Distributions from the appropriate Reserves to the Holder of any Disputed Administrative Claim, Disputed Priority Tax Claim, Disputed Non-Tax Priority Claim, Disputed Other Secured Claim or Disputed General Unsecured Claim that has become an Allowed Claim, on the first Quarterly Distribution Date following the date that such Disputed Administrative Claim, Disputed Priority Tax Claim, Disputed Non-Tax Priority Claim, Disputed Other Secured Claim or Disputed General Unsecured Claim, as the case may be, becomes an Allowed Claim. Such Distributions shall be made in accordance with the Plan and the Plan Administrator
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Agreement. With respect to Disputed General Unsecured Claims, such Distribution shall be based upon the cumulative Distributions that would have been made to such Holder under the Plan if the Disputed General Unsecured Claim had been Allowed on the Effective Date and shall not be limited by the Disputed Claim Amounts previously reserved with respect to such Disputed General Unsecured Claim to the extent that additional amounts are available therefor from the Disputed Claims Reserve, but only to the extent that such additional amounts have not yet been distributed to Holders of Allowed General Unsecured Claims.

#### ARTICLE XI

#### CONFIRMATION AND CONSUMMATION OF THE PLAN

- Confirmation Order unless and until (a) the Bankruptcy Court shall have approved the Disclosure Statement, in a manner reasonably acceptable in form and substance to the Plan Proponents, as containing adequate information with respect to the Plan within the meaning of section 1125 of the Bankruptcy Code; (b) the proposed Confirmation Order shall be reasonably acceptable in form and substance to the Plan Proponents; and (c) the Substantive Consolidation Order, which may be the Confirmation Order, shall be reasonably acceptable in form and substance to the Plan Proponents and shall have been approved by the Bankruptcy Court prior to or contemporaneously with the Confirmation Order; and (d) subject to sections 7.3 and 7.5 hereof, the most current version of the Plan Supplement and all of the schedules, documents, and exhibits contained therein (including the Plan Administrator Agreement) shall be Filed in form and substance reasonably acceptable to the Plan Proponents.
- the Confirmation Order include a finding by the Bankruptcy Court that, notwithstanding Rule 3020(e) of the Federal Rules of Bankruptcy Procedure, the Confirmation Order shall take effect immediately upon its entry. The following are conditions precedent to the occurrence of the Effective Date: (a) the Debtors shall have Cash on hand sufficient to pay all Allowed Administrative Claims and Allowed Priority Tax Claims; (b) the Confirmation Order shall be a Final Order in form and substance reasonably acceptable to the Plan Proponents and no request for revocation of the Confirmation Order shall have been made or, if made, shall remain pending; (c) any order necessary to satisfy any condition to the effectiveness of the Plan shall have become a Final Order; and (d) subject to sections 7.3 and 7.5 hereof, the final version of the Plan Supplement and all of the schedules, documents, and exhibits contained therein (including the Plan Administrator Agreement) shall be Filed in form and substance reasonably acceptable to the Plan Proponents.
- 11.3 Waiver of Conditions Precedent. The Plan Proponents (subject to the consent of the other Plan Proponent, as applicable, which consent shall not be unreasonably withheld) may waive any of the conditions set forth in Section 11.1 or Section 11.2 at any time,
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

without any notice to other parties in interest and without any further notice to or action, order, or approval of the Bankruptcy Court, and without any formal action other than proceeding to confirm or consummate the Plan.

Date does not occur within sixty (60) days after the Confirmation Date, or by such later date, after notice and hearing, as is proposed by the Plan Proponents, then upon motion by either of the Plan Proponents and upon notice to such parties in interest as the Bankruptcy Court may direct, the Confirmation Order will be vacated by the Bankruptcy Court; provided, however, that, notwithstanding the Filing of such motion, the Confirmation Order may not be vacated if the Effective Date occurs before the Bankruptcy Court approves an order granting such motion. If the Confirmation Order is vacated pursuant to this Section 11.3, (a) the Plan shall be null and void in all respects; (b) any settlement of Claims provided for hereby shall be null and void without further order of the Bankruptcy Court; (c) the time within which the Debtors may assume and assign, or reject all executory contracts and unexpired leases shall be extended for a period of thirty (30) days after the date the Confirmation Order is vacated; and (d) nothing contained in the Plan or Disclosure Statement shall constitute a waiver or release of any Claims, Interests or Litigation Claims.

#### ARTICLE XII

#### EFFECT OF PLAN CONFIRMATION

12.1 Binding Effect. The Plan shall be binding upon and inure to the benefit of the Debtors, all present and former Holders of Claims and Interests, and their respective successors and assigns, including, but not limited to, the Reorganized Debtors.

#### 12.2 Releases.

(a) Releases by the Debtors.

Pursuant to section 1123(b) of the Bankruptcy Code, and except as otherwise specifically provided in the Plan, upon the Effective Date, each of the Debtors shall release unconditionally, and hereby is deemed to forever release unconditionally (i) the Creditors' Committee and the Plan Committee and, solely in their respective capacities as members or representatives of the Creditors' Committee and the Plan Committee, as applicable (and not as individual lenders or creditors to or on behalf of the Debtors), each member of the Creditors' Committee and the Plan Committee; (ii) the Plan Administrator; (iii) the Unsecured Note Indenture Trustee; and (iv) each of their respective agents, advisors, accountants, investment bankers, consultants, attorneys and other representatives of any of the foregoing or of the Debtors, solely in their respective capacities as such, and only with respect to their activities and conduct during or in connection with the preparation and filing of the Chapter 11 Cases, from any and all

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

claims, obligations, suits, judgments, damages, rights, causes of action and liabilities whatsoever (other than the right to enforce the performance of their respective obligations, if any, to the Debtors or the Reorganized Debtors under the Plan, the Plan Administrator Agreement and the contracts, instruments, releases and other agreements delivered under the Plan and the Plan Administrator Agreement), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are based in whole or in part on any act or omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Chapter 11 Cases, the Plan or the Disclosure Statement, other than Claims or liabilities arising out of or relating to any act or omission that constitutes a failure to perform the duty to act in good faith and where such failure to perform constitutes willful misconduct, gross negligence, or fraud.

### (b) Injunction Related to Releases.

The Confirmation Order will permanently enjoin the commencement or prosecution by any entity, whether directly, derivatively or otherwise, of any claims, obligations, suits, judgments, damages, demands, debts, rights, causes of action or liabilities released pursuant to the Plan.

12.3 Discharge of Claims and Termination of Interests. Pursuant to section 1141(d)(3) of the Bankruptcy Code, Confirmation will not discharge Claims against the Debtors; provided, however, that no Holder of a Claim against any Debtor may, on account of such Claim, seek or receive any payment or other Distribution from, or seek recourse against, any Debtor or Reorganized Debtor or each of their respective successors or their respective property, except as expressly provided herein. Accordingly, except as otherwise provided herein, the Confirmation Order shall provide, among other things, that from and after the Confirmation Date, all Persons who have held, hold, or may hold Claims against or Interests in the Debtors are (i) permanently enjoined from taking the following actions against the Estate(s) of the Debtors, the Plan Administrator, the Plan Committee, the Reorganized Debtors, the Unsecured Note Indenture Trustee, or any of their property on account of such Claims or Interests and (ii) preliminarily enjoined from taking any of the following actions against any of the Debtors, the Plan Administrator, the Plan Committee, or the Reorganized Debtors, the Unsecured Note Indenture Trustee, or their property on account of such Claims or Interests: (A) commencing or continuing, in any manner or in any place, any action or other proceeding; (B) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree, or order; (C) creating, perfecting, or enforcing any lien or encumbrance; and (D) commencing or continuing, in any manner or in any place, any action that does not comply with or is inconsistent with the provisions of the Plan; provided, however, that (x) nothing contained herein shall preclude such Persons from exercising their rights pursuant to and consistent with the terms of the Plan and (v) the preliminary injunction of actions against the Debtors, the Plan Administrator, the Plan Committee, the Reorganized Debtors, the Unsecured Note Indenture Trustee, and their property (if any) shall be dissolved and terminate one (1) day following the dissolution of the Reorganized Debtors and the completion

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

and winding-up of their affairs. Notwithstanding anything to the contrary in the Plan, creditors' rights of setoff and recoupment are preserved, and the injunctions referenced in this section or section 12.6 of the Plan shall not enjoin the valid exercise of such rights of setoff and recoupment. By accepting Distributions pursuant to the Plan, each Holder of an Allowed Claim or Allowed Interest shall be deemed to have specifically consented to the injunctions set forth in this Article XII.

specifically provided in the Plan, to the maximum extent permitted by the Bankruptcy Code and applicable law, none of (a) the Debtors, (b) the Reorganized Debtors, (c) the Plan Administrator, (d) the Creditors' Committee, (e) the Plan Committee, (f) the Unsecured Note Indenture Trustee, nor (g) any of their respective members, officers, directors, shareholders, employees, advisors, attorneys or agents acting in such capacity on or after the Petition Date, shall have or incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or (with respect to such Claims or Interests) any of their respective agents, affiliates, or any of their successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Debtors' Chapter 11 Cases, the pursuit of confirmation of the Plan, the consummation of the Plan or the administration of the Plan or the property to be distributed under the Plan, except for their willful misconduct or gross negligence (the "Precluded Claims"), and in all respects shall be entitled to rely reasonably upon the advice of counsel with respect to their duties and responsibilities under the Plan.

12.5 Release by Holders of Claims. Except as otherwise specifically provided in the Plan, on the Effective Date, each Person (excluding any of the Debtors) that votes to accept the Plan shall release unconditionally, and hereby is deemed to forever release unconditionally (i) the Creditors' Committee and the Plan Committee and, solely in their respective capacities as members or representatives of the Creditors' Committee and the Plan Committee, as applicable (and not as individual lenders or creditors to or on behalf of the Debtors), each member of the Creditors' Committee and the Plan Committee; (ii) the Plan Administrator; (iii) the Unsecured Note Indenture Trustee: and (iv) each of their respective agents, advisors, accountants, investment bankers, consultants, attorneys and other representatives of any of the foregoing or of the Debtors. solely in their respective capacities as such, and only with respect to their activities and conduct during or in connection with the preparation and filing of the Chapter 11 Cases, from any and all claims, obligations, suits, judgments, damages, rights, causes of action and liabilities whatsoever (other than the right to enforce the performance of their respective obligations, if any, to the Debtors or the Reorganized Debtors under the Plan, the Plan Administrator Agreement and the contracts, instruments, releases and other agreements delivered under the Plan and the Plan Administrator Agreement), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are based in whole or in part on any act or omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Chapter 11 Cases, the Plan or the Disclosure Statement other than Claims or liabilities arising out of or relating to any act or omission that constitutes a failure to perform the duty to act in good

faith and where such failure to perform constitutes willful misconduct, gross negligence, or fraud; provided, that this Section 12.5 shall not release any Person from any Claim or cause of action existing as of the Effective Date, based on (x) the Internal Revenue Code or any other domestic state, city or municipal tax code, (y) the environmental laws of the United States or any domestic state, city or municipality or (z) any criminal laws of the United States or any domestic state, city or municipality.

- 12.6 *Injunction.* Except as otherwise provided in the Plan, the Confirmation Order shall provide, among other things, that from and after the Confirmation Date all Persons who have held, hold or may hold Claims against or Interests in the Debtors are (i) permanently enjoined from taking any of the following actions against the Estate(s), the Plan Administrator, the Creditors' Committee (or any of its members from time to time), the Plan Committee (or any of its members from time to time), the Unsecured Note Indenture Trustee, or any of their property on account of any such Claims, Interests or Precluded Claims and (ii) preliminarily enjoined from taking any of the following actions against any of the Debtors, the Reorganized Debtors, the Plan Administrator, the Creditors' Committee (or any of its members from time to time), the Plan Committee (or any of its members from time to time) or their property on account of such Claims, Interests or Precluded Claims: (A) commencing or continuing, in any manner or in any place, any action or other proceeding; (B) enforcing, attaching, collecting or recovering in any manner any judgment, award, decree or order; (C) creating, perfecting or enforcing any lien or encumbrance; (D) asserting a setoff, right of subrogation or recoupment of any kind against any debt, liability or obligation due to the Debtors; and (E) commencing or continuing, in any manner or in any place, any action that does not comply with or is inconsistent with the provisions of the Plan. By accepting Distributions pursuant to the Plan, each Holder of an Allowed Claim receiving any Distribution pursuant to the Plan will be deemed to have specifically consented to the injunctions set forth in this Article XII.
- provided for in the Chapter 11 Cases under section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until all property of the Estates of the Reorganized Debtors has been distributed, the Reorganized Debtors have been dissolved, the Plan Administrator Agreement has terminated and the Bankruptcy Court has entered an order closing the Chapter 11 Cases; provided, however, that any injunction that by its terms is permanent or otherwise is intended to survive the Effective Date and Distributions hereunder (whether by law or pursuant to order of the Court) shall be continued without modification, notwithstanding anything to the contrary in the Plan.

#### ARTICLE XIII

#### RETENTION OF JURISDICTION

13.1 Retention of Jurisdiction. Pursuant to sections 105(c) and 1142 of the Bankruptcy Code and notwithstanding entry of the Confirmation Order and the occurrence of the

DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Effective Date, the Bankruptcy Court shall retain exclusive jurisdiction over all matters arising out of, and related to, the Chapter 11 Cases and the Plan to the fullest extent permitted by law, including, among other things, jurisdiction to:

- (a) Allow, disallow, determine, liquidate, classify, estimate or establish the priority or secured or unsecured status of any Claim or Interest, including the resolution of any request for payment of any Administrative Claim and the resolution of any objections to the allowance or priority of Claims or Interests;
- (b) Resolve any matters related to the assumption, assumption and assignment, or rejection of any executory contract or unexpired lease to which any Debtor is a party or with respect to which any Debtor or the Reorganized Debtors may be liable and to hear, determine and, if necessary, liquidate any Claims arising therefrom;
- (c) Ensure that Distributions to Holders of Allowed Claims are accomplished pursuant to the provisions of the Plan;
- (d) Decide or resolve any motions, adversary proceedings, contested or litigated matters and any other matters and grant or deny any applications involving the Debtors that may be pending on the Effective Date;
- (e) Approve such orders as may be necessary or appropriate to implement or consummate the provisions of the Plan and all contracts, instruments, releases and other agreements or documents created in connection with the Plan, the Disclosure Statement or the Confirmation Order;
- (f) Resolve any cases, controversies, suits or disputes that may arise in connection with the consummation, interpretation or enforcement of the Plan or any contract, instrument, release or other agreement or document that is executed or created pursuant to the Plan, or any entity's rights arising from or obligations incurred in connection with the Plan or such documents:
- (g) Modify the Plan before or after the Effective Date pursuant to section 1127 of the Bankruptcy Code or modify the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with the Plan, the Disclosure Statement or the Confirmation Order, or remedy any defect or omission or reconcile any inconsistency in any Bankruptcy Court order, the Plan, the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with the Plan, the Disclosure Statement or the Confirmation Order, in such manner as may be necessary or appropriate to consummate the Plan;
- (h) Hear and determine all applications for compensation and reimbursement of expenses of Professionals under the Plan or under sections 330, 331 and 503(b) of the Bankruptcy Code, *provided, however*, that from and after the Effective Date the payment of
- 48 DeltaView comparison of pedoes://chisr02a/747634/25 and pedoes://chisr02a/747634/26. Performed on 10/20/2009.

fees and expenses of the Reorganized Debtors, the Plan Committee, the Plan Administrator, and the Fee Auditor, including counsel fees, shall be made in the ordinary course of business and shall not be subject to the approval of the Bankruptcy Court;

- (i) Issue injunctions, approve and implement other orders or take such other actions as may be necessary or appropriate to restrain interference by any entity with consummation, implementation or enforcement of the Plan or the Confirmation Order;
- (j) Hear and determine the causes of action by or on behalf of the Debtors or the Reorganized Debtors, including causes of action relating to the Litigation Claims, including avoidance actions under sections 544, 545, 547, 548, 549 and 553 of the Bankruptcy Code and turnover actions under sections 542 and 543 of the Bankruptcy Code;
- (k) Hear and determine matters concerning state, local and federal taxes in accordance with sections 346, 505 and 1146 of the Bankruptcy Code;
- (I) Approve and implement such orders as are necessary or appropriate if the Confirmation Order is for any reason or in any respect modified, stayed, reversed, revoked or vacated or Distributions pursuant to the Plan are enjoined or stayed;
- (m) Determine any other matters that may arise in connection with or relate to the Plan, the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with the Plan, the Disclosure Statement or the Confirmation Order:
- (n) Enforce all orders, judgments, injunctions, releases, exculpations, indemnifications and rulings entered in connection with the Chapter 11 Cases;
- (o) Hear and determine all matters related to (i) the property of the Estates from and after the Confirmation Date, (ii) the winding up of the Debtors' affairs and (iii) the activities of the Reorganized Debtors and the Plan Administrator, including (A) challenges to or approvals of the Reorganized Debtors' or the Plan Administrator's activities, (B) resignation, incapacity or removal of the Plan Administrator and selection of a successor Plan Administrator, (C) reporting by, termination of and accounting by the Reorganized Debtors and the Plan Administrator and (D) release of the Plan Administrator from its duties;
- (p) Hear and determine disputes with respect to compensation of (i) the Reorganized Debtors' professional advisors, (ii) the Plan Administrator and its professional advisors, and (iii) the Plan Committee, its members and its professional advisors;
- (q) Hear and determine such other matters as may be provided in the Confirmation Order or as may be authorized under the Bankruptcy Code; and
  - (r) Approve an order closing the Chapter 11 Cases.
- 49 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

#### ARTICLE XIV

#### MISCELLANEOUS PROVISIONS

- 14.1 Effectuating Documents and Further Transactions. Each of the Debtors or the Plan Administrator, on behalf of the Reorganized Debtors, as applicable, is authorized to execute, deliver, file or record such contracts, instruments, releases and other agreements or documents and take such actions as may be necessary or appropriate to effectuate, implement and further evidence the terms and conditions of the Plan and any notes or securities issued pursuant to the Plan.
- 14.2 Corporate Action. Prior to, on or after the Effective Date (as appropriate), all matters provided for under the Plan that would otherwise require approval of the stockholders or directors of one (1) or more of the Debtors or the Reorganized Debtors shall be deemed to have occurred and shall be in effect prior to, on or after the Effective Date (as appropriate) pursuant to the applicable general corporation law of the states in which the Debtors or the Reorganized Debtors are incorporated without any requirement of further action by the stockholders or directors of the Debtors or the Reorganized Debtors.

#### 14.3 Bar Dates for Certain Claims.

- Administrative Claims; Substantial Contribution Claims. The (a) Confirmation Order will establish an Administrative Claims Bar Date for Filing Administrative Claims (including requests under section 503(b) of the Bankruptcy Code by any entity for making a substantial contribution in the Chapter 11 Cases) which date will be thirty (30) days after the Confirmation Date, except with respect to Fee Claims, which shall be subject to the provisions of Section 14.4. Holders of asserted Administrative Claims, except for Fee Claims, not paid prior to the Confirmation Date shall submit proofs of Claim on or before such Administrative Claims Bar Date or forever be barred from doing so. The notice of Confirmation to be delivered pursuant to Fed. R. Bankr, P. 3020(c) and 2002(f) will set forth such date and constitute notice of this Administrative Claims Bar Date. The Reorganized Debtors (in consultation with the Plan Committee) shall have sixty (60) days from the Effective Date (or such longer period as may be allowed by order of the Bankruptcy Court upon the request of the Reorganized Debtors or the Plan Committee) following the Administrative Claims Bar Date to review and object to such Administrative Claims before a hearing for determination of allowance of such Administrative Claims.
- (b) Professional Fee Claims. All Professionals and other entities requesting compensation or reimbursement of Fee Claims pursuant to section 327, 328, 330, 331 or 503(b) of the Bankruptcy Code for services rendered prior to the Effective Date shall File and serve on the Reorganized Debtors, counsel for the Plan Administrator and counsel for the Plan Committee, an application for final allowance of compensation and reimbursement of expenses no later than sixty (60) days after the Effective Date, unless otherwise ordered by the Bankruptcy
- DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Court. Objections to applications of such Professionals or other entities for compensation or reimbursement of expenses must be Filed and served on the Reorganized Debtors, counsel for the Plan Administrator, counsel for the Plan Committee and the requesting Professional or other entity no later than thirty (30) days (or such longer period as may be allowed by order of the Bankruptcy Court) after the date on which the applicable request for compensation or reimbursement was served. Except as otherwise provided in the Plan, Professionals shall be paid pursuant to sections 328, 330, or 331 of the Bankruptcy Code and prior orders of the Bankruptcy Court for amounts earned through the Effective Date. After the Effective Date, the Plan Administrator shall, in the ordinary course of business and without the necessity for any approval by the Bankruptcy Court, pay the reasonable fees and expenses, incurred after the Effective Date, of the professional persons employed by the Debtors, the Reorganized Debtors, the Creditors' Committee and Plan Committee, and of the Fee Auditor, as the case may be, in connection with the implementation and consummation of the Plan, the claims reconciliation process, and any other matters as to which such professionals may be engaged. The fees and expenses of such professionals shall be paid within ten (10) calendar days after submission of a detailed invoice therefor. If the Plan Administrator disputes the reasonableness of any such invoice, the Plan Administrator shall timely pay the undisputed portion of such invoice, and the Plan Administrator or the affected professional may submit such dispute to the Bankruptcy Court for a determination of the reasonableness of such invoice. Upon the Effective Date, any requirement that Professionals comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date shall terminate, and the Plan Administrator, in consultation with the Plan Committee, may employ any Professional in the ordinary course of business.

Professional Fee Holdback. On or as soon as reasonably practicable after the Effective Date and entry of an order of the Bankruptcy Court authorizing the Debtors to pay all amounts owing to Professionals for all outstanding amounts relating to prior periods through the Effective Date, the Debtors shall pay such amounts to such Professionals in accordance with such order; provided, however, that Professionals shall continue to prepare monthly fee applications in accordance with the Fee Order up to the Effective Date. Professionals shall estimate fees and expenses due for periods that have not been billed as of the Effective Date. If a Professional does not provide an estimate, the Debtors may estimate the unbilled fees and expenses of such Professional. On the Effective Date, the Reorganized Debtors shall fund an escrow account in an amount equal to the aggregate amount of outstanding fee applications not ruled upon by the Bankruptcy Court as of the Effective Date plus the aggregate amount of all estimated fees and expenses due for periods that have not been billed as of the Effective Date. Such escrow account shall not be considered property of the Reorganized Debtors. Such escrow account shall be used by the Reorganized Debtors to pay the remaining Professional Claims owing to the Professionals as and when Allowed by the Bankruptcy Court. When all Professional Claims have been paid in full, amounts remaining in such escrow account, if any, shall be returned to the Reorganized Debtors. To the extent that the escrow does not contain sufficient funds to pay all Professional Claims, such Professional Claims shall be paid from funds held in the Operating Reserve.

- 14.4 Payment of Statutory Fees. All fees payable pursuant to 28 U.S.C. § 1930, as determined by the Bankruptcy Court at the Confirmation Hearing, shall be paid on the Effective Date. The Debtors and the Reorganized Debtors shall remain liable for any quarterly fees validly due and owing to the United States Trustee under 28 U.S.C. § 1930 through and including such dates that their respective Chapter 11 Cases are converted to cases under chapter 7, dismissed, or closed.
- Bankruptcy Code and, to the extent applicable, sections 1122, 1123 and 1125 of the Bankruptcy Code, the Plan Proponents (subject to the consent of the other Plan Proponent, as applicable) reserve the right to alter, amend or modify the Plan at any time prior to or after the Confirmation Date but prior to the Effective Date. A Holder of a Claim or Interest that has accepted the Plan shall be deemed to have accepted the Plan, as altered, amended or modified, if the proposed alteration, amendment or modification does not materially and adversely change the treatment of the Claim or Interest of such Holder.
- term or provision of the Plan is determined by the Bankruptcy Court to be invalid, void or unenforceable, the Bankruptcy Court shall have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision will then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of the Plan will remain in full force and effect and will in no way be affected, impaired or invalidated by such holding, alteration or interpretation. The Confirmation Order will constitute a judicial determination and will provide that each term and provision of the Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.
- 14.7 Successors and Assigns. The Plan shall be binding upon and inure to the benefit of the Debtors, and their respective successors and assigns, including, without limitation, the Reorganized Debtors. The rights, benefits and obligations of any entity named or referred to in the Plan shall be binding on, and shall inure to the benefit of any heir, executor, administrator, successor or assign of such entity.
- 14.8 Plan Supplement. The Plan Supplement, comprised of, among other things, the forms of the documents relating to the Amended Certificate of Incorporation, By-Laws and the Plan Administrator Agreement, shall be (i) Filed with the Bankruptcy Court and (ii) posted on the website established for these cases located at www.kccllc.net/verasun, no later than three business days prior to the Voting Deadline. Upon its Filing, the Plan Supplement may be inspected in the office of the clerk of the Bankruptcy Court or its designee during normal business hours. Holders of Claims and Interests may obtain a copy of the Plan Supplement upon written request to the Debtors in accordance with Section 14.10 of the Plan. The documents contained in

the Plan Supplement shall be approved by the Bankruptcy Court pursuant to the Confirmation Order.

Proponents, by mutual agreement, each reserve the right to revoke or withdraw the Plan as to any or all of the Debtors prior to the Confirmation Date and to File subsequent plans of reorganization. If either of the Plan Proponents revokes or withdraws the Plan as to any or all of the Debtors, or if confirmation or consummation of the Plan as to any or all of the Debtors does not occur, then, with respect to such Debtors, (a) the Plan shall be null and void in all respects, (b) any settlement or compromise embodied in the Plan (including the fixing or limiting to an amount certain any Claim or Class of Claims), assumption or rejection of executory contracts or leases affected by the Plan, and any document or agreement executed pursuant to the Plan, shall be deemed null and void, and (c) nothing contained in the Plan shall (i) constitute a waiver or release of any Claims by or against, or any Interests in, such Debtors or any other Person, (ii) prejudice in any manner the rights of such Debtors or any other Person, or (iii) constitute an admission of any sort by the Debtors or any other Person.

14.10 Notice. All notices, requests and demands to or upon the Debtors, the Reorganized Debtors, the Creditors' Committee or the Plan Committee, to be effective shall be in writing and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows (or at such other address for such entity as shall be specified by like notice):

If to the Debtors:

VeraSun Energy Corporation 110 N. Minnesota Ave., Suite 300 Sioux Falls, SD 57104 Telephone: (605) 696-7200

Facsimile: (605) 978-7050

Attn: Mark Dickey

#### with copies to:

SKADDEN, ARPS, SLATE, MEAGHER & FLOM LLP One Rodney Square P.O. Box 636 Wilmington, Delaware 19899 Telephone: (302) 651-3000 Facsimile: (302) 651-3001 Attn: Mark S. Chehi, Esq. Megan E. Cleghorn, Esq. —Davis Lee Wright, Esq.

- and -

SKADDEN, ARPS, SLATE, MEAGHER

& FLOM LLP

with copies to:

155 North Wacker Drive

Chicago, Illinois 60606-1285 Telephone: (312) 407-0700 Facsimile: (312) 407-0411

Attn: Felicia Gerber Perlman, Esq.

John K. Lyons, Esq.

# If to the Reorganized Debtors:

Telephone:
Facsimile: Reorganized VeraSun Energy Corporation
110 N. Minnesota Ave., Suite 300
Sioux Falls, SD 57104
Telephone; (605) 696-7200
<u>Facsimile: (605) 978-7050</u>
Attn: []Mark Dickey

54 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

Γ	
Г	
	I - B - offician discrimina
Telephone: [	
Faccinalle:	
	1
2 1 1 1 1 1	

# SKADDEN, ARPS, SLATE, MEAGHER

& FLOM LLP

One Rodney Square

P.O. Box 636

Wilmington, Delaware 19899

Telephone: (302) 651-3000

Facsimile: (302) 651-3001

Attn: Mark S. Chehi, Esq.

Davis Lee Wright, Esq.

<u>- and -</u>

# SKADDEN, ARPS, SLATE, MEAGHER

& FLOM LLP

155 North Wacker Drive

Chicago, Illinois 60606-1285

Telephone: (312) 407-0700

Facsimile: (312) 407-0411

Attn: Felicia Gerber Perlman, Esq.

John K. Lyons, Esq.

If to the Creditors' Committee:

#### AKIN GUMP STRAUSS HAUER

& FELD LLP

One Bryant Park

New York, New York 10036

Telephone: (212) 872-1000

Facsimile: (212) 872-1002

Attn: Michael S. Stamer, Esq.

David H. Botter, Esq. Alexis Freeman, Esq.

If to the Plan Committee:

<del>[+]</del>

55 DeltaView comparison of pcdocs://chisr02a/747634/25 and pcdocs://chisr02a/747634/26. Performed on 10/20/2009.

AKIN GUMP STRAUSS HAUER
& FELD LLP

One Bryant Park

New York, New York 10036

Telephone: (212) 872-1000 Facsimile: (212) 872-1002

Attn: Michael S. Stamer, Esq. David H. Botter, Esq.

Alexis Freeman, Esq.

14.11 Governing Law. Except to the extent the Bankruptcy Code, the Bankruptcy Rules or other federal law is applicable, or to the extent a schedule to the Plan or a document or instrument contained in the Plan Supplement provides otherwise, the rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, the laws of Delaware, without giving effect to the principles of conflicts of law of such jurisdiction.

14.12 Schedules. All schedules, exhibits or other related documents to the Plan and the Plan Supplement are incorporated and are a part of the Plan as if set forth in full herein.

14.13 Filing of Additional Documents. On or before the Effective Date, the Debtors shall File such agreements and other documents, in form and substance acceptable the Creditors' Committee, as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.

Wilmington, Delaware Dated:

July 31, 2009

Respectfully submitted,

VERASUN ENERGY CORPORATION (for itself and on behalf of the Affiliate Debtors)

By: <u>/s/ Mark Dickey</u> Name: Mark Dickey

Title: Senior Vice President, Secretary and

General Counsel

#### Counsel:

/s/ Mark S. Chehi

Mark S. Chehi (I.D. No. 2855)

Megan E. Cleghorn (I.D. No. 4080)

Davis Lee Wright (I.D. No. 4324)

Skadden, Arps, Slate, Meagher & Flom LLP

One Rodney Square

P.O. Box 636

Wilmington, DE 19899

(302) 651-3000

- and -

Felicia Gerber Perlman John K. Lyons Skadden, Arps, Slate, Meagher & Flom LLP 155 North Wacker Drive Chicago, Illinois 60606 (312) 407-0700

Attorneys for VeraSun Energy Corporation, et al., Debtors and Debtors-in-Possession.

/s/ Dennis A. Meloro

Dennis A. Meloro (No. 4435) Donald J. Detweiler (No. 3087) Sandra G. M. Selzer (No. 4283) Greenberg Traurig, LLP The Nemours Building 1007 North Orange Street, Suite 1200 Wilmington, DE 19801 (301) 661-7000

and –

Michael S. Stamer
David H. Botter
Akin Gump Strauss Hauer & Feld LLP
One Bryant Park
New York, New York 10036
(212) 872-1000

Attorneys for the Official Committee of Unsecured Creditors of VeraSun Energy Corporation, et al.

Document comparison done by DeltaView on Tuesday, October 20, 2009 6:25:01 PM

Document 1	pcdocs://chisr02a/747634/25
Document 2	pcdocs://chisr02a/747634/26
Rendering set	Option 3a strikethrough double score no moves

Insertion	•	·	
<del>Deletion</del>			
<moved from-=""></moved>			
> <u>Moved to</u> <			
Style change	•		
Format change		 	
Moved deletion			
Inserted cell			
Deleted cell		473	2,50
Moved cell			
Split/Merged cell	,	 	
Padding cell			

	Count	
Insertions		26
Deletions		25
Moved from		0
Moved to		0
Style change		0
Format changed		1
Total changes		52

<sup>3</sup> DeltaView comparison of pedocs://chisr02a/747634/25 and pedocs://chisr02a/747634/26. Performed on 10/20/2009.

# Exhibit B

Plan of Liquidation

# IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:	)
	) Case No. 08-12606 (BLS)
VERASUN ENERGY CORPORATION, et al.,	)
	) Chapter 11
Debtors.	)
	Jointly Administered
	,
	)

# JOINT PLAN OF LIQUIDATION OF VERASUN ENERGY CORPORATION AND ITS AFFILIATED DEBTORS

Mark S. Chehi (I.D. No. 2855) Davis Lee Wright (I.D. No. 4324) Skadden, Arps, Slate, Meagher & Flom LLP One Rodney Square P.O. Box 636 Wilmington, DE 19899 (302) 651-3000

- and -

Felicia Gerber Perlman John K. Lyons Skadden, Arps, Slate, Meagher & Flom LLP 155 North Wacker Drive Chicago, Illinois 60606 (312) 407-0700

Attorneys for VeraSun Energy Corporation, et al., Debtors and Debtors-in-Possession

Dated: Wilmington, Delaware July 31, 2009

Dennis A. Meloro (No. 4435)
Donald J. Detweiler (No. 3087)
Sandra G. M. Selzer (No. 4283)
Greenberg Traurig, LLP
The Nemours Building
1007 North Orange Street, Suite 1200
Wilmington, DE 19801
(301) 661-7000
- and -

Michael S. Stamer
David H. Botter
Alexis Freeman
Akin Gump Strauss Hauer & Feld LLP
One Bryant Park
New York, New York 10036
(212) 872-1000
Attorneys for the Official Committee of

Attorneys for the Official Committee of Unsecured Creditors of VeraSun Energy Corporation, et al.

# TABLE OF CONTENTS

INTRODUCTION1	
	ARTICLE I
	DEFINED TERMS AND RULES OF INTERPRETATION
1.1	Administrative Claim2
1.2	Administrative Claims Reserve2
1.3	Administrator Professionals
1.4	AgStar Litigation2
1.5	AgStar Settlement2
1.6	Affiliate Debtors2
1.7	Allowed Claim
1.8	Allowed Claim2
1.9	ASA Debtors3
1.10	Available Cash3
1.11	Ballot3
1,12	Bankruptcy Code3
1.13	Bankruptey Court3
1.14	Bankruptcy Rules3
1.15	Bar Date
1.16	Bar Date Order4
1.17	Business Day4
1.18	Cash4
1.19	Chapter 11 Case(s)4
1.20	Claim(s)4
1.21	Claims Objection Deadline4
1.22	Class
1.23	Collateral4
1.24	Confirmation Date
1.25	Confirmation Hearing5
1.26	Confirmation Order5
1.27	Creditors' Committee
1.28	Debtor(s)5
1.29	Deficiency Claim5
1.30	Disallowed Claim5
1.31	Disbursing Agent5
1.32	Disclosure Statement
1.33	Disputed Claim5
1.34	Disputed Claim5

1.35	Disputed Claim Amount	
1.36	Disputed Claims Reserve	
1.37	Distribution	
1.38	Distribution Date	6
1.39	Distribution Record Date	6
1.40	Effective Date	6
1.41	Estate(s)	6
1.42	Face Amount	6
1.43	Fee Auditor	7
1.44	Fee Claim	7
1.45	File, Filed or Filing	7
1.46	Final Order	
1.47	General Unsecured Claim	
1.48	Guaranty Claim	7
1.49	Holder	
1.50	Impaired	7
1.51	Intercompany Claim	
1.52	Intercompany Claims Settlement	8
1.53	Interest	8
1.54	Internal Revenue Code	8
1.55	Lender Deficiency Claim	8
1.56	Litigation Claims	8
1.57	Litigation Reserve	8
1.58	Non-Tax Priority Claim	
1.59	Non-VSE Debtors	8
1.60	Old Common Stock	8
1.61	Old Equity	
1.62	Operating Reserve	9
1.63	Other Deficiency Claim	9
1.64	Other Secured Claim	
1.65	Person	
1.66	Petition Date	
1.67	Plan	
1.68	Plan Administrator	
1.69	Plan Administrator Agreement	
1.70	Plan Committee	
1.71	Plan Proponents	
1.72	Plan Supplement	
1.73	Precluded Claims	
1.74	Prepetition Agstar Credit Agreements	
1.75	Prepetition Albert City Credit Agreement	
1.76	Prepetition Central City Credit Agreement	
1.77	Prepetition Credit Agreements	

1.79	Prepetition Dyersville Credit Agreement	11
1.80	Prepetition First Bank Letter of Credit	11
1.81	Prepetition Hankinson Credit Agreement	
1.82	Prepetition Janesville Credit Agreement	
1.83	Prepetition Ord Credit Agreement	
1.84	Prepetition Makewhole Claim	12
1.85	Prepetition Makewhole Claims Reserve	12
1.86	Prepetition Secured Lender Claim(s)	12
1.87	Prepetition WestLB/ASA Credit Agreement	
1.88	Prepetition Woodbury Credit Agreement	12
1.89	Priority Tax Claim	12
1.90	Professional	
1.91	Pro Rata	12
1,92	Quarter	
1.93	Quarterly Distribution Date	
1.94	Reorganized	
1.95	Reserves	
1.96	Restricted Cash	
1.97	Scheduled	13
1.98	Schedules	13
1.99	Solicitation Order	
1.100	Subsidiary Interests	13
1.101	Substantive Consolidation Order	
1,102	Unclaimed Distribution Reserve	
1.103	Unimpaired	
1.104	Unsecured Note Claim	
1.105	Unsecured Note Indenture	
1.106	Unsecured Note Indenture Trustee	
1.107	Unsecured Note Indenture Trustee Fees and Expenses	14
1.108	Unsecured Notes	
1.109		14
1.110		
1.111	VEC	15
1.112	VSE Debtors	15
	VSE Secured Notes	
1.114	Voting Deadline	15
	ARTICLE II	
	ADMINISTRATIVE EXPENSES AND PRIORITY TAX CLAIMS	
2,1	Administrative Claims	16

2.2	Priority Tax Claims	16
	ARTICLE III	
	CLASSIFICATION OF CLAIMS AND INTERESTS	
3.1	The Debtors	17
3.2	Classification of Claims and Interests	18
	ARTICLE IV	
	IDENTIFICATION OF CLASSES OF CLAIMS AND INTERESTS IMPAIRED AND UNIMPAIRED BY THE PLAN	
4.1 4.2	Classes of Claims that are Unimpaired	19 19
	ARTICLE V	
	PROVISIONS FOR TREATMENT OF CLAIMS AND INTERESTS	
5.1	Unimpaired Classes of Claims	20
5.2	Impaired Classes of Claims	22
5.3	Impaired Classes of Interests	24
	ARTICLE VI	
	ACCEPTANCE OR REJECTION OF THE PLAN	
6.1	Impaired Classes of Claims Entitled to Vote	
6.2	Classes Deemed to Accept the Plan	24
6.3	Acceptance by Impaired Classes	24
6.4	Classes Deemed to Reject the Plan	24
6.5	Elimination of Classes	25
6.6 6.7	Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code  Confirmability and Severability of a Plan	∠⊃ ⊃5
0.7		23
	ARTICLE VII	
	MEANS FOR IMPLEMENTATION OF THE PLAN	
7.1	Substantive Consolidation	25
7.2	Continued Corporate Existence; Dissolution of Reorganized Debtors	
7.3	Certificate of Incorporation and By-laws	27

7.4	Directors, Managers, and Officers; Effectuating Documents; Further Transactions27	
7.5	The Plan Administrator27	
7.6	Revesting of Assets31	
7.7	Preservation of Rights of Action; Settlement of Litigation Claims31	
7.8	Creditors' Committee and Plan Committee	
7.9	Cancellation of Securities, Instruments and Agreements Evidencing Claims and	
	Interests34	
7.10	Sources of Cash for Plan Distributions35	
7.11	Exemption from Transfer Taxes35	
7.12	Settlement of Certain Intercompany Claims35	
7.13	Agstar Settlement	
, <del></del>	·	
	ARTICLE VIII	
	PROVISIONS GOVERNING DISTRIBUTIONS	
8.1	Distributions for Claims Allowed as of the Effective Date36	
8.2	Interest on Claims	
8.3	Distributions by the Reorganized Debtors36	
8.4	Delivery of Distributions and Undeliverable or Unclaimed Distributions36	
8.5	Record Date for Distributions38	
8.6	Unsecured Note Indenture Trustee as Claim Holder38	
8.7	Allocation of Plan Distributions Between Principal and Interest39	
8.8	Means of Cash Payment39	
8.9	Withholding and Reporting Requirements39	
8.10	Setoffs	
8.11	Fractional Dollars; De Minimis Distributions40	
8.12	Release of Liens40	
	ARTICLE IX	
	TREATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES	
9.1	Rejected Contracts and Leases40	
9.2	Assumption of Director and Officer Liability Insurance Policies41	
9.3	Bar to Rejection Damages41	

## ARTICLE X

# PROCEDURES FOR RESOLVING DISPUTED, CONTINGENT AND UNLIQUIDATED CLAIMS

10.1	Objection Deadline; Prosecution of Objections	41
10.2	No Distributions Pending Allowance	
10.3	Accounts; Reserves	
10.4	Distributions After Allowance	43
	ARTICLE XI	
	CONFIRMATION AND CONSUMMATION OF THE PLAN	
11.1	Conditions to Confirmation	
11.2	Conditions to Effective Date	
11.3	Waiver of Conditions Precedent	44
11.4	Consequences of Non-Occurrence of Effective Date	44
	ARTICLE XII	
	EFFECT OF PLAN CONFIRMATION	
12.1	Binding Effect	45
12.2	Releases	45
12.3	Discharge of Claims and Termination of Interests	
12.4	Exculpation and Limitation of Liability	
12.5	Release by Holders of Claims	
12.6	Injunction	47
12.7	Term of Bankruptcy Injunction or Stays	48
	ARTICLE XIII	
	RETENTION OF JURISDICTION	
13.1	Retention of Jurisdiction	48
	ARTICLE XIV	
	MISCELLANEOUS PROVISIONS	
14.1	Effectuating Documents and Further Transactions	50
14.2	Corporate Action	51
14.3	Bar Dates for Certain Claims	

Payment of Statutory Fees	52
Amendment or Modification of the Plan	
Severability of Plan Provisions	
Successors and Assigns	53
<u>~</u>	
Revocation, Withdrawal or Non-Consummation	
Notice	
Governing Law	56
The state of the s	
	Severability of Plan Provisions Successors and Assigns Plan Supplement Revocation, Withdrawal or Non-Consummation

#### INTRODUCTION

VeraSun Energy Corporation, ASA OpCo Holdings, LLC, US BioEnergy Corporation, VeraSun Albert City, LLC, VeraSun Albion, LLC, VeraSun Aurora Corporation, VeraSun BioDiesel, LLC, VeraSun Bloomingburg, LLC, VeraSun Central City, LLC, VeraSun Charles City, LLC, VeraSun Dyersville, LLC, VeraSun Fort Dodge, LLC, VeraSun Granite City, LLC, VeraSun Hankinson, LLC, VeraSun Hartley, LLC, VeraSun Janesville, LLC, VeraSun Linden, LLC, VeraSun Litchfield, LLC, VeraSun Marion, LLC, VeraSun Marketing, LLC, VeraSun Ord, LLC, VeraSun Reynolds, LLC, VeraSun Tilton, LLC, VeraSun Welcome, LLC, and VeraSun Woodbury, LLC (collectively, the "Debtors") debtors and debtors-in-possession in the above-captioned jointly administered chapter 11 reorganization cases, and the Official Committee of Unsecured Creditors appointed in the above-captioned jointly administered chapter 11 reorganization cases pursuant to section 1102 of the Bankruptcy Code (the "Creditors' Committee"), hereby propose the following joint plan of liquidation (the "Plan") for the resolution of the outstanding claims against and interests in the Debtors pursuant to section 1121(a) of the Bankruptcy Code. Reference is made to the Disclosure Statement (as that term is defined herein), distributed contemporaneously herewith, for a discussion of the Debtors' history, business, properties and operations, a summary and analysis of the Plan, and certain related matters including, among other things, the proposed substantive consolidation of certain of the Debtors' cases. Subject to certain restrictions and requirements set forth in section 1127 of the Bankruptcy Code and Fed. R. Bankr. P. 3019, each of (I) the Creditors' Committee and (II) the Debtors reserve the right to alter, amend, modify, revoke or withdraw the Plan (subject to the consent of the Creditors' Committee or the Debtors, as applicable, which consent shall not be unreasonably withheld) prior to the Effective Date (as that term is defined herein).

As set forth herein, the Plan contemplates the substantive consolidation of the VSE Debtors as well as the substantive consolidation of the ASA Debtors. The Plan does not contemplate the substantive consolidation of the USBio Debtors, but rather constitutes an individual Plan with respect to each of the USBio Debtors. Each Debtor and the Creditors' Committee is a proponent of the Plan contained herein within the meaning of section 1129 of the Bankruptcy Code. For voting and Distribution purposes, the Plan contemplates separate classes for the VSE Debtors, the ASA Debtors and each of the USBio Debtors.

#### ARTICLE I

#### DEFINED TERMS AND RULES OF INTERPRETATION

As used herein, capitalized terms shall have the meanings set forth below. Any term that is not otherwise defined herein, but that is used in the Bankruptcy Code or the

Bankruptcy Rules, shall have the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable.

- 1.1 Administrative Claim means a Claim for costs and expenses of administration of the Chapter 11 Cases Allowed under section 503(b) or 507(a)(2) of the Bankruptcy Code, including: (a) any actual and necessary costs and expenses incurred after the Petition Date of preserving the Debtors' Estates and operating the businesses of the Debtors (such as wages, salaries, commissions for services and payments for inventories, leased equipment and premises) and Claims of governmental units for taxes; (b) compensation for legal, financial, advisory, accounting and other services and reimbursement of expenses Allowed by the Bankruptcy Court under section 330, 331 or 503(b) of the Bankruptcy Code to the extent incurred prior to the Effective Date; and (c) all fees and charges assessed against the Debtors' Estates under section 1930, chapter 123 of title 28, United States Code.
- 1.2 Administrative Claims Reserve means the reserve established and maintained by the Plan Administrator, on behalf of the Reorganized Debtors, to pay Administrative Claims, Priority Tax Claims, Non-Tax Priority Claims and Other Secured Claims that become Allowed Claims after the Effective Date.
- 1.3 Administrator Professionals means the agents, financial advisors, attorneys, consultants, independent contractors, representatives and other professionals of the Plan Administrator and the Reorganized Debtors (in their capacities as such).
- 1.4 AgStar Litigation means that certain Complaint for Declaratory Judgment and Other Relief filed by AgStar Financial Services, PCA on December 23, 2008 as adversary case number 08-51897, pursuant to which AgStar Financial Services, PCA claimed that the liens it had on ethanol produced by the USBio Debtors were not released when the USBio Debtors transferred that ethanol to VeraSun Marketing, LLC.
- 1.5 AgStar Settlement means the settlement of the AgStar Litigation entered into by the Debtors and approved by the Bankruptcy Court on August 10, 2009, pursuant to the Order Pursuant to Fed. R. of Bankr. P. 9019 Authorizing the Debtors to Enter Into a Settlement Agreement With AgStar Financial Services, PCA (Docket No. 1680).
  - 1.6 Affiliate Debtors means the Debtors other than VEC.
- 1.7 Allowed... Claim means an Allowed Claim in the Class or category specified. Any reference herein to a particular Allowed Claim includes both the secured and unsecured portions of such Claim.
- 1.8 Allowed Claim means (a) any Claim for which a proof of claim has been timely Filed by the applicable Bar Date (or which pursuant to the Bankruptcy Code or a Final Order is not or shall not be required to be Filed); (b) any Claim that is listed in the Schedules as of the Effective Date as not disputed, not contingent, and not unliquidated, and for which no

proof of claim has been timely Filed; or (c) any Claim allowed pursuant to the Plan; provided, however, that with respect to any Claim in clauses (a) or (b) above, such Claim shall be considered Allowed only if and to the extent that (i) no objection to the allowance thereof has been interposed within the applicable period of time fixed by the Plan, the Bankruptcy Code, the Bankruptcy Rules or the Bankruptcy Court, or (ii) such an objection is so interposed and the Claim shall have been Allowed for Distribution purposes only by a Final Order. Any Claim that has been or is hereafter listed in the Schedules as disputed, contingent, or unliquidated, and for which no proof of claim has been timely Filed is not considered Allowed and shall be expunged without further action by the Reorganized Debtors and without any further notice to or action, order, or approval of the Bankruptcy Court.

- 1.9 ASA Debtors means ASA OpCo Holdings, LLC, ASA Albion, LLC, ASA Bloomingburg, LLC, and ASA Linden, LLC.
- specified Debtor(s) as of the date ten (10) Business Days prior to the date of any Distribution, other than: (a) with respect to Distributions to be made on the Distribution Date, (i) Cash to be distributed on the Distribution Date to Holders of Allowed Prepetition Secured Lender Claims, Allowed Administrative Claims, Allowed Priority Tax Claims, Allowed Other Secured Claims, and Allowed Non-Tax Priority Claims that became Allowed Claims on or before the Effective Date and (ii) Restricted Cash; and (b) with respect to Distributions to be made on any Quarterly Distribution Date, Restricted Cash.
- 1.11 Ballot means each of the ballot form or forms distributed to each Holder of an Impaired Claim, on which the Holder is to indicate acceptance or rejection of the Plan.
- **1.12 Bankruptcy Code** means title 11, United States Code, as now in effect or hereafter amended.
- 1.13 Bankruptcy Court means the United States Bankruptcy Court for the District of Delaware.
- 1.14 Bankruptcy Rules means the Federal Rules of Bankruptcy Procedure and the Local Rules of Bankruptcy Practice and Procedure of the United States Bankruptcy Court for the District of Delaware, as now in effect or hereafter amended.
- 1.15 Bar Date means the applicable deadline by which a proof of claim must have been or must be Filed, as established by an order of the Bankruptcy Court, including the Bar Date Order and the Confirmation Order. The term "Bar Date" also includes the deadline for Filing Fee Claims established pursuant to Article XIV hereof, the deadline for Filing Administrative Claims established pursuant to Article XIV hereof, and the deadline for Filing Claims arising from rejection of executory contracts and unexpired leases established pursuant to Article IX hereof.

- 1.16 Bar Date Order means the Order Under 11 U.S.C. §§ 105 and 502 and Fed. R. Bankr. P. 3003(c)(3) and 9007 (I) Establishing Deadline for Filing Proofs of Claim; (II) Approving Procedures for Filing Proofs of Claim; and (III) Approving the Form, Manner, and Sufficiency of Providing Notice Thereof (Docket No. 970), approved by the Bankruptcy Court on April 2, 2009, as the same may have been or hereafter may be amended, modified or supplemented.
- 1.17 Business Day means any day, other than a Saturday, Sunday or "legal holiday" (as defined in Bankruptcy Rule 9006(a)).
- 1.18 Cash means legal tender of the United States of America and equivalents thereof.
- 1.19 Chapter 11 Case(s) means (a) when used with reference to a particular Debtor, the chapter 11 case pending for that Debtor in the Bankruptcy Court and (b) when used with reference to all Debtors, the chapter 11 cases pending for the Debtors in the Bankruptcy Court.
- 1.20 Claim(s) means a "claim," as defined in section 101(5) of the Bankruptcy Code.
- Claims Objection Deadline means the last day for Filing objections to Claims against the Debtors, which day shall be the later of (a) one hundred eighty (180) days after the Effective Date or (b) sixty (60) days after the Filing of a proof of claim for, or request for payment of, such Claim or such other date as the Bankruptcy Court may order. The Filing of a motion to extend the Claims Objection Deadline shall automatically extend the Claims Objection Deadline until a Final Order is entered on such motion. In the event that such motion to extend the Claims Objection Deadline is denied by the Bankruptcy Court, or approved by the Bankruptcy Court and reversed on appeal, the Claims Objection Deadline shall be the later of the current Claims Objection Deadline (as previously extended, as applicable) or 30 days after entry of a Final Order denying the motion to extend the Claims Objection Deadline.
- 1.22 Class means a category of Holders of Claims or Interests, as described in Article III hereof.
- 1.23 Collateral means any property or interest in property of the Debtors' Estates that is subject to a valid, perfected, and enforceable lien to secure a Claim.
- 1.24 Confirmation Date means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the docket of the Bankruptcy Court for the Chapter 11 Cases.

- 1.25 Confirmation Hearing means the hearing held by the Bankruptcy Court pursuant to section 1128 of the Bankruptcy Code to consider confirmation of the Plan, as such hearing may be adjourned or continued from time to time.
- 1.26 *Confirmation Order* means the order of the Bankruptcy Court confirming the Plan pursuant to section 1129 of the Bankruptcy Code.
- 1.27 *Creditors' Committee* has the meaning ascribed to it in the Introduction hereto.
  - 1.28 **Debtor(s)** has the meaning ascribed to it in the Introduction hereto.
- 1.29 Deficiency Claim means the Claim of a Holder of a Prepetition Secured Lender Claim or Other Secured Claim equal to the amount by which such creditor's Allowed Claim exceeds the value of the Collateral securing such Claim, as determined pursuant to section 506 of the Bankruptcy Code.
- been disallowed by Final Order, (b) is Scheduled at zero or as contingent, disputed, or unliquidated and as to which a Bar Date has been established but no proof of claim has been timely Filed or deemed timely Filed with the Bankruptcy Court pursuant to either the Bankruptcy Code or any Final Order of the Bankruptcy Court or otherwise deemed timely filed under applicable law, or (c) is not Scheduled and as to which a Bar Date has been established but no proof of claim has been timely Filed or deemed timely Filed with the Bankruptcy Court pursuant to either the Bankruptcy Code or any Final Order of the Bankruptcy Court or otherwise deemed timely filed under applicable law.
- 1.31 Disbursing Agent means any party designated by the Creditors' Committee in its sole discretion, after consultation with the Debtors, to serve as disbursing agent under the Plan, and any agent appointed by such Disbursing Agent for the purpose of effectuating such Distributions.
- 1.32 *Disclosure Statement* means the written disclosure statement (including all schedules thereto or referenced therein) in form and substance reasonably acceptable to the Plan Proponents, that relates to the Plan, as approved by the Bankruptcy Court pursuant to section 1125 of the Bankruptcy Code, as the same may be amended, modified or supplemented.
- 1.33 Disputed ... Claim means a Disputed Claim in the Class or category specified.
- 1.34 *Disputed Claim* means any Claim, or any portion thereof, that is neither an Allowed Claim nor a Disallowed Claim.

- set forth in the proof of claim relating to a Disputed Claim, (b) the amount estimated by the Bankruptcy Court for purposes of Distributions in respect of such Disputed Claim in accordance with section 502(c) of the Bankruptcy Code pursuant to Section 8.3 hereof, and (c) the amount of such Disputed Claim Allowed by the Bankruptcy Court pursuant to section 502 of the Bankruptcy Code, or zero, if such Disputed Claim is disallowed in its entirety by the Bankruptcy Court pursuant to such section, in either case, regardless of whether the order or judgment allowing or disallowing such Claim has become a Final Order; provided, however, that in the event a Claim has been disallowed, but the order of disallowance has not yet become a Final Order, the Bankruptcy Court may require the Disbursing Agent to reserve, and hold in trust for the benefit of each Holder of such Claim, Cash in an amount equal to the Pro Rata Distributions which the Bankruptcy Court, in its discretion, determines will protect the rights of such Holder under all of the facts and circumstances relating to the order of disallowance and the appeal of such Holder from such order.
- 1.36 Disputed Claims Reserve means the reserve established and maintained by the Plan Administrator, on behalf of the Reorganized Debtors, to pay Disputed Other Deficiency Claims and Disputed General Unsecured Claims.
- 1.37 *Distribution* means the payment or delivery to any Holder of an Allowed Claim or Interest of the consideration payable to such Holder under the terms of and in accordance with the Plan.
- 1.38 Distribution Date means the date, occurring as soon as possible after the Effective Date (but not later than ten calendar days after the Effective Date, unless extended by the Plan Committee and the Reorganized Debtors), upon which initial Distributions from the Estates are made to Holders of Allowed Claims without further order of the Bankruptcy Court.
- 1.39 Distribution Record Date means (a) for all Holders of Allowed Claims other than Holders of Unsecured Note Claims, the Confirmation Date; and (b) for all Holders of Unsecured Note Claims, the close of business on the Business Day immediately preceding (i) the Distribution Date; and (ii) any other dates(s) Distributions are made to the Holders of Unsecured Note Claims.
- **1.40** Effective Date means the Business Day upon which the Plan becomes effective as provided in Article XI hereof.
- 1.41 Estate(s) means, individually, the estate of any of the Debtors and, collectively, the estates of all of the Debtors created under section 541 of the Bankruptcy Code.
- 1.42 Face Amount means (a) when used in reference to a Disputed or Disallowed Claim, the full stated amount claimed by the Claim Holder in any proof of claim timely filed with the Bankruptcy Court or otherwise deemed timely filed by any Final Order of

the Bankruptcy Court or other applicable bankruptcy law, and (b) when used in reference to an Allowed Claim, the allowed amount of such claim.

- 1.43 Fee Auditor means Warren H. Smith & Associates, P.C.
- 1.44 Fee Claim means an Administrative Claim under section 330(a), 331 or 503 of the Bankruptcy Code for compensation of a Professional or other entity for services rendered or expenses incurred in the Chapter 11 Cases on or prior to the Effective Date (including expenses of the members of the Creditors' Committee incurred as members of the Creditors' Committee in discharge of their duties as such).
- 1.45 *File, Filed or Filing* means file, filed or filing with the Bankruptcy Court or its authorized designee in the Chapter 11 Cases.
- 1.46 Final Order means an order or judgment of the Bankruptcy Court, or other court of competent jurisdiction, as entered on the docket in any of the Chapter 11 Cases, the operation or effect of which has not been stayed, reversed, or amended and as to which order or judgment (or any revision, modification, or amendment thereof) the time to appeal or seek review or rehearing has expired and as to which no appeal or petition for review or rehearing was filed or, if filed, remains pending; provided, however, that the possibility that a motion under section 502(j) of the Bankruptcy Code, Rule 59 or Rule 60 of the Bankruptcy Rules or any analogous rule may be, but has not been filed shall not cause an order not to be a Final Order.
- 1.47 General Unsecured Claim means a Claim that is not an Administrative Claim, Priority Tax Claim, Prepetition Secured Lender Claim, Other Secured Claim, Prepetition Makewhole Claim, Non-Tax Priority Claim, Lender Deficiency Claim, Other Deficiency Claim, or Guaranty Claim. For the avoidance of doubt, General Unsecured Claims include Unsecured Note Claims and Intercompany Claims (other than those Intercompany Claims described in the Intercompany Claims Settlement).
- 1.48 Guaranty Claim means a Claim arising from guaranties under the Prepetition Agstar Credit Agreements.
- 1.49 Holder means the Person holding the beneficial interest in a Claim or Interest.
- 1.50 *Impaired* means, when used in reference to a Claim or Interest, a Claim or Interest that is impaired within the meaning of section 1124 of the Bankruptcy Code.
- 1.51 *Intercompany Claim* means any Claim held by a Debtor against any other Debtor, including any intercompany book entries reflecting obligations owed by one Debtor with respect to any other Debtor.

- 1.52 Intercompany Claims Settlement means the agreement between VeraSun Marketing, LLC and the USBio Debtors to settle all intercompany claims between VeraSun Marketing, LLC and the USBio Debtors pursuant to which the VSE Debtors shall pay the Allowed USBio Intercompany Administrative Claim in full.
- 1.53 Interest means the legal, equitable, contractual and other rights of the Holders of Old Equity, including the rights of any entity to purchase or demand the issuance of any of the foregoing, including (a) conversion, exchange, voting, participation and dividend rights; (b) liquidation preferences; (c) stock options, warrants and put rights; and (d) share-appreciation rights.
- 1.54 Internal Revenue Code means the Internal Revenue Code of 1986, as amended.
- 1.55 Lender Deficiency Claim means the Deficiency Claim of a Holder of a Prepetition Secured Lender Claim.
- 1.56 Litigation Claims means the Claims, rights of action, suits or proceedings, whether in law or in equity, whether known or unknown, that any Debtor or Estate may hold against any Person, including but not limited to the Claims, rights of action, suits and proceedings listed in the Plan Supplement, to be retained by the Estates pursuant to Section 7.7 of the Plan including, but not limited to, (i) any claims, rights of action, suits or proceedings arising under or resulting from contractual subordination or section 510(b) of the Bankruptcy Code or (ii) any claims under sections 544, 545, 547 and 548 of the Bankruptcy Code that may exist.
- 1.57 Litigation Reserve means the reserve account to be established and maintained by the Plan Administrator, on behalf of the Reorganized Debtors, into which the Reorganized Debtors shall from time to time deposit Cash (in such amounts and at such times as directed by the Plan Committee) to fund the costs and expenses associated with investigating and, as necessary, prosecuting, settling, dismissing or otherwise disposing of any or all of the Litigation Claims; provided, however, that at no time shall the amount of the Litigation Claims Reserve exceed \$100,000; provided, further, that the Plan Administrator may at any time increase or decrease the amount of the Litigation Claims Reserve with the consent of the Plan Committee.
- 1.58 Non-Tax Priority Claim means a Claim, other than an Administrative Claim or Priority Tax Claim, that is entitled to priority in payment pursuant to section 507(a) of the Bankruptcy Code.
  - 1.59 Non-VSE Debtors means the Debtors other than the VSE Debtors.
- 1.60 Old Common Stock means the common stock of VEC outstanding immediately prior to the Petition Date, including (i) treasury stock and all options, warrants, calls,

rights, puts, awards, commitments or any other agreements of any character to acquire such common stock; and (ii) claims arising from rescission of a purchase or sale of the common stock of VEC, for damages arising from the purchase or sale of the common stock of VEC, or for reimbursement or contribution allowed under section 502 of the Bankruptcy Code on account of such a claim, which claims are subordinated pursuant to section 510(b) of the Bankruptcy Code.

- 1.61 Old Equity means, collectively, the Old Common Stock and the Subsidiary Interests.
- 1.62 Operating Reserve means the reserve account to be established and maintained by the Plan Administrator, on behalf of the Reorganized Debtors, into which the Reorganized Debtors shall from time to time deposit Cash (in such amounts and at such times as directed by the Plan Committee, as the case may be) to fund, among other things, the expenses of the Plan Administrator, the Reorganized Debtors and the Plan Committee and its professionals, as set forth more fully in the Plan Administrator Agreement; provided, however, that the Plan Committee shall not unreasonably refuse to fund the Operating Reserve in an amount sufficient to pay the reasonable fees and expenses of the Plan Administrator; provided, further, that if the Plan Committee should refuse to fund the Operating Reserve in an amount sufficient to pay the reasonable fees and expenses of the Plan Administrator, the Plan Administrator shall have the right to file an appropriate motion with the Bankruptcy Court requesting that the Court direct that the Operating Reserve be funded in an amount sufficient to pay the reasonable fees and expenses of the Plan Administrator.
- 1.63 Other Deficiency Claim means the Deficiency Claim of a Holder of an Other Secured Claim.
- Lender Claim) that is secured by a valid, perfected and enforceable lien on property in which a Debtor's Estate has an interest or that is subject to setoff under section 553 of the Bankruptcy Code, to the extent of the value of the Claim Holder's interest in the applicable Estate's interest in such property or to the extent of the amount subject to setoff, as applicable, as determined pursuant to section 506(a) of the Bankruptcy Code or, in the case of the setoff, pursuant to section 553 of the Bankruptcy Code.
- 1.65 *Person* means an individual, corporation, partnership, joint venture, association, joint stock company, limited liability company, limited liability partnership, trust, estate, unincorporated organization, or other entity.
- 1.66 *Petition Date* means October 31, 2008, the date on which the Debtors Filed the petitions for relief commencing the Chapter 11 Cases.
- 1.67 *Plan* means this chapter 11 plan of liquidation, including the Plan Supplement and all supplements, appendices and schedules thereto, either in its present form or

as the same may be altered, amended or modified from time to time; *provided*, that each of the documents referenced in this definition shall be in form and substance reasonably acceptable to the Debtors, the Reorganized Debtors, the Creditors' Committee or the Plan Committee, as the case may be.

- 1.68 *Plan Administrator* means the person designated by the Creditors' Committee in its sole discretion, after consultation with the Debtors, prior to the Confirmation Date and approved by the Bankruptcy Court pursuant to the Confirmation Order to administer the Plan in accordance with the terms of the Plan and the Plan Administrator Agreement and to take such other actions as may be authorized under the Plan Administrator Agreement, and any successor thereto.
- 1.69 Plan Administrator Agreement means the agreement between and among the Debtors and the Plan Administrator, in form and substance acceptable to the Creditors' Committee in its sole discretion, specifying the rights, duties and responsibilities of and to be performed by the Plan Administrator under the Plan, in substantially the form set forth in the Plan Supplement; provided, that after the Effective Date, any modifications made to the Plan Administrator Agreement shall be in form and substance acceptable to the Plan Committee.
- 1.70 Plan Committee means the committee established pursuant to Section 7.8 of the Plan to supervise and direct the Plan Administrator, to monitor implementation of the Plan, and to take such other actions and have such other rights as are set forth in the Plan, including, without limitation, as set forth in Section 7.8 of the Plan and Article X of the Plan, or as may be approved by the Bankruptcy Court.
- 1.71 *Plan Proponents* means, individually, either the Creditors' Committee or the Debtors, as applicable, and, collectively, the Creditors' Committee and the Debtors.
- 1.72 Plan Supplement means the forms of documents specified in Section 14.8 of the Plan, which documents shall be in form and substance reasonably acceptable to the Plan Proponents, subject to sections 7.3 and 7.5 herein; provided, that after the Effective Date, any modifications made to any of the documents specified in Section 14.8 of the Plan shall be in form and substance reasonably acceptable to the Reorganized Debtors and the Plan Committee, subject to sections 7.3 and 7.5 herein.
  - 1.73 **Precluded Claims** means the claims defined in Section 12.4 of the Plan.
- 1.74 Prepetition Agstar Credit Agreements means the Prepetition Janesville Credit Agreement, the Prepetition Albert City Credit Agreement, the Prepetition Central City Credit Agreement, the Prepetition Dyersville Credit Agreement, the Prepetition Hankinson Credit Agreement, the Prepetition Ord Credit Agreement, and the Prepetition Woodbury Credit Agreement.

- 1.75 Prepetition Albert City Credit Agreement means the Master Loan Agreement dated as of November 15, 2005, by and among the VeraSun Albert City, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.76 Prepetition Central City Credit Agreement means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Central City, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.77 Prepetition Credit Agreements means the Prepetition Dougherty/Marion Credit Agreement, the Prepetition First Bank Letter of Credit, the Prepetition WestLB/ASA Credit Agreement, and the Prepetition Agstar Credit Agreements.
- 1.78 Prepetition Dougherty/Marion Credit Agreements means (i) that certain Loan and Security Agreement, dated as of June 22, 2006, by and between Millennium Ethanol, LLC, a South Dakota limited liability company, and Dougherty Funding LLC, a Delaware limited liability company, as amended, restated, supplemented or otherwise modified from time to time; and (ii) that certain Revolving Credit and Security Agreement, dated as of June 29, 2006, by and between Millennium Ethanol, LLC, a South Dakota limited liability company, and Dougherty Funding LLC, a South Dakota state bank, as amended, restated, supplemented or otherwise modified from time to time.
- 1.79 Prepetition Dyersville Credit Agreement means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Dyersville, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.80 Prepetition First Bank Letter of Credit means that certain Letter of Credit Application, Reimbursement and Security Agreement, dated December 18, 2006, by and between Millennium Ethanol, LLC, a South Dakota limited liability company, and First Bank & Trust, a South Dakota state bank, as amended, restated, supplemented or otherwise modified from time to time.
- 1.81 Prepetition Hankinson Credit Agreement means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Hankinson, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.82 *Prepetition Janesville Credit Agreement* means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Janesville, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.83 Prepetition Ord Credit Agreement means the Credit Agreement dated as of February 7, 2007, by and among VeraSun Ord, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.

- 1.84 Prepetition Makewhole Claim means any Claim for any makewhole amount, prepayment premium, early termination fee, change of control premium or other similar amount asserted to be payable in respect of the VSE Secured Notes.
- 1.85 Prepetition Makewhole Claims Reserve means the reserve maintained by the Plan Administrator, on behalf of the Reorganized Debtors, to pay Prepetition Makewhole Claims that become Allowed Claims after the Effective Date.
- 1.86 Prepetition Secured Lender Claim(s) means a Claim against a Debtor arising out of the Prepetition Credit Agreements that is secured by a valid, perfected and enforceable lien on property in which a Debtor's Estate has an interest, to the extent of the value of the Claim Holder's interest in the applicable Estate's interest in such property, as determined pursuant to Section 506(a) of the Bankruptcy Code.
- 1.87 Prepetition WestLB/ASA Credit Agreement means the Credit Agreement, dated as of February 6, 2006 (as amended, restated, supplemented or otherwise modified from time to time), among the ASA Debtors, as borrowers, ASA OpCo Holdings, LLC, as borrowers' agent, the lenders party thereto, WestLB AG, New York Branch, as administrative agent, First National Bank of Omaha as collateral agent and as the accounts bank, WestLB AG, New York Branch, as co-syndication agent, lead arranger, and sole lead bookrunner, First National Bank of Omaha and Standard Chartered Bank, as co-syndication agents and lead arrangers and CIT Capital USA INC. and ING Capital LLC, as co-documentation agents and lead arrangers.
- 1.88 Prepetition Woodbury Credit Agreement means the Master Loan Agreement dated as of November 15, 2005, by and among VeraSun Woodbury, LLC, the lenders party thereto, and AgStar Financial Services, PCA, as administrative agent.
- 1.89 **Priority Tax Claim** means a Claim of a governmental unit of the kind specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code.
- 1.90 Professional means (a) any professional employed in the Chapter 11 Cases pursuant to section 327 or 1103 of the Bankruptcy Code or otherwise and (b) any professional or other entity seeking compensation or reimbursement of expenses in connection with the Chapter 11 Cases pursuant to section 503(b)(4) of the Bankruptcy Code.
- 1.91 *Pro Rata* means, with respect to Claims, at any time, the proportion of the Face Amount of a Claim in a particular Class or Classes bears to the aggregate Face Amount of all Claims (including Disputed Claims, but excluding Disallowed Claims) in such Class or Classes, unless the Plan provides otherwise.
- 1.92 Quarter means the period beginning on the Effective Date and ending on the next June 30, September 30, December 31 and March 31, and each three-month period thereafter.

- 1.93 Quarterly Distribution Date means the last Business Day of the month following the end of each calendar quarter after the Effective Date; provided, however, that if the Effective Date is within 30 days of the end of a calendar quarter, the first Quarterly Distribution Date shall be the last Business Day of the month following the end of the first calendar quarter after the calendar quarter in which the Effective Date falls.
- 1.94 Reorganized means the applicable Debtor from and after the Effective Date.
- 1.95 Reserves means, collectively, the Administrative Claims Reserve, Disputed Claims Reserve, the Litigation Claims Reserve, the Operating Reserve, the Prepetition Makewhole Claims Reserve, and the Unclaimed Distribution Reserve.
- 1.96 Restricted Cash means the Cash segregated (whether physically or merely on the books and records of the Reorganized Debtors) by the Plan Administrator, on behalf of the Reorganized Debtors, to fund the Reserves.
- 1.97 Scheduled means with respect to any Claim or Interest, the status and amount, if any, of such Claim or Interest as set forth in the Schedules.
- 1.98 Schedules means the schedules of assets and liabilities, the list of Holders of Interests and the statements of financial affairs Filed by the Debtors on January 14, 2009, as such schedules have been or may be further modified, amended or supplemented in accordance with Fed. R. Bankr. P. 1009 or orders of the Bankruptcy Court.
- 1.99 Solicitation Order means the Order Under Bankruptcy Code Sections 105, 502, 1125 and 1126, Bankruptcy Rules 2002, 3003, 3017, 3018, 3020 and 9007 and Local Bankruptcy Rule 3017-1 (I) Approving the Form and Manner of Notice of the Disclosure Statement Hearing; (II) Determining Adequacy of the Disclosure Statement; (III) Establishing a Record Date and Voting, Objection and Other Deadlines With Respect to Confirmation of Plan; (IV) Scheduling Hearing on Confirmation of Plan; (V) Approving (A) the Form and Manner of Notice of the Hearing on Confirmation and Related Matters and (B) the Deadline for Filing Confirmation Objections; (VI) Approving the Solicitation Procedures; and (VII) Granting Certain Other Relief (Docket No. 1797), approved by the Bankruptcy Court on September 10, 2009, as the same may have been or hereafter may be amended, modified or supplemented; provided, that the Solicitation Order shall be in form and substance reasonably acceptable to the Plan Proponents.
- 1.100 Subsidiary Interests means, collectively, the issued and outstanding shares of stock of the Affiliate Debtors, as of the Petition Date.
- 1.101 Substantive Consolidation Order means the order, or provision of the Confirmation Order, substantively consolidating (a) the Chapter 11 Cases of the VSE Debtors as one consolidated group of Debtors and (b) the Chapter 11 Cases of the ASA Debtors as another

consolidated group of Debtors, as provided in Sections 3.1 and 7.1 of the Plan; *provided*, that the Substantive Consolidation Order shall be in form and substance reasonably acceptable to the Plan Proponents.

- 1.102 Unclaimed Distribution Reserve means the reserve established pursuant to Section 8.4(b)(i) of the Plan.
- 1.103 *Unimpaired* means, when used in reference to a Claim or Interest, a Claim or Interest that is not impaired within the meaning of section 1124 of the Bankruptcy Code.
- 1.104 Unsecured Note Claim means individually, a Claim of a Holder of the Unsecured Notes, and collectively, the Claims of all such Holders arising under or as a result of such Unsecured Notes.
- 1.105 Unsecured Note Indenture means that certain Indenture, dated as of May 16, 2007, as supplemented by the First Supplemental Indenture, dated as of December 31, 2007, among VEC, as Issuer, each of the Subsidiary Guarantors named therein, and Wells Fargo Bank, N.A., as the original Trustee, pursuant to which \$437,000,000 aggregate principal amount of 9\% Senior Exchange Notes due 2017 were issued by VEC.
- Association, as successor in interest to Wells Fargo Bank, N.A., in its capacity as indenture trustee under the Unsecured Note Indenture. Any reference to a Distribution or other payment being made "to the Unsecured Note Indenture Trustee" shall mean a Distribution or payment "to the Unsecured Note Indenture Trustee, for the benefit of Holders of Unsecured Notes (except to the extent the Distribution or other payment is in payment of the Unsecured Note Indenture Trustee Fees and Expenses)."
- 1.107 Unsecured Note Indenture Trustee Fees and Expenses means any and all fees, expenses, disbursements and advances of the Unsecured Note Indenture Trustee (and its counsel, agents and advisors) that are provided for under the Unsecured Note Indenture (including, without limitation, in connection with service on the Creditors' Committee, the Plan Committee (if applicable) and in connection with Distributions under the Plan), which are incurred at any time prior to or after the Effective Date.
- 1.108 *Unsecured Notes* means the 9%% Senior Exchange Notes issued by VEC, under the Unsecured Note Indenture.
- 1.109 USBio Debtors means US BioEnergy Corporation, VeraSun Albert City, LLC, VeraSun Central City, LLC, VeraSun Dyersville, LLC, VeraSun Hankinson, LLC, VeraSun Janesville, LLC, VeraSun Ord, LLC, VeraSun Woodbury, LLC, and US Bio Marion, LLC.

- 1.110 USBio Intercompany Administrative Claim means the aggregate Allowed Administrative Claim of all the USBio Debtors against the VSE Debtors in the amount of \$1.5 million.
  - **1.111 VEC** means VeraSun Energy Corporation.
- 1.112 *VSE Debtors* means VEC, VeraSun Granite City, LLC, VeraSun Reynolds, LLC, VeraSun BioDiesel, LLC, VeraSun Litchfield, LLC, VeraSun Tilton, LLC, VeraSun Aurora Corporation, VeraSun Charles City, LLC, VeraSun Marketing, LLC, VeraSun Welcome, LLC, VeraSun Fort Dodge, LLC, and VeraSun Hartley, LLC.
- 1.113 *VSE Secured Notes* means the 9 7/8% million Secured Notes Due 2012 issued by VEC and guaranteed by the VSE Debtors other than VEC.
- 1.114 Voting Deadline means the date set by the Bankruptcy Court pursuant to the Solicitation Order by which all Ballots to accept or reject the Plan in accordance with section 1126 of the Bankruptcy Code must be received by the Debtors or their Bankruptcy Court appointed agent.

Rules of Interpretation and Computation of Time. For purposes of the Plan, unless otherwise provided herein: (a) whenever from the context it is appropriate, each term, whether stated in the singular or the plural, will include both the singular and the plural; (b) unless otherwise provided in the Plan, any reference in the Plan to a contract, instrument, release or other agreement or document being in a particular form or on particular terms and conditions means that such document will be substantially in such form or substantially on such terms and conditions; (c) any reference in the Plan to an existing document or schedule Filed or to be Filed means such document or schedule, as it may have been or may be amended, modified or supplemented pursuant to the Plan; (d) any reference to an entity as a Holder of a Claim or Interest includes that entity's successors and assigns; (e) all references in the Plan to Sections, Articles and Schedules are references to Sections, Articles and Schedules of or to the Plan; (f) the words "herein," "hereunder" and "hereto" refer to the Plan in its entirety rather than to a particular portion of the Plan; (g) captions and headings to Articles and Sections are inserted for convenience of reference only and are not intended to be a part of or to affect the interpretation of the Plan; (h) subject to the provisions of any contract, certificates of incorporation, by-laws, instrument, release or other agreement or document entered into in connection with the Plan, the rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, federal law, including the Bankruptcy Code and Bankruptcy Rules; (i) the rules of construction set forth in section 102 of the Bankruptcy Code will apply; and (j) in computing any period of time prescribed or allowed by the Plan, the provisions of Bankruptcy Rule 9006(a) will apply.

#### ARTICLE II

#### ADMINISTRATIVE EXPENSES AND PRIORITY TAX CLAIMS

- Administrative Claims Except as otherwise provided for herein, and subject to the requirements of Article XII and Section 14.3 hereof, on, or as soon as reasonably practicable after (i) the Distribution Date, if such Administrative Claim is an Allowed Administrative Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Administrative Claim becomes an Allowed Administrative Claim, each Holder of an Allowed Administrative Claim shall receive in full satisfaction, settlement and release of and in exchange for such Allowed Administrative Claim (A) Cash of the Debtor or consolidated group of Debtors against which such Administrative Claim is Allowed equal to the unpaid portion of such Allowed Administrative Claim or (B) such less favorable treatment as to which such Debtor, Reorganized Debtor, or consolidated group of Debtors or Reorganized Debtors (subject to the consent of the Plan Committee, which consent shall not be unreasonably withheld) and the Holder of such Allowed Administrative Claim shall have agreed upon in writing; provided, however, that Allowed Administrative Claims with respect to liabilities incurred by the Debtors or consolidated group of Debtors in the ordinary course of business during the Chapter 11 Cases shall be paid in the ordinary course of business in accordance with the terms and conditions of any agreements relating thereto. For the avoidance of doubt, the payment of Professional Fee Claims shall be governed by Section 14.3 herein.
- 2.2 **Priority Tax Claims.** On, or as soon as reasonably practicable after (i) the Distribution Date, if such Priority Tax Claim is an Allowed Priority Tax Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Priority Tax Claim becomes an Allowed Priority Tax Claim, each Holder of an Allowed Priority Tax Claim shall receive in full satisfaction, settlement and release of and in exchange for such Allowed Priority Tax Claim (A) Cash of the Debtor or consolidated group of Debtors against which such Priority Tax Claim is Allowed equal to the amount of such Allowed Priority Tax Claim, (B) such less favorable treatment as to which such Debtor, Reorganized Debtor, or consolidated group of Debtors or Reorganized Debtors (subject to the consent of the Plan Committee, which consent shall not be unreasonably withheld) and the Holder of such Allowed Priority Tax Claims shall have agreed upon in writing; provided, however, that any Priority Tax Claim not due and owing on the Effective Date will be paid in accordance with this Section when such Claim becomes due and owing; or (C) at the option of the Reorganized Debtors and the Plan Committee, Cash of the Debtor or consolidated group of Debtors in an aggregate amount of such Allowed Priority Tax Claim payable in installment payments over a period of not more than five years after the Petition Date, pursuant to section 1129(a)(9)(c) of the Bankruptcy Code.

#### ARTICLE III

#### CLASSIFICATION OF CLAIMS AND INTERESTS

3.1 The Debtors. There are a total of 25 Debtors. The VSE Debtors and the ASA Debtors shall be substantively consolidated for Plan voting and Distribution purposes as described in Article VII. The USBio Debtors shall not be substantively consolidated and this Plan is a separate Plan for each USBio Debtor. Each Debtor or group of consolidated Debtors has been assigned a number below for the purposes of classifying and treating Claims against and Interests in each Debtor or consolidated group of Debtors for balloting purposes. The Claims against and Interests in each Debtor or consolidated group of Debtors, in turn, have been assigned to separate lettered Classes with respect to each Debtor or consolidated group of Debtors, based on the type of Claim involved. Accordingly, the classification of any particular Claim against or Interest in any Debtor or consolidated group of Debtors depends on the particular Debtor against which such Claim is asserted (or in which such Interest is held) and the type of Claim or Interest in question. The number will denote which Debtor or consolidated group of Debtors against which a Claim or Interest has been asserted, and the letter will denote the Class of such Claim or Interest. The numbers applicable to the various Debtor groups are as follows:

<u>Number</u>	Consolidated Debtor Group or Debtor Name
1	VSE Debtors
2	ASA Debtors
3	US BioEnergy Corporation
4	VeraSun Albert City, LLC
5	VeraSun Central City, LLC
6	VeraSun Dyersville, LLC
7	VeraSun Hankinson, LLC
8	VeraSun Janesville, LLC
9	VeraSun Ord, LLC

- 10 VeraSun Woodbury, LLC
- 11 US Bio Marion, LLC

#### 3.2 Classification of Claims and Interests.

- (a) Pursuant to section 1122 of the Bankruptcy Code, set forth below is a designation of classes of Claims against and Interests in each of the Debtors. A Claim or Interest is placed in a particular Class for the purposes of voting on the Plan and of receiving Distributions pursuant to the Plan only to the extent that such Claim or Interest is an Allowed Claim or an Allowed Interest in that Class and such Claim or Interest has not been paid, released, or otherwise settled prior to the Effective Date. In accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Claims and Priority Tax Claims have not been classified and their treatment is set forth in Article II above.
- (b) Claims against and Interests in each of the Debtors are divided into lettered Classes. Not all of the Classes apply to every Debtor, and consequently not all of the lettered Classes appear in the case of each Debtor or group of consolidated Debtors. For purposes of voting, Claims within the Class shall be counted for each applicable Debtor or group of consolidated Debtors. Whenever such a Class of Claims or Equity Interests is relevant to a particular Debtor, that class of Claims or Interests shall be grouped under the appropriate lettered Class from the following list:
  - Class A Class A consists of Prepetition Secured Lender Claims against the applicable Debtor or consolidated group of Debtors
  - Class B consists of Other Secured Claims against the applicable Debtor or consolidated group of Debtors
  - Class C Class C consists of Prepetition Makewhole Claims against the applicable Debtor or consolidated group of Debtors
  - Class D Class D consists of all Non-Tax Priority Claims against the applicable Debtor or consolidated group of Debtors
  - Class E consists of all Lender Deficiency Claims against the applicable Debtor or consolidated group of Debtors
  - Class F Class F consists of all Other Deficiency Claims against the

applicable Debtor or consolidated group of Debtors

Class G consists of all General Unsecured Claims against the applicable Debtor or consolidated group of Debtors

Class H Class H consists of all Guaranty Claims against the applicable Debtor or consolidated group of Debtors

Class I Class I consists of all Interests in the applicable Debtor or consolidated group of Debtors

#### ARTICLE IV

# IDENTIFICATION OF CLASSES OF CLAIMS AND INTERESTS IMPAIRED AND UNIMPAIRED BY THE PLAN

4.1 Classes of Claims that are Unimpaired. The following Classes of Claims are Unimpaired by the Plan:

Classes 2A and 4A through (Prepetition Secured Lender

11A Claims against Non-VSE Debtors

other than US Bio Energy

Corporation)<sup>1</sup>

Class 1B through 11B (Other Secured Claims)

Class 1C (Prepetition Makewhole Claims

against the VSE Debtors)

Class 1D through 11D (Non-Tax Priority Claims)

**4.2** Impaired Classes of Claims and Interests. The following Classes of Claims and Interests are Impaired by the Plan:

Classes 2E and 4E through (Lender Deficiency Claims against

11E Non-VSE Debtors other than US

Bio Energy Corporation)<sup>2</sup>

Classes 1F through 11F (Other Deficiency Claims)

<sup>&</sup>lt;sup>1</sup> There is no Class 3A because there are no Prepetition Secured Lender Claims against US BioEnergy Corporation.

<sup>&</sup>lt;sup>2</sup> There is no Class 3E because there are no Lender Deficiency Claims against US BioEnergy Corporation.

Classes 1G through 11G

(General Unsecured Claims)

Class 3H

(Guaranty Claims against US Bio

Energy Corporation)

Classes 11 through 111

(Interests)

#### ARTICLE V

#### PROVISIONS FOR TREATMENT OF CLAIMS AND INTERESTS

#### 5.1 Unimpaired Classes of Claims.

- (a) <u>Class 2A (Prepetition Secured Lender Claims against the ASA</u>

  <u>Debtors)</u>. On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against the ASA Debtors shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing the obligations of the ASA Debtors under the Prepetition WestLB/ASA Credit Agreement. Any Deficiency Claim against the ASA Debtors that is ultimately Allowed shall be a Class 2E Claim.
- (b) Class 4A (Prepetition Secured Lender Claims against VeraSun Albert City, LLC). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Albert City, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing the obligations under the Prepetition Albert City Credit Agreement. Any Deficiency Claim against VeraSun Albert City, LLC that is ultimately Allowed shall be a Class 4E Claim.
- (c) <u>Class 5A (Prepetition Secured Lender Claims against VeraSun Central City, LLC)</u>. On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Central City, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Central City, LLC's obligations under the Prepetition Central City Credit Agreement. Any Deficiency Claim against VeraSun Central City, LLC that is ultimately Allowed shall be a Class 5E Claim.
- (d) Class 6A (Prepetition Secured Lender Claims against VeraSun Dyersville, LLC). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Dyersville, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro

Rata share of the proceeds of the Collateral, if any, securing VeraSun Dyersville, LLC's obligations under the Prepetition Dyersville Credit Agreement. Any Deficiency Claim against Verasun Dyersville, LLC that is ultimately Allowed shall be a Class 6E Claim.

- (e) <u>Class 7A (Prepetition Secured Lender Claims against VeraSun Hankinson, LLC)</u>. On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Hankinson, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Hankinson, LLC's obligations under the Prepetition Hankinson Credit Agreement. Any Deficiency Claim against VeraSun Hanksinson, LLC that is ultimately Allowed shall be a Class 7E Claim.
- Janesville, LLC). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Janesville, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Janesville, LLC's obligations under the Prepetition Janesville Credit Agreement. Any Deficiency Claim against VeraSun Janesville, LLC that is ultimately Allowed shall be a Class 8E Claim.
- (g) Class 9A (Prepetition Secured Lender Claims against VeraSun Ord, LLC). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Ord, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Ord, LLC's obligations under the Prepetition Ord Credit Agreement. Any Deficiency Claim against VeraSun Ord, LLC that is ultimately Allowed shall be a Class 9E Claim.
- (h) Class 10A (Prepetition Secured Lender Claims against VeraSun Woodbury, LLC). On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against VeraSun Woodbury, LLC shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing VeraSun Woodbury, LLC's obligations under the Prepetition Woodbury Credit Agreement. Any Deficiency Claim against VeraSun Woodbury, LLC that is ultimately Allowed shall be a Class 10E Claim.
- (i) <u>Class 11A (Prepetition Secured Lender Claims against US Bio</u>

  <u>Marion, LLC)</u>. On, or as soon as reasonably practicable after the Distribution Date, each Holder of an Allowed Prepetition Secured Lender Claim against US Bio Marion, LLC shall receive, to

the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Secured Lender Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing the obligations of US Bio Marion, LLC under the Prepetition Dougherty/Marion Credit Agreements. Any Deficiency Claim US Bio Marion, LLC that is ultimately Allowed shall be a Class 11E Claim.

- (j) Class 1B through 11B (Other Secured Claims). On, or as soon as reasonably practicable after (i) the Distribution Date, if such Other Secured Claim is an Allowed Other Secured Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Other Secured Claim becomes an Allowed Other Secured Claim, the Holder of an Allowed Other Secured Claim shall receive, to the extent not previously received, in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Other Secured Claim, its Pro Rata share of the proceeds of the Collateral, if any, securing such Other Secured Claim in an amount up to the amount of such Allowed Other Secured Claim.
- (k) Class 1C (Prepetition Makewhole Claims against the VSE Debtors). On, or as soon as reasonably practicable after (i) the Distribution Date, if such Prepetition Makewhole Claim is an Allowed Prepetition Makewhole Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Prepetition Makewhole Claim becomes an Allowed Prepetition Makewhole Claim, each Holder of an Allowed Prepetition Makewhole Claim against the VSE Debtors shall receive in full satisfaction, settlement, release and compromise of and in exchange for such Allowed Prepetition Makewhole Claim its Pro Rata share of the proceeds of the Collateral, if any, securing the obligations of the VSE Debtors under the VSE Secured Notes.
- (l) Class 1D through 11D (Non-Tax Priority Claims). On, or as soon as reasonably practicable after (i) the Distribution Date, if such Non-Tax Priority Claim is an Allowed Non-Tax Priority Claim as of the Effective Date, or (ii) the first Quarterly Distribution Date after the date such Non-Tax Priority Claim becomes an Allowed Non-Tax Priority Claim, each Holder of an Allowed Non-Tax Priority Claim shall receive, to the extent not previously received, in full satisfaction, settlement, compromise and release of and in exchange for such Allowed Non-Tax Priority Claim (A) Cash equal to the amount of such Allowed Non-Tax Priority Claim or (B) such less favorable treatment as to which the Debtors or Reorganized Debtors (subject to the consent of the Plan Committee) and the Holder of such Allowed Non-Tax Priority Claim have agreed upon in writing.

#### 5.2 Impaired Classes of Claims.

(a) <u>Classes 2E and 4E through 11E (Lender Deficiency Claims against Non-VSE Debtors other than US Bio Energy Corporation)</u>. On, or as soon as reasonably practicable after, (i) the Distribution Date if such Lender Deficiency Claim is an Allowed Lender Deficiency Claim as of the Effective Date or (ii) the first Quarterly Distribution Date after the date a Lender Deficiency Claim becomes an Allowed Lender Deficiency Claim, each Holder of

an Allowed Lender Deficiency Claim, together with Holders of Allowed Other Deficiency Claims and Allowed General Unsecured Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed. On each ensuing Quarterly Distribution Date, each Holder of an Allowed Lender Deficiency Claim, along with Holders of Allowed Other Deficiency Claims and Allowed General Unsecured Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed.

- as reasonably practicable after, (i) the Distribution Date if such Other Deficiency Claim is an Allowed Other Deficiency Claim as of the Effective Date or (ii) the first Quarterly Distribution Date after the date that an Other Deficiency Claim becomes an Allowed Other Deficiency Claim, each Holder of an Allowed Other Deficiency Claim, together with Holders of Allowed Lender Deficiency Claims, Allowed General Unsecured Claims, and Guaranty Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed. On each ensuing Quarterly Distribution Date, each Holder of an Allowed Other Deficiency Claims, along with Holders of Allowed Lender Deficiency Claims, Allowed General Unsecured Claims, and Allowed Guaranty Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed.
- (c) <u>Classes 1G-11G (General Unsecured Claims</u>). On, or as soon as reasonably practicable after, (i) the Distribution Date if such General Unsecured Claim is an Allowed General Unsecured Claim as of the Effective Date or (ii) the first Quarterly Distribution Date after the date a General Unsecured Claim becomes an Allowed General Unsecured Claim, each Holder of an Allowed General Unsecured Claim, together with Holders of Allowed Lender Deficiency Claims, Allowed Other Deficiency Claims, and Allowed Guaranty Claims, shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed. On each ensuing Quarterly Distribution Date, each Holder of an Allowed General Unsecured Claim, along with Holders of Allowed Lender Deficiency Claims, Allowed General Unsecured Claims, and Allowed Guaranty Claims shall receive its Pro Rata share of Available Cash, if any, of the Debtor or consolidated group of Debtors against which the Claim is Allowed.

Notwithstanding the foregoing, the Reorganized Debtor shall pay, on or as soon as reasonably practicable after the Effective Date, all reasonable Unsecured Note Indenture Trustee Fees and Expenses, in full in Cash, without application to or approval of the Bankruptcy Court and without a reduction to the recoveries of the Holders of Unsecured Notes. Notwithstanding the foregoing, to the extent any Unsecured Note Indenture Trustee Fees and Expenses are not paid (including, without limitation, any fees or expenses incurred in connection with any unresolved litigation relating to Disputed Claims), the Unsecured Note Indenture Trustee may assert its charging lien against any recoveries received on behalf of its Holders for payment of such unpaid amounts.

- (d) Class 3H (Guaranty Claims against US Bio Energy Corporation). On, or as soon as reasonably practicable after, (i) the Distribution Date if such Guaranty Claim is an Allowed Guaranty Claim as of the Effective Date or (ii) the first Quarterly Distribution Date after the date a Guaranty Claim becomes an Allowed Guaranty Claim, each Holder of an Allowed Guaranty Claim, together with Holders of Allowed Other Deficiency Claims and Allowed General Unsecured Claims against US Bio Energy Corporation, shall receive its Pro Rata share of Available Cash, if any, of US Bio Energy Corporation. On each ensuing Quarterly Distribution Date, each Holder of an Allowed Guaranty Claim, along with Holders of Other Deficiency Claims and General Unsecured Claims against US Bio Energy Corporation, shall receive its Pro Rata share of Available Cash of US Bio Energy Corporation.
- 5.3 Impaired Classes of Interests. Class II-1II (Interests). All Interests, and all stock certificates, instruments, and other documents evidencing such Interests, shall be cancelled as of the Effective Date. The Holders of Interests in Class II-1II shall not receive or retain any Distribution or other property on account of such Interests.

#### ARTICLE VI

#### ACCEPTANCE OR REJECTION OF THE PLAN

- 6.1 Impaired Classes of Claims Entitled to Vote. Each Impaired Class of Claims that will (or may) receive or retain property or any interest in property under the Plan shall be entitled to vote to accept or reject the Plan; provided, however, that the votes of Holders of Intercompany Claims shall not be solicited, and any Ballots submitted by such Holders shall not be counted.
- 6.2 Classes Deemed to Accept the Plan. Class 2A, Classes 4A through 11A, Classes 1B through 11B, Class 1C and Classes 1D through 11D are Unimpaired under the Plan. Pursuant to section 1126(f) of the Bankruptcy Code, such Classes are conclusively presumed to have accepted the Plan, and the votes of Holders of Claims in such Classes therefore shall not be solicited.
- Classes 1F through 11F, Classes 1G through 11G, and Class 3H are Impaired under the Plan and thus the votes of Holders of Claims in such Classes shall be solicited. Pursuant to section 1126(c) of the Bankruptey Code and except as otherwise provided in section 1126(e) of the Bankruptey Code, an Impaired Class of Claims has accepted the Plan if the Plan is accepted by the Holders of at least two-thirds in dollar amount and more than one-half in number of the Allowed Claims of such Class that have timely and properly voted to accept or reject the Plan.
- 6.4 Classes Deemed to Reject the Plan. Holders of Claims and Interests in Classes 11-111 are not entitled to receive any Distribution under the Plan on account of their

Interests and, therefore, votes to accept or reject the Plan shall not be solicited from Holders of Claims or Interests in such Classes.

- 6.5 Elimination of Classes. Any Class that does not contain any Allowed Claims or any Claims temporarily allowed for voting purposes under Bankruptcy Rule 3018, as of the date of the commencement of the Confirmation Hearing, shall be deemed to have been deleted from the Plan for purposes of (a) voting to accept or reject the Plan and (b) determining whether it has accepted or rejected the Plan under section 1129(a)(8) of the Bankruptcy Code.
- 6.6 Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code. To the extent that any Impaired Class votes to reject the Plan or is deemed to have rejected it, Debtors shall request confirmation of the Plan under section 1129(b) of the Bankruptcy Code.
- requirements of section 1129 of the Bankruptcy Code must be satisfied separately with respect to each Debtor or consolidated group of Debtors. The Plan Proponents reserve the right to alter, amend, modify, revoke or withdraw the Plan (subject to the consent of the other Plan Proponent, as applicable, which consent shall not be unreasonably withheld) as it applies to any particular Debtor or consolidated group of Debtors. A determination by the Bankruptcy Court that the Plan, as it applies to a particular Debtor, is not confirmable pursuant to section 1129 of the Bankruptcy Code shall not limit or affect: (i) the confirmability of the Plan as it applies to any other Debtor; or (ii) the Debtors' ability to modify the Plan, subject to the limitations in this Section 6.7, as it applies to any particular Debtor, to satisfy the confirmation requirements of section 1129 of the Bankruptcy Code.

#### ARTICLE VII

#### MEANS FOR IMPLEMENTATION OF THE PLAN

#### 7.1 Substantive Consolidation.

- (a) The Plan contemplates and is predicated upon the substantive consolidation of certain of the Debtors' Estates, but only for purposes of voting on the Plan and making Distributions to Holders of Claims under the Plan. For purposes of the Plan, the VSE Debtors shall be substantively consolidated; the ASA Debtors shall be substantively consolidated; and the remaining Debtors shall not be substantively consolidated. Notwithstanding the foregoing, the Debtors reserve all rights (subject to the consent of the Creditors' Committee and, as applicable, the Plan Committee, which consent shall not be unreasonably withheld) with respect to substantive consolidation of any and all of the Debtors.
- (b) With respect to the consolidated Debtor entities, on the Effective Date, and only as to the consolidated Debtor entities, (i) all assets and liabilities of the VSE Debtors and the ASA Debtors, respectively, will, for voting and Distribution purposes only, be

merged or treated as if they were merged, (ii) each Claim against the VSE Debtors or the ASA Debtors, respectively, will be deemed a single Claim against and a single obligation of the VSE Debtors and the ASA Debtors, respectively, (iii) all Intercompany Claims by, between, and among the VSE Debtors and the ASA Debtors, respectively, will, for voting and Distribution purposes only, be eliminated, and (iv) any obligation of the VSE Debtors and the ASA Debtors, respectively, and all guaranties thereof by one or more of the other VSE Debtors and ASA Debtors, respectively, will be deemed to be one obligation of all of the VSE Debtors and the ASA Debtors, respectively. Except as set forth in this Article, such substantive consolidation shall not (other than for purposes related to the Plan) (x) affect the legal and corporate structures of the Debtors or Reorganized Debtors, (y) cause any Debtor to be liable for any Claim or Interest under the Plan which it otherwise is not liable, and the liability of any Debtor for such Claim or Interest shall not be affected by such substantive consolidation, and (z) except as otherwise stated in this Article VII, affect Intercompany Claims of Debtors against Debtors.

(c) Unless the Bankruptcy Court has approved by a prior order the substantive consolidation of certain of the Debtors' Estates, which order shall be in form and substance reasonably acceptable to the Plan Proponents, the Plan shall serve as, and shall be deemed to be, a request for entry of an order substantively consolidating certain of the Debtors' Estates, but only for purposes of voting on the Plan and making Distributions to Holders of Claims and Interests under the Plan. If no objection to substantive consolidation of certain of the Debtors' Estates is timely filed and served by any Holder of an Impaired Claim affected by the Plan as provided herein on or before the Voting Deadline or such other date as may be established by the Bankruptcy Court, the Confirmation Order shall serve as the order approving the substantive consolidation of certain of the Debtors' Estates, but only for purposes of voting on the Plan and making Distributions to Holders of Claims and Interests under the Plan. If any such objections are timely filed and served, a hearing with respect to the substantive consolidation of certain of the Debtors' Estates, but only for purposes of voting on the Plan and making Distributions to Holders of Claims and Interests under the Plan, and any objections thereto shall be part of the Confirmation Hearing.

### 7.2 Continued Corporate Existence; Dissolution of Reorganized Debtors.

- (a) The Debtors shall continue to exist as the Reorganized Debtors after the Effective Date in accordance with the laws of their respective states of incorporation or organization and pursuant to their respective certificates of incorporation, by-laws, articles of formation, operating agreements, and other organizational documents in effect prior to the Effective Date, except to the extent such certificate of incorporation and by-laws are amended under the Plan, for the limited purposes of liquidating all of the assets of the Estates and making Distributions in accordance with the Plan.
- (b) As soon as practicable after the Plan Administrator exhausts the assets of the Debtors' Estates by making the final Distribution of Cash under the Plan and the Plan Administrator Agreement, the Plan Administrator shall at the expense of the Debtors'

26

Estates (i) provide for the retention and storage of the books, records and files that shall have been delivered to or created by the Plan Administrator until such time as all such books, records and files are no longer required to be retained under applicable law, and File a certificate informing the Bankruptcy Court of the location at which such books, records and files are being stored; (ii) File a certification stating that the assets of the Debtors' Estates have been exhausted and final Distributions of Cash have been made under the Plan; (iii) File the necessary paperwork in the respective jurisdictions in which they are incorporated or otherwise organized to effectuate the dissolution of the Reorganized Debtors in accordance with the laws of such jurisdiction or jurisdictions; and (iv) resign as the sole officer, manager, or director, as applicable, of the Reorganized Debtors. Upon the Filing of the certificate described in section (ii) of the preceding sentence, the Reorganized Debtors shall be deemed dissolved for all purposes without the necessity for any other or further actions to be taken by or on behalf of the Reorganized Debtors or payments to be made in connection therewith other than the filing of a motion for final decree with respect to a particular Debtor.

- 7.3 Certificate of Incorporation and By-laws. The certificate and articles of incorporation and by-laws of each Debtor shall be amended as necessary to satisfy the provisions of the Plan and the Bankruptcy Code and shall include, among other things, a provision (a) prohibiting the issuance of non-voting equity securities under Section 1123(a)(6) of the Bankruptcy Code and (b) limiting the activities of each Reorganized Debtor to matters authorized under the Plan. The amended certificate of incorporation and by-laws of each Debtor, in form and substance acceptable to the Creditors' Committee in its sole discretion and subject to the reasonable consent of the Debtors, shall be Filed on or before the date of the Confirmation Hearing.
- 7.4 Directors, Managers, and Officers; Effectuating Documents; Further Transactions. From and after the Effective Date, the Plan Administrator shall serve as the sole officer and director or manager, as applicable, of the Reorganized Debtors. Subject to the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan, the Plan Administrator shall be authorized to execute, deliver, file or record such documents, instruments, releases and other agreements and to take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.

#### 7.5 The Plan Administrator.

(a) Appointment. From and after the Effective Date, an entity to be designated by the Creditors' Committee in its sole discretion, in consultation with the Debtors, prior to the Confirmation Date shall serve as the Plan Administrator pursuant to the Plan Administrator Agreement and the Plan, until the resignation or discharge and the appointment of a successor Plan Administrator in accordance with the Plan Administrator Agreement and the Plan.

- (b) Plan Administrator Agreement. Prior to or on the Effective Date, the Debtors shall execute a Plan Administrator Agreement in substantially the same form as set forth in the Plan Supplement which Plan Administrator Agreement shall be in form and substance acceptable to the Creditors' Committee in its sole discretion. Any nonmaterial modifications to the Plan Administrator Agreement made by the Debtors, with the consent of the Creditors' Committee, prior to the Effective Date are hereby ratified. The Plan Administrator Agreement will contain provisions permitting the amendment or modification of the Plan Administrator Agreement necessary to implement the provisions of the Plan.
- (c) Rights, Powers and Duties of the Reorganized Debtors and the Plan Administrator. The Reorganized Debtors shall retain and have all the rights, powers and duties necessary to carry out their responsibilities under the Plan. Subject to the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan, such rights, powers and duties, which shall be exercisable by the Plan Administrator on behalf of the Reorganized Debtors and the Estates pursuant to the Plan and the Plan Administrator Agreement, shall include, among others:
  - (i) liquidating the Reorganized Debtors' assets;
  - (ii) investing the Cash of the Estates of the Reorganized Debtors, including, but not limited to, the Cash held in the Reserves in (A) direct obligations of the United States of America or obligations of any agency or instrumentality thereof that are backed by the full faith and credit of the United States of America, including funds consisting solely or predominantly of such securities; (B) money market deposit accounts, checking accounts, savings accounts or certificates of deposit, or other time deposit accounts that are issued by a commercial bank or savings institution organized under the laws of the United States of America or any state thereof; or (C) any other investments that may be permissible under section 345 of the Bankruptcy Code or as otherwise ordered by the Bankruptcy Court;
  - (iii) calculating and paying all Distributions to be made under the Plan, the Plan Administrator Agreement and other orders of the Bankruptcy Court to Holders of Allowed Claims;
  - (iv) employing, supervising and compensating professionals (including, for the avoidance of doubt, professionals retained by the Creditors' Committee and the Plan Committee) retained to represent the interests of and serve on behalf of the Reorganized Debtors and the Estates;
  - (v) making and filing tax returns for any of the Debtors or the Reorganized Debtors;

- (vi) as provided in Section 7.5(g) hereof, objecting to Claims or Interests Filed against any of the Debtors' Estates on any basis;
- (vii) seeking estimation of contingent or unliquidated claims under section 502(c) of the Bankruptcy Code;
- (viii) seeking determination of tax liability under section 505 of the Bankruptcy Code;
- (ix) prosecuting avoidance actions under sections 544, 545, 547, 548, 549 and 553 of the Bankruptcy Code;
- (x) prosecuting turnover actions under sections 542 and 543 of the Bankruptcy Code;
- (xi) prosecuting, settling, dismissing or otherwise disposing of the Litigation Claims;
  - (xii) closing the Chapter 11 Cases;
  - (xiii) dissolving and winding up the Reorganized Debtors;
- (xiv) exercising all powers and rights, and taking all actions, contemplated by or provided for in the Plan Administrator Agreement; and
- (xv) taking any and all other actions necessary or appropriate to implement or consummate the Plan and the provisions of the Plan Administrator Agreement.
- (d) Compensation of the Plan Administrator. The Plan Administrator shall be compensated from the Operating Reserve pursuant to the terms of the Plan Administrator Agreement. Any professionals retained by the Plan Administrator shall be entitled to reasonable compensation for services rendered and reimbursement of expenses incurred from the Operating Reserve. The payment of the fees and expenses of the Plan Administrator and its retained professionals shall be made in the ordinary course of business and shall not be subject to the approval of the Bankruptcy Court; provided, however, that any disputes related to such fees and expenses shall be brought before the Bankruptcy Court.
- (e) Indemnification. The Reorganized Debtors shall indemnify and hold harmless (i) the Plan Administrator (in its capacity as such and as officer, director, or manager, as applicable, of the Reorganized Debtors), (ii) such individuals that may serve as officers and directors of the Reorganized Debtors, if any, and (iii) the Administrator Professionals (collectively, the "Indemnified Parties"), from and against and with respect to any and all

liabilities, losses, damages, claims, costs and expenses, including but not limited to attorneys' fees arising out of or due to their actions or omissions, or consequences of such actions or omissions, other than acts or omissions resulting from such Indemnified Party's willful misconduct or gross negligence, with respect to the Reorganized Debtors or the implementation or administration of the Plan or Plan Administrator Agreement. To the extent an Indemnified Party asserts a claim for indemnification as provided above, the legal fees and related costs incurred by counsel to the Plan Administrator in monitoring and participating in the defense of such claims giving rise to the asserted right of indemnification shall be advanced to such Indemnified Party (and such Indemnified Party undertakes to repay such amounts if it ultimately shall be determined that such Indemnified Party is not entitled to be indemnified therefore) out of the Operating Reserve or any insurance purchased using the Operating Reserve. The indemnification provisions of the Plan Administrator Agreement shall remain available to and be binding upon any former Plan Administrator or the estate of any decedent of the Plan Administrator and shall survive the termination of the Plan Administrator Agreement.

- (f) Insurance. The Plan Administrator shall be authorized to obtain and pay for out of the Operating Reserve all reasonably necessary insurance coverage for itself, its agents, representatives, employees or independent contractors, and the Reorganized Debtors, including, but not limited to, coverage with respect to (i) any property that is or may in the future become the property of the Reorganized Debtors or their Estates and (ii) the liabilities, duties and obligations of the Plan Administrator and its agents, representatives, employees or independent contractors under the Plan Administrator Agreement (in the form of an errors and omissions policy or otherwise), the latter of which insurance coverage may, at the sole option of the Plan Administrator, remain in effect for a reasonable period (not to exceed seven years) after the termination of the Plan Administrator Agreement.
- (g) Authority to Object to Claims and Interests and to Settle Disputed Claims. From and after the Effective Date, the Plan Administrator, on behalf of the Reorganized Debtors, shall be authorized, with respect to those Claims or Interests which are not Allowed hereunder or by Court order, (i) to object to any Claims or Interests Filed against any of the Debtors' Estates and (ii) subject to Section 7.5(h) hereof, pursuant to Fed. R. Bankr. P. 9019(b) and section 105(a) of the Bankruptcy Code, to compromise and settle Disputed Claims, in accordance with the following procedures, which shall constitute sufficient notice in accordance with the Bankruptcy Code and the Bankruptcy Rules for compromises and settlements of claims:
  - (i) If the proposed amount at which the Disputed Claim is to be Allowed is less than or equal to \$500,000 or if the difference between the Scheduled amount of a Disputed Claim and the proposed amount at which the Disputed Claim is to be Allowed is less than \$100,000, the Reorganized Debtors and the Plan Administrator shall be authorized and empowered to settle the Disputed Claim and execute necessary documents, including a stipulation of settlement or release, in their sole discretion and without notice to any party, and

the Plan Administrator shall have no liability to any party for the reasonableness of such settlement.

- (ii) If the proposed amount at which the Disputed Claim is to be allowed is greater than \$500,000 but less than or equal to \$10 million or if the difference between the Scheduled amount of a Disputed Claim and the proposed amount at which the Disputed Claim is to be Allowed is equal to or greater than \$100,000 but less than \$500,000, the Reorganized Debtors and the Plan Administrator shall be authorized and empowered to settle such Disputed Claim and execute necessary documents, including a stipulation of settlement or release, only if the Plan Committee after ten (10) days notice does not object to such proposed settlement. If the Plan Committee objects to the settlement, the Plan Administrator may seek Bankruptcy Court approval of such settlement.
- (iii) If the proposed amount at which the Disputed Claim is to be allowed is greater than \$10 million, the Reorganized Debtors and the Plan Administrator shall be authorized and empowered to settle the Disputed Claim and execute necessary documents, including a stipulation of settlement or release, only upon receipt of Bankruptcy Court approval of such settlement.
- (h) Authority to Settle Disputed Prepetition Makewhole Claims.

  Notwithstanding anything contained herein, the procedures set forth in Section 7.5(g) shall not apply to Disputed Prepetition Makewhole Claims. With respect to any such Disputed Prepetition Makewhole Claims, from and after the Effective Date, the Plan Administrator, on behalf of the Reorganized Debtors, shall be authorized to litigate, compromise and settle, pursuant to Fed. R. Bankr. P. 9019(b) and section 105(a) of the Bankruptcy Code, any Disputed Prepetition Makewhole Claims, subject to the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan. If the Plan Committee does not consent or objects to the settlement, the Plan Administrator may seek Bankruptcy Court approval of such settlement.
- 7.6 Revesting of Assets. Except as expressly provided elsewhere in this Plan, on the Effective Date, the property of each Debtor's Estate shall revest in the applicable Reorganized Debtor.

### 7.7 Preservation of Rights of Action; Settlement of Litigation Claims.

(a) Preservation of Rights of Action. Except as otherwise provided in the Plan, the Confirmation Order, or in any document, instrument, release or other agreement entered into in connection with the Plan, in accordance with section 1123(b) of the Bankruptcy Code, the Debtors and their Estates shall retain the Litigation Claims. The Plan Administrator, on behalf of the Reorganized Debtors, may, subject to the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan, enforce, sue on, settle or compromise (or decline to do any of the foregoing) any or all of the Litigation Claims. Entry of the Confirmation

Order shall not constitute a waiver or release by the Debtors or their Estates of any cause of action except as expressly provided for by the Plan. On and after the Effective Date, the Plan Administrator shall be assigned all Causes of Action arising under Sections 542, 543, 544, 547 through 551, and 553 of the Bankruptcy Code including Causes of Action arising under fraudulent transfer laws under federal and state statutes and common law and all Causes of Action against the Debtors' current and former officers, directors, employees, shareholders or non-legal professionals under section 1123(b)(3) of the Bankruptcy Code related to conduct in connection with the Debtors prior to the Petition Date, and any and all proceeds from insurance policies related thereto. The Plan Administrator shall be authorized to enforce, prosecute, settle or compromise the Causes of Action. The Plan Administrator may pursue such Causes of Action, in accordance with the obligations of the Plan and the best interests of all of the beneficiaries of the Plan. The method of distribution of the Estates' assets pursuant to the Plan shall not, and shall not be deemed to, prejudice the Causes of Action, which shall survive entry of the Confirmation Order for the beneficiaries of the Plan.

(b) Settlement of Litigation Claims. At any time after the Confirmation Date and before the Effective Date, notwithstanding anything in the Plan to the contrary, but subject to the ultimate supervisory authority of the Creditors' Committee, as set forth in section 7.8 of the Plan, the Debtors may settle any or all of the Litigation Claims with the approval of the Bankruptcy Court pursuant to Fed. R. Bankr. P. 9019.

#### 7.8 Creditors' Committee and Plan Committee.

(a) Dissolution of Creditors' Committee. On the Effective Date, the Creditors' Committee shall dissolve automatically and its members shall be deemed released of all their duties, responsibilities and obligations in connection with the Chapter 11 Cases or the Plan and its implementation, and the retention or employment of the Creditors' Committee's attorneys, accountants and other agents shall terminate, except with respect to (i) all matters related to Fee Claims and (ii) any appeals of the Confirmation Order. All expenses of members of the Creditors' Committee and the fees and expenses of their professionals through the Effective Date shall be paid in accordance with the terms and conditions of the applicable orders of the Bankruptcy Court. Counsel to the Creditors' Committee shall be entitled to reasonable compensation and reimbursement of expenses for post-Effective Date activities authorized hereunder upon the submission of invoices to the Reorganized Debtors.

#### (b) Creation of Plan Committee; Procedures.

(i) On the Effective Date, the Plan Committee shall be formed and constituted unless otherwise determined in the sole discretion of the Creditors' Committee. The Plan Committee shall consist of three (3) members who shall be appointed by the Creditors' Committee and whose identities shall be

disclosed to the Bankruptcy Court at the Confirmation Hearing. Membership on the Plan Committee shall be on an institutional and not on an individual basis. In the event that a member of the Plan Committee resigns from its position on the Plan Committee, such member shall have the right to designate its successor on the Plan Committee. To the extent that a resigning member of the Plan Committee does not exercise its right to designate its successor on the Plan Committee within ten (10) days after such member's resignation from the Plan Committee, the non-resigning members of the Plan Committee shall have the right to designate a successor to the resigning member of the Plan Committee.

- (ii) In the event that there have been no Plan Committee members for a period of thirty (30) consecutive days, then the Plan Administrator may, during such vacancy and thereafter, in its sole discretion, ignore any reference in the Plan, the Plan Administrator Agreement or the Confirmation Order to a Plan Committee, and all references to the Plan Committee's ongoing duties and rights in the Plan, the Plan Administrator Agreement and the Confirmation Order shall be null and void.
- (c) Standing of Plan Committee. The Plan Committee shall have independent standing to appear and be heard in the Bankruptcy Court as to any matter relating to the Plan, the Plan Administrator Agreement, the Plan Administrator, the Estates or the Reorganized Debtors, including any matter as to which the Bankruptcy Court has retained jurisdiction pursuant to Article XIII of the Plan.
- Function and Duration; Compensation and Expenses. The Plan Committee shall have ultimate supervisory authority over the Plan Administrator. The Plan Administrator shall report to the Plan Committee, and the Plan Committee shall have the power to remove the Plan Administrator. The Plan Committee (i) shall be responsible for (A) instructing and supervising the Reorganized Debtors and the Plan Administrator with respect to their responsibilities under the Plan and the Plan Administrator Agreement, (B) reviewing and approving the prosecution of adversary and other proceedings, if any, including approving proposed settlements thereof, (C) reviewing and approving objections to and proposed settlements of Disputed Claims, (D) performing such other duties that may be necessary and proper to assist the Plan Administrator and its retained professionals, and (ii) shall remain in existence until such time as the final Distributions under the Plan have been made by the Reorganized Debtors. The members of the Plan Committee shall serve without compensation for their performance of services as members of the Plan Committee, except that they shall be entitled to reimbursement of reasonable expenses by the Reorganized Debtors from the Operating Reserve, provided, however, that if the Unsecured Note Indenture Trustee shall be a member of the Plan Committee, it shall be entitled to be paid its Unsecured Note Indenture Trustee Fees and Expenses. The Plan Committee may retain counsel or other professionals who shall be entitled to reasonable compensation and reimbursement of actual, necessary expenses

upon the submission of invoices to the Reorganized Debtors; *provided*, *however*, that any disputes related to such fees and expenses may be brought before the Bankruptcy Court.

Liability; Indemnification. Neither the Plan Committee, nor any of its members or designees, nor any duly designated agent or representative of the Plan Committee, or their respective employees, shall be liable for the act or omission of any other member, designee, agent or representative of the Plan Committee, nor shall any member be liable for any act or omission taken or omitted to be taken in its capacity as a member of the Plan Committee, other than acts or omissions resulting from such member's willful misconduct or gross negligence. The Reorganized Debtors shall indemnify and hold harmless the Plan Committee and its members and designees, and any duly designated agent or representative thereof (in their capacity as such), from and against and with respect to any and all liabilities, losses, damages, claims, costs and expenses, including but not limited to attorneys' fees arising out of or due to their actions or omissions, or consequences of such actions or omissions, other than as a result of their willful misconduct or gross negligence, with respect to the Reorganized Debtors or the implementation or administration of the Plan. To the extent the Reorganized Debtors indemnify and hold harmless the Plan Committee and its members and designees, or any duly designated agent or representative thereof (in their capacity as such), as provided above, the legal fees and related costs incurred by counsel to the Plan Committee in monitoring and participating in the defense of such claims giving rise to the right of indemnification shall be paid out of the Operating Reserve.

7.9 Cancellation of Securities. Instruments and Agreements Evidencing Claims and Interests. Except as otherwise provided in the Plan, on the Effective Date and concurrently with the applicable Distributions made pursuant to this Article VII, the promissory notes, share certificates (including treasury stock), the Unsecured Notes, the Unsecured Note Indenture, other instruments evidencing any Claims or Interests, and all options, warrants, calls, rights, puts, awards, commitments or any other agreements of any character to acquire such Interests shall be deemed automatically extinguished, canceled and of no further force and effect, without any further act or action under any applicable agreement, law, regulation, order or rule, and the obligations of the Debtors under the notes, share certificates, Unsecured Notes, the Unsecured Note Indenture and other agreements and instruments governing such Claims and Interests shall be automatically discharged. The Holders of or parties to such canceled notes, share certificates, Unsecured Notes, the Unsecured Note Indenture and other agreements and instruments shall have no rights arising from or relating to such notes, share certificates, Unsecured Notes, the Unsecured Note Indenture, and other agreements and instruments or the cancellation thereof, except the rights provided pursuant to the Plan; provided, however, that the Unsecured Notes and the Unsecured Note Indenture shall continue in effect solely for the purposes of (i) allowing the Unsecured Note Indenture Trustee or its agents to make Distributions to Holders of Unsecured Notes; (ii) allowing Holders of the Unsecured Notes to receive Distributions hereunder; and (iii) preserving the rights and liens of the Unsecured Note Indenture Trustee with respect to the Unsecured Note Indenture Trustee Fees and Expenses to the extent not otherwise paid. The Unsecured Note Indenture shall terminate completely upon

the completion of all Distributions to the Holders of the Unsecured Notes and the payment in full of the Unsecured Note Indenture Trustee Fees and Expenses. Notwithstanding the foregoing, the indenture governing the Prepetition Makewhole Claims and any related documents thereto governing the rights and obligations of the Debtors and non-Debtor parties with respect to the Prepetition Makewhole Claims shall survive for purposes of the Prepetition Makewhole Claim litigation, until such time as the Prepetition Makewhole Claims have been Allowed by a Final Order and the obligations are satisfied in accordance with the terms of the Plan or the Prepetition Makewhole Claim have been disallowed by Final Order or expunged.

After the performance by the Unsecured Note Indenture Trustee or its agents of all duties that are required under the Plan, the Confirmation Order and/or under the terms of the Unsecured Note Indenture, the Unsecured Note Indenture Trustee and its agents and advisors shall be relieved of, and released from, all obligations associated with the Unsecured Notes arising under the Unsecured Note Indenture or under other applicable agreements or law, and the Unsecured Note Indenture Trustee shall be deemed fully released and discharged.

- 7.10 Sources of Cash for Plan Distributions. Except as otherwise provided in the Plan or the Confirmation Order, all Cash necessary for the Reorganized Debtors and the Plan Administrator to make payments pursuant to the Plan shall be obtained from the Debtors' Cash balances and the liquidation of the Debtors' and the Estates' remaining non-Cash assets, if any. Cash payments to be made pursuant to the Plan shall be made by the Reorganized Debtors (or any successor thereto) or, if the Disbursing Agent is an entity other than the Reorganized Debtors, the Disbursing Agent.
- 7.11 Exemption from Transfer Taxes. Pursuant to section 1146 of the Bankruptcy Code, (a) the issuance, transfer or exchange of notes or equity securities under the Plan; (b) the creation of any mortgage, deed of trust, lien, pledge or other security interest; (c) the making or assignment of any contract, lease or sublease; or (d) the making or delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with, the Plan, including, without limitation, any merger agreements, agreements of consolidation, restructuring, disposition, liquidation or dissolution, deeds, bills of sale, and transfers of tangible property, will not be subject to any stamp tax, recording tax, personal property tax, real estate transfer tax, sales or use tax or other similar tax. The Confirmation Order shall direct the appropriate state or local government officials or agents to forgo the collection of any such tax or governmental assessment and to accept for filing and recordation any of the foregoing instruments or other documents without the payment of any such tax or governmental assessment.
- 7.12 Settlement of Certain Intercompany Claims. The Intercompany Claims Settlement shall be in full satisfaction of all of the USBio Debtors' claims against VeraSun Marketing, LLC and all of VeraSun Marketing, LLC's claims against the USBio Debtors. VeraSun Marketing, LLC and the USBio Debtors hereby release each other of and from any and all claims, causes of action or demands, whether known or unknown, that either now has or may

35

ever have had against each other; *provided*, *however*, that VeraSun Marketing shall not be released from its obligation to transfer funds pursuant to the Intercompany Claims Settlement.

7.13 Agstar Settlement. Nothing set forth in this Plan shall be deemed to modify or amend the AgStar Settlement.

#### ARTICLE VIII

#### PROVISIONS GOVERNING DISTRIBUTIONS

- 8.1 Distributions for Claims Allowed as of the Effective Date. Except as otherwise provided herein or as ordered by the Bankruptcy Court, initial Distributions to be made on account of Claims that are Allowed Claims as of the Effective Date shall be made on the Distribution Date. Any payment or Distribution required to be made under the Plan on a day other than a Business Day shall be made on the next Business Day. Distributions on account of Claims that first become Allowed Claims after the Effective Date shall be made pursuant to the terms and conditions of the Plan Administrator Agreement and Articles V, VIII and X of the Plan.
- 8.2 Interest on Claims. Unless otherwise specifically provided for in the Plan, the Confirmation Order or another order of the Bankruptcy Court, or if required by applicable bankruptcy law, interest, fees, costs and other charges accruing or incurred on or after the Petition Date shall not accrue or be paid on any Claims, and no Holder of a Claim shall be entitled to interest, fees, costs or other charges accruing or incurred on or after the Petition Date through the date such Claim is satisfied in accordance with the terms of the Plan. In no event shall interest accrue after the Effective Date on account of any Allowed Claim unless specifically provided for in the Plan, applicable bankruptcy law, the Confirmation Order or another Final Order of the Bankruptcy Court.
- 8.3 Distributions by the Reorganized Debtors. The Reorganized Debtors shall make all Distributions of Cash required to be distributed under the applicable provisions of the Plan and the Plan Administrator Agreement. The Reorganized Debtors may employ or contract with other entities to assist in or make the Distributions required by the Plan and the Plan Administrator Agreement.

# 8.4 Delivery of Distributions and Undeliverable or Unclaimed Distributions.

(a) Delivery of Distributions in General. Distributions to Holders of Allowed Claims shall be made at the addresses set forth in the Schedules unless such addresses are superseded by proofs of claim or transfers of claim Filed pursuant to Bankruptcy Rule 3001 (or at the last known addresses of such Holders if the Debtors or the Reorganized Debtors have been notified in writing of a change of address); provided, however, that in the case of (i) the

Holders of Unsecured Notes, Distributions shall be made as set forth in clause (b) below; and (ii) a Claim Holder whose Claim is governed by an indenture or other agreement and is administered by an indenture trustee other than the Unsecured Note Indenture Trustee, Distributions shall be made at the addresses contained in the official records of the indenture trustee, including as set forth in any Ballots cast with respect to such Claims.

(b) The Reorganized Debtors or any entities employed or contracted with to make Distributions pursuant to Section 8.3 above shall make all Distributions of Cash required to be made to Holders of Unsecured Notes to the Unsecured Note Indenture Trustee or its designated agent. The Unsecured Note Indenture Trustee or its designated agent shall make all Distributions to Holders of Unsecured Notes in accordance with the terms of the Unsecured Note Indenture, subject to the terms of the Plan; provided, however, that under all circumstances, the fees and expenses of the Unsecured Note Indenture or its designated agent incurred in connection with such Distributions shall be paid by the Reorganized Debtors without further approval from the Bankruptcy Court.

#### (c) Surrender of Canceled Notes

As a condition precedent to receiving any Distribution hereunder on account of an Allowed Claim evidenced by an Unsecured Note canceled pursuant to Section 7.9 hereof, the Holder of such Allowed Claim shall surrender such Unsecured Note to the Unsecured Note Indenture Trustee for cancellation. All payments to Holders of Unsecured Notes shall only be made after such surrender. Upon surrender of such Unsecured Notes certificates, the Unsecured Note Indenture Trustee shall cancel and destroy such Unsecured Notes. As soon as practicable after surrender of the Unsecured Notes certificates, the Unsecured Note Indenture Trustee shall distribute to the Holder thereof such Holder's pro rata share of the Distribution, but subject to the rights of the Unsecured Note Indenture Trustee to assert its charging lien to the extent the Unsecured Note Indenture Trustee Fees and Expenses are not paid pursuant to the Plan.

#### (d) Undeliverable and Unclaimed Distributions.

(i) <u>Holding and Investment of Undeliverable and Unclaimed Distributions</u>. If the Distribution to any Holder of an Allowed Claim is returned to the Disbursing Agent as undeliverable or is otherwise unclaimed, no further Distributions shall be made to such Holder unless and until the Disbursing Agent is notified in writing of such Holder's then-current address. Undeliverable and unclaimed Distributions shall be deposited in a segregated account, designated as an "unclaimed distribution reserve" (the "Unclaimed Distribution Reserve"), for the benefit of all such similarly situated Persons until such time as a Distribution becomes deliverable or is claimed. The account for the Unclaimed Distribution Reserve may be an interest-bearing account, provided that any

interest accruing on funds in the Unclaimed Distribution Reserve shall be transferred to the Reorganized Debtors for Distribution in accordance with the terms of the Plan.

- (ii) After Distributions Become Deliverable. The Disbursing Agent and the Unsecured Note Indenture Trustee, as applicable, shall make all Distributions that have become deliverable or have been claimed since the Distribution Date on the next Quarterly Distribution Date, or as soon as practicable after such Distribution has become deliverable.
- (iii) Failure to Claim Undeliverable Distributions. Any Holder of an Allowed Claim that does not assert a claim pursuant to the Plan for an undeliverable or unclaimed Distribution within six months after the applicable date of Distribution shall be deemed to have forfeited its claim for such undeliverable or unclaimed Distribution and shall be forever barred and enjoined, notwithstanding any federal or state escheat laws to the contrary, from asserting any such claim for an undeliverable or unclaimed Distribution against the Debtors and the Estates, the Reorganized Debtors, the Plan Administrator, the Unsecured Note Indenture Trustee or their property. In such cases, any Cash in the Unclaimed Distribution Reserve, or held in trust by the Unsecured Note Indenture Trustee, for Distribution on account of such claims for undeliverable or unclaimed Distributions shall be Distributed to the other Holders in the same Class, free of any restrictions thereon. Nothing contained in the Plan or the Plan Administrator Agreement shall require the Unsecured Note Indenture Trustee and/or any Disbursing Agent, including, but not limited to, the Plan Administrator or the Reorganized Debtors, to attempt to locate any Holder of an Allowed Claim for purposes of making any Distribution under the Plan.
- Bebtors or the Unsecured Note Indenture Trustee, as the case may be, will have no obligation to recognize the transfer of, or the sale of any participation in, any Allowed Claim that occurs after the close of business on the Distribution Record Date, and will be entitled for all purposes herein to recognize and distribute only to those Holders of Allowed Claims who are Holders of such Claims, or participants therein, as of the close of business on the Distribution Record Date. The Disbursing Agent, the Reorganized Debtors, and the Unsecured Note Indenture Trustee shall instead be entitled to recognize and deal for all purposes under the Plan with only those record Holders stated on the official claims register or the transfer books and records as maintained by the Unsecured Note Indenture Trustee or its agents, as the case may be, as of the close of business on the Distribution Record Date.
- 8.6 Unsecured Note Indenture Trustee as Claim Holder. Consistent with Bankruptcy Rule 3003(c), the Debtors or the Reorganized Debtors, as the case may be, shall recognize proofs of claim filed by the Unsecured Note Indenture Trustee in respect of the

Unsecured Note Claims. Accordingly, any Unsecured Note Claim, proof of which is filed by the registered or beneficial Holder of an Unsecured Note Claim, may be disallowed as duplicative of the Unsecured Note Claims of the Unsecured Note Indenture Trustee, without need for any further action or Bankruptcy Court order.

- 8.7 Allocation of Plan Distributions Between Principal and Interest. To the extent that any Allowed Claim entitled to a Distribution under the Plan is comprised of indebtedness and accrued but unpaid interest thereon, such Distribution shall, for federal and all other income tax purposes, be allocated to the principal amount of the Claim first and then, to the extent the consideration exceeds the principal amount of the claim, to the portion of such Claim representing unpaid interest accrued through the Effective Date.
- 8.8 Means of Cash Payment. Payments of Cash made pursuant to the Plan shall be in U.S. dollars and shall be made, at the option and in the sole discretion of the Plan Administrator, by (a) checks drawn on or (b) wire transfer from a domestic bank selected by the Plan Administrator. Checks issued by the Disbursing Agent shall be null and void if not negotiated within ninety days after issuance, but may be requested to be reissued until the distribution revests to the Reorganized Debtors pursuant to Section 8.4(b)(iii) hereof. Cash payments to foreign creditors may be made, at the option of the Plan Administrator, in such funds and by such means as are necessary or customary in a particular foreign jurisdiction.
- 8.9 Withholding and Reporting Requirements. In connection with the Plan and all Distributions thereunder, the Disbursing Agent or the Plan Administrator, as applicable, on behalf of the Reorganized Debtors, is authorized to take any and all actions that may be necessary or appropriate to comply with all withholding, payment and reporting requirements imposed by any federal, state, local or foreign taxing authority, and all Allowed Claims and Distributions hereunder shall be subject to any such withholding and reporting requirements. The Disbursing Agent or the Plan Administrator, as applicable, on behalf of the Reorganized Debtors, shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding, payment and reporting requirements. All Holders of Claims shall be required to provide any information necessary to effect information reporting, payment and withholding with respect to such taxes.
- 8.10 Setoffs. The Plan Administrator may, pursuant to section 553 of the Bankruptcy Code or applicable nonbankruptcy laws, but shall not be required to, set off against any Claim and the payments or other Distributions to be made pursuant to the Plan in respect of such Claim, or claims of any nature whatsoever that the Debtors or Reorganized Debtors may have against the Holder of such Claim (except to the extent that such claims have been waived by the Debtors in the Plan or otherwise); provided, however, that neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Reorganized Debtors of any such claim that the Debtors or Reorganized Debtors may have against such Holder. In no event shall any Holder of Claims or Interests be entitled to setoff any Claim or Interest against any Claim, right, or cause of action of the Reorganized Debtors, unless such

39

Holder has filed a motion with the Bankruptcy Court requesting the authority to perform such setoff on or before the Voting Deadline.

- other provision of the Plan or the Plan Administrator Agreement, (a) neither the Plan Administrator nor the Reorganized Debtors shall be required to make Distributions or payments of fractions of dollars, and whenever any payment of a fraction of a dollar under the Plan would otherwise be called for, the actual payment made shall reflect a rounding of such fraction to the nearest whole dollar (up or down), with half dollars being rounded down, and (b) the Plan Administrator shall have no obligation to make a Distribution on account of an Allowed Claim from any Reserve or account (i) to any Holder of an Allowed Claim if the aggregate amount of all Distributions authorized to be made from all such Reserves or accounts on the Quarterly Distribution Date in question is less than \$250,000, in which case such Distributions shall be deferred to the next Quarterly Distribution Date, or (ii) to a specific Holder of an Allowed Claim if the amount to be distributed to that Holder on the particular Distribution Date is less than \$50.00, unless such Distribution constitutes the final Distribution to such Holder.
- 8.12 Release of Liens. Except as otherwise provided in the Plan or in any contract, instrument, release or other agreement or document created or assumed in connection with the Plan, on the Effective Date and concurrently with the applicable Distributions made pursuant to the Plan, all mortgages, deeds of trust, liens, pledges or other security interests against the property of any Debtor's Estate shall be fully released and discharged (except for charging Liens of the Unsecured Note Indenture Trustee to the extent the Unsecured Note Indenture Trustee Fees and Expenses are not paid pursuant to the Plan), and all of the right, title and interest of any holder of such mortgages, deeds of trust, liens, pledges or other security interests shall revert to the applicable Reorganized Debtor.

#### ARTICLE IX

### TREATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES

Plan, the Confirmation Order shall constitute an order under section 365 of the Bankruptcy Code rejecting all prepetition executory contracts and unexpired leases to which any Debtor is a party, on and subject to the occurrence of the Effective Date, unless such contract or lease (a) previously shall have been assumed or rejected by the Debtors, (b) previously shall have expired or terminated pursuant to its own terms before the Effective Date, or (c) is identified in the Plan Supplement as a contract or lease to be assumed; provided, however, that the Debtors may, subject to the consent of the Creditors' Committee (which consent shall not be unreasonably withheld), amend such list of contracts or leases to be assumed at any time prior to the Confirmation Date.

- 9.2 Assumption of Director and Officer Liability Insurance Policies. Notwithstanding anything herein to the contrary, as of the Effective Date, the Debtors shall assume (and assign to the Reorganized Debtors if necessary to continue the Debtors director and officer liability insurance policies in full force) all of the director and officer liability insurance policies pursuant to section 365(a) of the Bankruptcy Code. Entry of the Confirmation Order shall constitute the Bankruptcy Court's approval of the Debtors' foregoing assumption of each of the director and officer liability insurance policies. Notwithstanding anything to the contrary contained herein. Confirmation of the Plan shall not discharge, impair, or otherwise modify any obligations assumed by the foregoing assumption of the director and officer liability insurance policies, and each such obligation shall be deemed and treated as an Executory Contract that has been assumed by the Debtors under the Plan as to which no Proof of Claim need be Filed. On or before the Effective Date, the Debtors may obtain reasonably sufficient tail coverage (i.e., director and officer insurance coverage that extends beyond the end of the policy period) under a directors and officers' liability insurance policy for the current and former directors, officers, and managers for such terms or periods of time, and placed with such insurers as are determined by the Debtors and the Creditors' Committee to be reasonable under the circumstances or as otherwise specified and ordered by the Bankruptcy Court in the Confirmation Order.
- 9.3 Bar to Rejection Damages. If the rejection of an executory contract or unexpired lease pursuant to Section 9.1 above gives rise to a Claim by the other party or parties to such contract or lease, such Claim shall be forever barred and shall not be enforceable against the applicable Debtor or its Estate, the Reorganized Debtors, the Plan Administrator or their respective successors or properties unless a proof of Claim is Filed and served on the Reorganized Debtors and counsel for the Reorganized Debtors within thirty (30) days after service of a notice of entry of the Confirmation Order or such other date as prescribed by the Bankruptcy Court.

#### ARTICLE X

# PROCEDURES FOR RESOLVING DISPUTED, CONTINGENT AND UNLIQUIDATED CLAIMS

Claims Objection Deadline (unless extended by an order of the Bankruptcy Court), the Debtors, the Creditors' Committee, the Plan Committee or the Plan Administrator, on behalf of the Reorganized Debtors, as the case may be, shall File objections to Claims with the Bankruptcy Court and serve such objections upon the Holders of each of the Claims to which objections are made. Nothing contained herein, however, shall limit the right to object to Claims, if any, Filed or amended after the Claims Objection Deadline. Subject to the limitations set forth in the Plan Administrator Agreement and Section 7.5 of the Plan, and the ultimate supervisory authority of the Plan Committee as set forth in Section 7.8 of the Plan, the Plan Administrator, on behalf of the Reorganized Debtors, shall be authorized to, and shall, resolve all Disputed Claims by

withdrawing or settling such objections thereto, or by litigating to judgment in the Bankruptcy Court or such other court having jurisdiction over the validity, nature and/or amount thereof.

- 10.2 No Distributions Pending Allowance. Notwithstanding any other provision of the Plan, no payments or Distributions shall be made with respect to all or any portion of a Disputed Claim unless and until all objections to such Disputed Claim have been settled or withdrawn or have been determined by Final Order, and the Disputed Claim, or some portion thereof, has become an Allowed Claim.
- Reorganized Debtors, shall, subject to and in accordance with the provisions of the Plan Administrator Agreement, (a) establish one or more general accounts into which shall be deposited all funds not required to be deposited into any other account or Reserve and (b) create, fund and withdraw funds from, as appropriate, the Operating Reserve, Administrative Claims Reserve, Disputed Claims Reserve, the Prepetition Makewhole Claims Reserve, the Litigation Claims Reserve and Unclaimed Distributions Reserve. The Reorganized Debtors shall sell non-Cash assets of the Estates, if any, in accordance with the provisions of the Plan and the Plan Administrator Agreement. The net proceeds of any such sales shall be deposited in an account or Reserve pursuant to the terms of the Plan Administrator Agreement.
- Disputed Claims Reserve. Prior to the Distribution Date, subject to and in accordance with the provisions of the Plan Administrator Agreement, the Plan Administrator, in consultation with the Plan Committee, shall create and fund the Disputed Claims Reserve with an amount of the Estates' Cash equal to one hundred percent (100%) of Distributions to which Holders of Disputed General Unsecured Claims would be entitled under the Plan as of such date if such Disputed General Unsecured Claims were Allowed Claims in their Disputed Claims Amount; provided, however, that the Debtors, the Reorganized Debtors, the Creditors' Committee or the Plan Committee, as applicable, may File motion(s) pursuant to section 502(c) of the Bankruptcy Code for order(s) estimating and limiting the amount of Cash which shall be deposited in the Disputed Claims Reserve in respect of any Disputed General Unsecured Claims, with notice and an opportunity to be heard to the affected Holders of such Disputed General Unsecured Claims and the Plan Committee; provided, further, that the Reorganized Debtors must within sixty (60) days after the Effective Date (or such other date as the Bankruptcy Court may order) File a motion(s) seeking to estimate any contingent or unliquidated Claims, with a notice and an opportunity to be heard to the affected Holders of such Disputed Claims and the Plan Committee. The Disputed Claims Reserve shall be fully funded no later than ten (10) days after the Bankruptcy Court has approved an order resolving such motion(s) and such order has been entered by the Clerk of the Bankruptcy Court.
- (b) Administrative Claims Reserve. Prior to the Distribution Date and on the first Quarterly Distribution Date after the Effective Date, subject to and in accordance with the provisions of the Plan Administrator Agreement, the Plan Administrator, in consultation with the Plan Committee, shall create and fund the Administrative Claims Reserve with an

42

amount of the Estates' Cash equal to the aggregate Disputed Claim Amount of all Disputed Administrative Claims, Disputed Priority Tax Claims, Disputed Non-Tax Priority Claims, and Disputed Other Secured Claims.

- (c) Prepetition Makewhole Claims Reserve. Prior to the Distribution Date, subject to and in accordance with the provisions of the Plan Administrator Agreement, the Plan Administrator, in consultation with the Plan Committee, shall create and fund the Prepetition Makewhole Claims Reserve with an amount of the Estates' Cash equal to the aggregate Disputed Claim Amount of the Disputed Prepetition Makewhole Claims.
- and in accordance with the Plan Administrator Agreement, the Plan Administrator shall create and fund the Litigation Claims Reserve with Cash of the Estates to fund the costs and expenses associated with investigating and, as necessary, prosecuting, settling, dismissing or otherwise disposing of any or all of the Litigation Claims; provided, however, that at no time shall the amount of the Litigation Claims Reserve exceed \$100,000; provided, further, that the Plan Administrator may at any time increase or decrease the amount of the Litigation Claims Reserve with the consent of the Plan Committee.
- payments and Distributions from the appropriate Reserves to the Holder of any Disputed Administrative Claim, Disputed Priority Tax Claim, Disputed Non-Tax Priority Claim, Disputed Other Secured Claim or Disputed General Unsecured Claim that has become an Allowed Claim, on the first Quarterly Distribution Date following the date that such Disputed Administrative Claim, Disputed Priority Tax Claim, Disputed Non-Tax Priority Claim, Disputed Other Secured Claim or Disputed General Unsecured Claim, as the case may be, becomes an Allowed Claim. Such Distributions shall be made in accordance with the Plan and the Plan Administrator Agreement. With respect to Disputed General Unsecured Claims, such Distribution shall be based upon the cumulative Distributions that would have been made to such Holder under the Plan if the Disputed General Unsecured Claim had been Allowed on the Effective Date and shall not be limited by the Disputed Claim Amounts previously reserved with respect to such Disputed General Unsecured Claims Reserve, but only to the extent that such additional amounts have not yet been distributed to Holders of Allowed General Unsecured Claims.

#### ARTICLE XI

#### CONFIRMATION AND CONSUMMATION OF THE PLAN

11.1 Conditions to Confirmation. The Bankruptcy Court shall not approve the Confirmation Order unless and until (a) the Bankruptcy Court shall have approved the Disclosure Statement, in a manner reasonably acceptable in form and substance to the Plan Proponents, as containing adequate information with respect to the Plan within the meaning of

section 1125 of the Bankruptcy Code; (b) the proposed Confirmation Order shall be reasonably acceptable in form and substance to the Plan Proponents; and (c) the Substantive Consolidation Order, which may be the Confirmation Order, shall be reasonably acceptable in form and substance to the Plan Proponents and shall have been approved by the Bankruptcy Court prior to or contemporaneously with the Confirmation Order; and (d) subject to sections 7.3 and 7.5 hereof, the most current version of the Plan Supplement and all of the schedules, documents, and exhibits contained therein (including the Plan Administrator Agreement) shall be Filed in form and substance reasonably acceptable to the Plan Proponents.

- 11.2 Conditions to Effective Date. The Plan Proponents intend to request that the Confirmation Order include a finding by the Bankruptcy Court that, notwithstanding Rule 3020(e) of the Federal Rules of Bankruptcy Procedure, the Confirmation Order shall take effect immediately upon its entry. The following are conditions precedent to the occurrence of the Effective Date: (a) the Debtors shall have Cash on hand sufficient to pay all Allowed Administrative Claims and Allowed Priority Tax Claims; (b) the Confirmation Order shall be a Final Order in form and substance reasonably acceptable to the Plan Proponents and no request for revocation of the Confirmation Order shall have been made or, if made, shall remain pending; (c) any order necessary to satisfy any condition to the effectiveness of the Plan shall have become a Final Order; and (d) subject to sections 7.3 and 7.5 hereof, the final version of the Plan Supplement and all of the schedules, documents, and exhibits contained therein (including the Plan Administrator Agreement) shall be Filed in form and substance reasonably acceptable to the Plan Proponents.
- 11.3 Waiver of Conditions Precedent. The Plan Proponents (subject to the consent of the other Plan Proponent, as applicable, which consent shall not be unreasonably withheld) may waive any of the conditions set forth in Section 11.1 or Section 11.2 at any time, without any notice to other parties in interest and without any further notice to or action, order, or approval of the Bankruptcy Court, and without any formal action other than proceeding to confirm or consummate the Plan.
- Date does not occur within sixty (60) days after the Confirmation Date, or by such later date, after notice and hearing, as is proposed by the Plan Proponents, then upon motion by either of the Plan Proponents and upon notice to such parties in interest as the Bankruptcy Court may direct, the Confirmation Order will be vacated by the Bankruptcy Court; provided, however, that, notwithstanding the Filing of such motion, the Confirmation Order may not be vacated if the Effective Date occurs before the Bankruptcy Court approves an order granting such motion. If the Confirmation Order is vacated pursuant to this Section 11.3, (a) the Plan shall be null and void in all respects; (b) any settlement of Claims provided for hereby shall be null and void without further order of the Bankruptcy Court; (c) the time within which the Debtors may assume and assign, or reject all executory contracts and unexpired leases shall be extended for a period of thirty (30) days after the date the Confirmation Order is vacated; and (d) nothing

contained in the Plan or Disclosure Statement shall constitute a waiver or release of any Claims, Interests or Litigation Claims.

#### ARTICLE XII

#### EFFECT OF PLAN CONFIRMATION

12.1 Binding Effect. The Plan shall be binding upon and inure to the benefit of the Debtors, all present and former Holders of Claims and Interests, and their respective successors and assigns, including, but not limited to, the Reorganized Debtors.

#### 12.2 Releases.

(a) Releases by the Debtors.

Pursuant to section 1123(b) of the Bankruptcy Code, and except as otherwise specifically provided in the Plan, upon the Effective Date, each of the Debtors shall release unconditionally, and hereby is deemed to forever release unconditionally (i) the Creditors' Committee and the Plan Committee and, solely in their respective capacities as members or representatives of the Creditors' Committee and the Plan Committee, as applicable (and not as individual lenders or creditors to or on behalf of the Debtors), each member of the Creditors' Committee and the Plan Committee; (ii) the Plan Administrator; (iii) the Unsecured Note Indenture Trustee; and (iv) each of their respective agents, advisors, accountants, investment bankers, consultants, attorneys and other representatives of any of the foregoing or of the Debtors, solely in their respective capacities as such, and only with respect to their activities and conduct during or in connection with the preparation and filling of the Chapter 11 Cases, from any and all claims, obligations, suits, judgments, damages, rights, causes of action and liabilities whatsoever (other than the right to enforce the performance of their respective obligations, if any, to the Debtors or the Reorganized Debtors under the Plan, the Plan Administrator Agreement and the contracts, instruments, releases and other agreements delivered under the Plan and the Plan Administrator Agreement), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are based in whole or in part on any act or omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Chapter 11 Cases, the Plan or the Disclosure Statement, other than Claims or liabilities arising out of or relating to any act or omission that constitutes a failure to perform the duty to act in good faith and where such failure to perform constitutes willful misconduct, gross negligence, or fraud.

#### (b) Injunction Related to Releases.

The Confirmation Order will permanently enjoin the commencement or prosecution by any entity, whether directly, derivatively or otherwise, of any claims, obligations, suits, judgments, damages, demands, debts, rights, causes of action or liabilities released pursuant to the Plan.

12.3 Discharge of Claims and Termination of Interests. Pursuant to section 1141(d)(3) of the Bankruptcy Code, Confirmation will not discharge Claims against the Debtors; provided, however, that no Holder of a Claim against any Debtor may, on account of such Claim, seek or receive any payment or other Distribution from, or seek recourse against, any Debtor or Reorganized Debtor or each of their respective successors or their respective property, except as expressly provided herein. Accordingly, except as otherwise provided herein, the Confirmation Order shall provide, among other things, that from and after the Confirmation Date, all Persons who have held, hold, or may hold Claims against or Interests in the Debtors are (i) permanently enjoined from taking the following actions against the Estate(s) of the Debtors, the Plan Administrator, the Plan Committee, the Reorganized Debtors, the Unsecured Note Indenture Trustee, or any of their property on account of such Claims or Interests and (ii) preliminarily enjoined from taking any of the following actions against any of the Debtors, the Plan Administrator, the Plan Committee, or the Reorganized Debtors, the Unsecured Note Indenture Trustee, or their property on account of such Claims or Interests: (A) commencing or continuing, in any manner or in any place, any action or other proceeding; (B) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree, or order; (C) creating, perfecting, or enforcing any lien or encumbrance; and (D) commencing or continuing, in any manner or in any place, any action that does not comply with or is inconsistent with the provisions of the Plan; provided, however, that (x) nothing contained herein shall preclude such Persons from exercising their rights pursuant to and consistent with the terms of the Plan and (y) the preliminary injunction of actions against the Debtors, the Plan Administrator, the Plan Committee, the Reorganized Debtors, the Unsecured Note Indenture Trustee, and their property (if any) shall be dissolved and terminate one (1) day following the dissolution of the Reorganized Debtors and the completion and winding-up of their affairs. Notwithstanding anything to the contrary in the Plan, creditors' rights of setoff and recoupment are preserved, and the injunctions referenced in this section or section 12.6 of the Plan shall not enjoin the valid exercise of such rights of setoff and recoupment. By accepting Distributions pursuant to the Plan, each Holder of an Allowed Claim or Allowed Interest shall be deemed to have specifically consented to the injunctions set forth in this Article XII.

12.4 Exculpation and Limitation of Liability. Except as otherwise specifically provided in the Plan, to the maximum extent permitted by the Bankruptcy Code and applicable law, none of (a) the Debtors, (b) the Reorganized Debtors, (c) the Plan Administrator, (d) the Creditors' Committee, (e) the Plan Committee, (f) the Unsecured Note Indenture Trustee, nor (g) any of their respective members, officers, directors, shareholders, employees, advisors,

attorneys or agents acting in such capacity on or after the Petition Date, shall have or incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or (with respect to such Claims or Interests) any of their respective agents, affiliates, or any of their successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Debtors' Chapter 11 Cases, the pursuit of confirmation of the Plan, the consummation of the Plan or the administration of the Plan or the property to be distributed under the Plan, except for their willful misconduct or gross negligence (the "Precluded Claims"), and in all respects shall be entitled to rely reasonably upon the advice of counsel with respect to their duties and responsibilities under the Plan.

Release by Holders of Claims. Except as otherwise specifically provided in the Plan, on the Effective Date, each Person (excluding any of the Debtors) that votes to accept the Plan shall release unconditionally, and hereby is deemed to forever release unconditionally (i) the Creditors' Committee and the Plan Committee and, solely in their respective capacities as members or representatives of the Creditors' Committee and the Plan Committee, as applicable (and not as individual lenders or creditors to or on behalf of the Debtors), each member of the Creditors' Committee and the Plan Committee; (ii) the Plan Administrator; (iii) the Unsecured Note Indenture Trustee; and (iv) each of their respective agents, advisors, accountants, investment bankers, consultants, attorneys and other representatives of any of the foregoing or of the Debtors, solely in their respective capacities as such, and only with respect to their activities and conduct during or in connection with the preparation and filing of the Chapter 11 Cases, from any and all claims, obligations, suits, judgments, damages, rights, causes of action and liabilities whatsoever (other than the right to enforce the performance of their respective obligations, if any, to the Debtors or the Reorganized Debtors under the Plan, the Plan Administrator Agreement and the contracts, instruments, releases and other agreements delivered under the Plan and the Plan Administrator Agreement), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are based in whole or in part on any act or omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Chapter 11 Cases, the Plan or the Disclosure Statement other than Claims or liabilities arising out of or relating to any act or omission that constitutes a failure to perform the duty to act in good faith and where such failure to perform constitutes willful misconduct, gross negligence, or fraud; provided, that this Section 12.5 shall not release any Person from any Claim or cause of action existing as of the Effective Date, based on (x) the Internal Revenue Code or any other domestic state, city or municipal tax code, (y) the environmental laws of the United States or any domestic state, city or municipality or (z) any criminal laws of the United States or any domestic state, city or municipality.

12.6 Injunction. Except as otherwise provided in the Plan, the Confirmation Order shall provide, among other things, that from and after the Confirmation Date all Persons who have held, hold or may hold Claims against or Interests in the Debtors are (i) permanently enjoined from taking any of the following actions against the Estate(s), the Plan Administrator,

the Creditors' Committee (or any of its members from time to time), the Plan Committee (or any of its members from time to time), the Unsecured Note Indenture Trustee, or any of their property on account of any such Claims, Interests or Precluded Claims and (ii) preliminarily enjoined from taking any of the following actions against any of the Debtors, the Reorganized Debtors, the Plan Administrator, the Creditors' Committee (or any of its members from time to time), the Plan Committee (or any of its members from time to time) or their property on account of such Claims, Interests or Precluded Claims: (A) commencing or continuing, in any manner or in any place, any action or other proceeding; (B) enforcing, attaching, collecting or recovering in any manner any judgment, award, decree or order; (C) creating, perfecting or enforcing any lien or encumbrance; (D) asserting a setoff, right of subrogation or recoupment of any kind against any debt, liability or obligation due to the Debtors; and (E) commencing or continuing, in any manner or in any place, any action that does not comply with or is inconsistent with the provisions of the Plan. By accepting Distributions pursuant to the Plan, each Holder of an Allowed Claim receiving any Distribution pursuant to the Plan will be deemed to have specifically consented to the injunctions set forth in this Article XII.

provided for in the Chapter 11 Cases under section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until all property of the Estates of the Reorganized Debtors has been distributed, the Reorganized Debtors have been dissolved, the Plan Administrator Agreement has terminated and the Bankruptcy Court has entered an order closing the Chapter 11 Cases; provided, however, that any injunction that by its terms is permanent or otherwise is intended to survive the Effective Date and Distributions hereunder (whether by law or pursuant to order of the Court) shall be continued without modification, notwithstanding anything to the contrary in the Plan.

#### ARTICLE XIII

#### RETENTION OF JURISDICTION

- 13.1 Retention of Jurisdiction. Pursuant to sections 105(c) and 1142 of the Bankruptcy Code and notwithstanding entry of the Confirmation Order and the occurrence of the Effective Date, the Bankruptcy Court shall retain exclusive jurisdiction over all matters arising out of, and related to, the Chapter 11 Cases and the Plan to the fullest extent permitted by law, including, among other things, jurisdiction to:
- (a) Allow, disallow, determine, liquidate, classify, estimate or establish the priority or secured or unsecured status of any Claim or Interest, including the resolution of any request for payment of any Administrative Claim and the resolution of any objections to the allowance or priority of Claims or Interests;
- (b) Resolve any matters related to the assumption, assumption and assignment, or rejection of any executory contract or unexpired lease to which any Debtor is a

party or with respect to which any Debtor or the Reorganized Debtors may be liable and to hear, determine and, if necessary, liquidate any Claims arising therefrom;

- (c) Ensure that Distributions to Holders of Allowed Claims are accomplished pursuant to the provisions of the Plan;
- (d) Decide or resolve any motions, adversary proceedings, contested or litigated matters and any other matters and grant or deny any applications involving the Debtors that may be pending on the Effective Date;
- (e) Approve such orders as may be necessary or appropriate to implement or consummate the provisions of the Plan and all contracts, instruments, releases and other agreements or documents created in connection with the Plan, the Disclosure Statement or the Confirmation Order;
- (f) Resolve any cases, controversies, suits or disputes that may arise in connection with the consummation, interpretation or enforcement of the Plan or any contract, instrument, release or other agreement or document that is executed or created pursuant to the Plan, or any entity's rights arising from or obligations incurred in connection with the Plan or such documents;
- (g) Modify the Plan before or after the Effective Date pursuant to section 1127 of the Bankruptcy Code or modify the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with the Plan, the Disclosure Statement or the Confirmation Order, or remedy any defect or omission or reconcile any inconsistency in any Bankruptcy Court order, the Plan, the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with the Plan, the Disclosure Statement or the Confirmation Order, in such manner as may be necessary or appropriate to consummate the Plan;
- (h) Hear and determine all applications for compensation and reimbursement of expenses of Professionals under the Plan or under sections 330, 331 and 503(b) of the Bankruptcy Code, *provided, however*, that from and after the Effective Date the payment of fees and expenses of the Reorganized Debtors, the Plan Committee, the Plan Administrator, and the Fee Auditor, including counsel fees, shall be made in the ordinary course of business and shall not be subject to the approval of the Bankruptcy Court;
- (i) Issue injunctions, approve and implement other orders or take such other actions as may be necessary or appropriate to restrain interference by any entity with consummation, implementation or enforcement of the Plan or the Confirmation Order;
- (j) Hear and determine the causes of action by or on behalf of the Debtors or the Reorganized Debtors, including causes of action relating to the Litigation Claims,

including avoidance actions under sections 544, 545, 547, 548, 549 and 553 of the Bankruptcy Code and turnover actions under sections 542 and 543 of the Bankruptcy Code;

- (k) Hear and determine matters concerning state, local and federal taxes in accordance with sections 346, 505 and 1146 of the Bankruptcy Code;
- (l) Approve and implement such orders as are necessary or appropriate if the Confirmation Order is for any reason or in any respect modified, stayed, reversed, revoked or vacated or Distributions pursuant to the Plan are enjoined or stayed;
- (m) Determine any other matters that may arise in connection with or relate to the Plan, the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with the Plan, the Disclosure Statement or the Confirmation Order;
- (n) Enforce all orders, judgments, injunctions, releases, exculpations, indemnifications and rulings entered in connection with the Chapter 11 Cases;
- (o) Hear and determine all matters related to (i) the property of the Estates from and after the Confirmation Date, (ii) the winding up of the Debtors' affairs and (iii) the activities of the Reorganized Debtors and the Plan Administrator, including (A) challenges to or approvals of the Reorganized Debtors' or the Plan Administrator's activities, (B) resignation, incapacity or removal of the Plan Administrator and selection of a successor Plan Administrator, (C) reporting by, termination of and accounting by the Reorganized Debtors and the Plan Administrator and (D) release of the Plan Administrator from its duties;
- (p) Hear and determine disputes with respect to compensation of (i) the Reorganized Debtors' professional advisors, (ii) the Plan Administrator and its professional advisors, and (iii) the Plan Committee, its members and its professional advisors;
- (q) Hear and determine such other matters as may be provided in the Confirmation Order or as may be authorized under the Bankruptcy Code; and
  - (r) Approve an order closing the Chapter 11 Cases.

#### ARTICLE XIV

#### MISCELLANEOUS PROVISIONS

14.1 Effectuating Documents and Further Transactions. Each of the Debtors or the Plan Administrator, on behalf of the Reorganized Debtors, as applicable, is authorized to execute, deliver, file or record such contracts, instruments, releases and other agreements or documents and take such actions as may be necessary or appropriate to effectuate,

implement and further evidence the terms and conditions of the Plan and any notes or securities issued pursuant to the Plan.

14.2 Corporate Action. Prior to, on or after the Effective Date (as appropriate), all matters provided for under the Plan that would otherwise require approval of the stockholders or directors of one (1) or more of the Debtors or the Reorganized Debtors shall be deemed to have occurred and shall be in effect prior to, on or after the Effective Date (as appropriate) pursuant to the applicable general corporation law of the states in which the Debtors or the Reorganized Debtors are incorporated without any requirement of further action by the stockholders or directors of the Debtors or the Reorganized Debtors.

#### 14.3 Bar Dates for Certain Claims.

- (a) Administrative Claims; Substantial Contribution Claims. The Confirmation Order will establish an Administrative Claims Bar Date for Filing Administrative Claims (including requests under section 503(b) of the Bankruptcy Code by any entity for making a substantial contribution in the Chapter 11 Cases) which date will be thirty (30) days after the Confirmation Date, except with respect to Fee Claims, which shall be subject to the provisions of Section 14.4. Holders of asserted Administrative Claims, except for Fee Claims, not paid prior to the Confirmation Date shall submit proofs of Claim on or before such Administrative Claims Bar Date or forever be barred from doing so. The notice of Confirmation to be delivered pursuant to Fed. R. Bankr. P. 3020(c) and 2002(f) will set forth such date and constitute notice of this Administrative Claims Bar Date. The Reorganized Debtors (in consultation with the Plan Committee) shall have sixty (60) days from the Effective Date (or such longer period as may be allowed by order of the Bankruptcy Court upon the request of the Reorganized Debtors or the Plan Committee) following the Administrative Claims Bar Date to review and object to such Administrative Claims before a hearing for determination of allowance of such Administrative Claims.
- (b) Professional Fee Claims. All Professionals and other entities requesting compensation or reimbursement of Fee Claims pursuant to section 327, 328, 330, 331 or 503(b) of the Bankruptcy Code for services rendered prior to the Effective Date shall File and serve on the Reorganized Debtors, counsel for the Plan Administrator and counsel for the Plan Committee, an application for final allowance of compensation and reimbursement of expenses no later than sixty (60) days after the Effective Date, unless otherwise ordered by the Bankruptcy Court. Objections to applications of such Professionals or other entities for compensation or reimbursement of expenses must be Filed and served on the Reorganized Debtors, counsel for the Plan Administrator, counsel for the Plan Committee and the requesting Professional or other entity no later than thirty (30) days (or such longer period as may be allowed by order of the Bankruptcy Court) after the date on which the applicable request for compensation or reimbursement was served. Except as otherwise provided in the Plan, Professionals shall be paid pursuant to sections 328, 330, or 331 of the Bankruptcy Code and prior orders of the Bankruptcy Court for amounts earned through the Effective Date. After the Effective Date, the Plan

Administrator shall, in the ordinary course of business and without the necessity for any approval by the Bankruptcy Court, pay the reasonable fees and expenses, incurred after the Effective Date, of the professional persons employed by the Debtors, the Reorganized Debtors, the Creditors' Committee and Plan Committee, and of the Fee Auditor, as the case may be, in connection with the implementation and consummation of the Plan, the claims reconciliation process, and any other matters as to which such professionals may be engaged. The fees and expenses of such professionals shall be paid within ten (10) calendar days after submission of a detailed invoice therefor. If the Plan Administrator disputes the reasonableness of any such invoice, the Plan Administrator shall timely pay the undisputed portion of such invoice, and the Plan Administrator or the affected professional may submit such dispute to the Bankruptcy Court for a determination of the reasonableness of such invoice. Upon the Effective Date, any requirement that Professionals comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date shall terminate, and the Plan Administrator, in consultation with the Plan Committee, may employ any Professional in the ordinary course of business.

- Professional Fee Holdback. On or as soon as reasonably practicable after the Effective Date and entry of an order of the Bankruptcy Court authorizing the Debtors to pay all amounts owing to Professionals for all outstanding amounts relating to prior periods through the Effective Date, the Debtors shall pay such amounts to such Professionals in accordance with such order; provided, however, that Professionals shall continue to prepare monthly fee applications in accordance with the Fee Order up to the Effective Date. Professionals shall estimate fees and expenses due for periods that have not been billed as of the Effective Date. If a Professional does not provide an estimate, the Debtors may estimate the unbilled fees and expenses of such Professional. On the Effective Date, the Reorganized Debtors shall fund an escrow account in an amount equal to the aggregate amount of outstanding fee applications not ruled upon by the Bankruptcy Court as of the Effective Date plus the aggregate amount of all estimated fees and expenses due for periods that have not been billed as of the Effective Date. Such escrow account shall not be considered property of the Reorganized Debtors. Such escrow account shall be used by the Reorganized Debtors to pay the remaining Professional Claims owing to the Professionals as and when Allowed by the Bankruptcy Court. When all Professional Claims have been paid in full, amounts remaining in such escrow account, if any, shall be returned to the Reorganized Debtors. To the extent that the escrow does not contain sufficient funds to pay all Professional Claims, such Professional Claims shall be paid from funds held in the Operating Reserve.
- 14.4 Payment of Statutory Fees. All fees payable pursuant to 28 U.S.C. § 1930, as determined by the Bankruptcy Court at the Confirmation Hearing, shall be paid on the Effective Date. The Debtors and the Reorganized Debtors shall remain liable for any quarterly fees validly due and owing to the United States Trustee under 28 U.S.C. § 1930 through and including such dates that their respective Chapter 11 Cases are converted to cases under chapter 7, dismissed, or closed.

52

- Bankruptcy Code and, to the extent applicable, sections 1122, 1123 and 1125 of the Bankruptcy Code, the Plan Proponents (subject to the consent of the other Plan Proponent, as applicable) reserve the right to alter, amend or modify the Plan at any time prior to or after the Confirmation Date but prior to the Effective Date. A Holder of a Claim or Interest that has accepted the Plan shall be deemed to have accepted the Plan, as altered, amended or modified, if the proposed alteration, amendment or modification does not materially and adversely change the treatment of the Claim or Interest of such Holder.
- term or provision of the Plan is determined by the Bankruptcy Court to be invalid, void or unenforceable, the Bankruptcy Court shall have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision will then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of the Plan will remain in full force and effect and will in no way be affected, impaired or invalidated by such holding, alteration or interpretation. The Confirmation Order will constitute a judicial determination and will provide that each term and provision of the Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.
- 14.7 Successors and Assigns. The Plan shall be binding upon and inure to the benefit of the Debtors, and their respective successors and assigns, including, without limitation, the Reorganized Debtors. The rights, benefits and obligations of any entity named or referred to in the Plan shall be binding on, and shall inure to the benefit of any heir, executor, administrator, successor or assign of such entity.
- things, the forms of the documents relating to the Amended Certificate of Incorporation, By-Laws and the Plan Administrator Agreement, shall be (i) Filed with the Bankruptcy Court and (ii) posted on the website established for these cases located at www.kccllc.net/verasun, no later than three business days prior to the Voting Deadline. Upon its Filing, the Plan Supplement may be inspected in the office of the clerk of the Bankruptcy Court or its designee during normal business hours. Holders of Claims and Interests may obtain a copy of the Plan Supplement upon written request to the Debtors in accordance with Section 14.10 of the Plan. The documents contained in the Plan Supplement shall be approved by the Bankruptcy Court pursuant to the Confirmation Order.
- 14.9 Revocation, Withdrawal or Non-Consummation. (a) The Plan Proponents, by mutual agreement, each reserve the right to revoke or withdraw the Plan as to any or all of the Debtors prior to the Confirmation Date and to File subsequent plans of reorganization. If either of the Plan Proponents revokes or withdraws the Plan as to any or all of the Debtors, or if confirmation or consummation of the Plan as to any or all of the Debtors does

not occur, then, with respect to such Debtors, (a) the Plan shall be null and void in all respects, (b) any settlement or compromise embodied in the Plan (including the fixing or limiting to an amount certain any Claim or Class of Claims), assumption or rejection of executory contracts or leases affected by the Plan, and any document or agreement executed pursuant to the Plan, shall be deemed null and void, and (c) nothing contained in the Plan shall (i) constitute a waiver or release of any Claims by or against, or any Interests in, such Debtors or any other Person, (ii) prejudice in any manner the rights of such Debtors or any other Person, or (iii) constitute an admission of any sort by the Debtors or any other Person.

14.10 *Notice.* All notices, requests and demands to or upon the Debtors, the Reorganized Debtors, the Creditors' Committee or the Plan Committee, to be effective shall be in writing and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows (or at such other address for such entity as shall be specified by like notice):

#### If to the Debtors:

VeraSun Energy Corporation 110 N. Minnesota Ave., Suite 300 Sioux Falls, SD 57104 Telephone: (605) 696-7200 Facsimile: (605) 978-7050

Attn: Mark Dickey

#### with copies to:

SKADDEN, ARPS, SLATE, MEAGHER & FLOM LLP
One Rodney Square
P.O. Box 636
Wilmington, Delaware 19899
Telephone: (302) 651-3000
Facsimile: (302) 651-3001

Attn: Mark S. Chehi, Esq. Davis Lee Wright, Esq.

#### - and -

SKADDEN, ARPS, SLATE, MEAGHER & FLOM LLP 155 North Wacker Drive Chicago, Illinois 60606-1285 Telephone: (312) 407-0700 Facsimile: (312) 407-0411

Attn: Felicia Gerber Perlman, Esq.

John K. Lyons, Esq.

#### If to the Reorganized Debtors:

Reorganized VeraSun Energy Corporation 110 N. Minnesota Ave., Suite 300 Sioux Falls, SD 57104 Telephone: (605) 696-7200 Facsimile: (605) 978-7050 Attn: Mark Dickey

with copies to:

SKADDEN, ARPS, SLATE, MEAGHER & FLOM LLP
One Rodney Square
P.O. Box 636
Wilmington, Delaware 19899
Telephone: (302) 651-3000
Facsimile: (302) 651-3001
Attn: Mark S. Chehi, Esq.
Davis Lee Wright, Esq.

- and -

SKADDEN, ARPS, SLATE, MEAGHER & FLOM LLP
155 North Wacker Drive
Chicago, Illinois 60606-1285
Telephone: (312) 407-0700
Facsimile: (312) 407-0411
Attn: Felicia Gerber Perlman, Esq.
John K. Lyons, Esq.

#### If to the Creditors' Committee:

AKIN GUMP STRAUSS HAUER & FELD LLP One Bryant Park New York, New York 10036 Telephone: (212) 872-1000 Facsimile: (212) 872-1002 Attn: Michael S. Stamer, Esq. David H. Botter, Esq. Alexis Freeman, Esq.

If to the Plan Committee:

AKIN GUMP STRAUSS HAUER & FELD LLP One Bryant Park New York, New York 10036 Telephone: (212) 872-1000 Facsimile: (212) 872-1002 Attn: Michael S. Stamer, Esq. David H. Botter, Esq. Alexis Freeman, Esq.

- 14.11 Governing Law. Except to the extent the Bankruptcy Code, the Bankruptcy Rules or other federal law is applicable, or to the extent a schedule to the Plan or a document or instrument contained in the Plan Supplement provides otherwise, the rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, the laws of Delaware, without giving effect to the principles of conflicts of law of such jurisdiction.
- 14.12 *Schedules.* All schedules, exhibits or other related documents to the Plan and the Plan Supplement are incorporated and are a part of the Plan as if set forth in full herein.
- 14.13 Filing of Additional Documents. On or before the Effective Date, the Debtors shall File such agreements and other documents, in form and substance acceptable the Creditors' Committee, as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.

Dated: Wilmington, Delaware

July 31, 2009

Respectfully submitted,

VERASUN ENERGY CORPORATION (for itself and on behalf of the Affiliate Debtors)

By: /s/ Mark Dickey

Name: Mark Dickey

Title: Senior Vice President, Secretary and

General Counsel

#### Counsel:

#### /s/ Mark S. Chehi

Mark S. Chehi (I.D. No. 2855)
Davis Lee Wright (I.D. No. 4324)
Skadden, Arps, Slate, Meagher & Flom LLP
One Rodney Square
P.O. Box 636
Wilmington, DE 19899
(302) 651-3000

- and -

Felicia Gerber Perlman John K. Lyons Skadden, Arps, Slate, Meagher & Flom LLP 155 North Wacker Drive Chicago, Illinois 60606 (312) 407-0700

Attorneys for VeraSun Energy Corporation, et al., Debtors and Debtors-in-Possession.

#### /s/ Dennis A. Meloro

Dennis A. Meloro (No. 4435) Donald J. Detweiler (No. 3087) Sandra G. M. Selzer (No. 4283) Greenberg Traurig, LLP The Nemours Building 1007 North Orange Street, Suite 1200 Wilmington, DE 19801 (301) 661-7000

- and -

Michael S. Stamer
David H. Botter
Akin Gump Strauss Hauer & Feld LLP
One Bryant Park
New York, New York 10036
(212) 872-1000

Attorneys for the Official Committee of Unsecured Creditors of VeraSun Energy Corporation, et al.

### Exhibit C

Form of Notice of Effective Date

### IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

ln re:	Case No. 08-12606 (BLS)
VERASUN ENERGY CORPORATION, et al.,  Debtors. 1	Chapter 11 Jointly Administered
	) 

# NOTICE OF (A) ENTRY OF ORDER CONFIRMING THE JOINT PLAN OF LIQUIDATION OF VERASUN ENERGY CORPORATION AND ITS AFFILIATED DEBTORS AND (B) OCCURRENCE OF THE EFFECTIVE DATE

- Court for the District of Delaware (the "Bankruptcy Court") entered an order (the "Confirmation Order") confirming the Joint Plan of Liquidation of VeraSun Energy Corporation and Its Affiliated Debtors, dated July 31, 2009 (the "Plan"), in the Chapter 11 Cases of the above-captioned debtors and debtors-in-possession (collectively, the "Debtors"). Unless otherwise defined in this Notice of (A) Entry of Order Confirming the Joint Plan of Reorganization of VeraSun Energy Corporation and its Affiliated Debtors and (B) Occurrence of the Effective Date (the "Notice"), capitalized terms and phrases used herein have the meaning(s) given to them in the Plan and the Confirmation Order.
- 2. **Discharge of Claims and Termination of Interests.** Except as otherwise provided in the Plan or in the Confirmation Order, all consideration distributed under the Plan shall be in exchange for, and in complete satisfaction, settlement, discharge, and release of, all Claims and Interests (other than those Claims that are Unimpaired under the Plan) of any nature whatsoever against the Debtors or any of their assets or properties, and regardless of whether any property shall have been distributed or retained pursuant to the Plan on account of such Claims and Interests. Upon the Effective Date, the Debtors, and each of them, shall be deemed discharged and released under section 1141(d)(1)(A) of the Bankruptcy Code from any and all Claims and Interests (other than Claims that are not Impaired), including, but not limited to, demands and liabilities that arose before the Confirmation Date, and all debts of the kind specified in section 502(g), 502(h), or 502(i) of the Bankruptcy Code, and the Old Equity shall be terminated.

The Debtors consist of: VeraSun Energy Corporation (EIN: 20-3430241); ASA OpCo Holdings, LLC (EIN: 68-0609122); US BioEnergy Corporation (EIN: 20-1811472); VeraSun Albert City, LLC (EIN: 20-2264707); VeraSun Albion, LLC (EIN: 55-0907221); VeraSun Aurora Corporation (EIN: 46-0462174); VeraSun BioDiesel, LLC (EIN: 20-3790860); VeraSun Bloomingburg, LLC (EIN: 55-0907224); VeraSun Central City, LLC (EIN: 55-0816855); VeraSun Charles City, LLC (EIN: 20-3735184); VeraSun Dyersville, LLC (EIN: 26-3594945); VeraSun Fort Dodge, LLC (EIN: 42-1630527); VeraSun Granite City, LLC (EIN: 20-5909621); VeraSun Hankinson, LLC (EIN: 20-2852457); VeraSun Hartley, LLC (EIN: 20-5381200); VeraSun Janesville, LLC (EIN: 20-4420290); VeraSun Linden, LLC (EIN: 55-0907228); VeraSun Litchfield, LLC (EIN: 20-8621370); VeraSun Marion, LLC (EIN: 20-3477343); VeraSun Marketing, LLC (EIN: 20-3693800); VeraSun Ord, LLC (EIN: 75-3204878); VeraSun Reynolds, LLC (EIN: 20-5914827); VeraSun Tilton, LLC (EIN: 26-1539139); VeraSun Welcome, LLC (EIN: 20-4115888); VeraSun Woodbury, LLC (EIN: 20-0647425).

#### 3. Injunctions.

- (a) Except as otherwise provided in the Plan, the Confirmation Order shall provide, among other things, that from and after the Confirmation Date, the satisfaction, release, and discharge pursuant to Article XII of the Plan shall act as an injunction against any Person commencing or continuing any action, employment of process, or act to collect, offset, or recover any Claim, Interest, or cause of action satisfied, released, or discharged under the Plan to the fullest extent authorized or provided by the Bankruptcy Code, including, without limitation, to the extent provided for or authorized by sections 524 and 1141 thereof.
- (b) By accepting Distributions pursuant to the Plan, each Holder of an Allowed Claim will be deemed to have specifically consented to the injunctions set forth in Article XII of the Plan.
- 4. Release by Debtors and Debtors in Possession. Pursuant to section 1123(b) of the Bankruptcy Code, and except as otherwise specifically provided in the Plan, upon the Effective Date, each of the Debtors shall release unconditionally, and hereby is deemed to forever release unconditionally (i) the Creditors' Committee and the Plan Committee and, solely in their respective capacities as members or representatives of the Creditors' Committee and the Plan Committee, as applicable (and not as individual lenders or creditors to or on behalf of the Debtors), each member of the Creditors' Committee and the Plan Committee; (ii) the Plan Administrator; (iii) the Unsecured Note Indenture Trustee; and (iv) each of their respective agents, advisors, accountants, investment bankers, consultants, attorneys and other representatives of any of the foregoing or of the Debtors, solely in their respective capacities as such, and only with respect to their activities and conduct during or in connection with the preparation and filing of the Chapter 11 Cases, from any and all claims, obligations, suits, judgments, damages, rights, causes of action and liabilities whatsoever (other than the right to enforce the performance of their respective obligations, if any, to the Debtors or the Reorganized Debtors under the Plan, the Plan Administrator Agreement and the contracts, instruments, releases and other agreements delivered under the Plan and the Plan Administrator Agreement), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are based in whole or in part on any act or omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Chapter 11 Cases, the Plan or the Disclosure Statement, other than Claims or liabilities arising out of or relating to any act or omission that constitutes a failure to perform the duty to act in good faith and where such failure to perform constitutes willful misconduct, gross negligence, or fraud.
- Release by Holders of Claims and Interests. Pursuant to section 12.5 of the Plan, on the Effective Date, each Person (excluding any of the Debtors) that votes to accept the Plan shall release unconditionally, and hereby is deemed to forever release unconditionally (i) the Creditors' Committee and the Plan Committee and, solely in their respective capacities as members or representatives of the Creditors' Committee and the Plan Committee, as applicable (and not as individual lenders or creditors to or on behalf of the Debtors), each member of the Creditors' Committee and the Plan Committee; (ii) the Plan Administrator; (iii) the Unsecured Note Indenture Trustee; and (iv) each of their respective agents, advisors, accountants, investment bankers, consultants, attorneys and other representatives of any of the foregoing or of the Debtors, solely in their respective capacities as such, and only with respect to their activities and conduct during or in connection with the preparation and filing of the Chapter 11 Cases, from any and all claims, obligations, suits, judgments, damages, rights, causes of action and liabilities whatsoever (other than the right to enforce the performance of their respective obligations, if any, to the Debtors or the Reorganized Debtors under the Plan, the Plan Administrator Agreement and the contracts,

instruments, releases and other agreements delivered under the Plan and the Plan Administrator Agreement), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are based in whole or in part on any act or omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Chapter 11 Cases, the Plan or the Disclosure Statement other than Claims or liabilities arising out of or relating to any act or omission that constitutes a failure to perform the duty to act in good faith and where such failure to perform constitutes willful misconduct, gross negligence, or fraud; provided, that this Section 12.5 shall not release any Person from any Claim or cause of action existing as of the Effective Date, based on (x) the Internal Revenue Code or any other domestic state, city or municipality or (z) any criminal laws of the United States or any domestic state, city or municipality or (z) any criminal laws of the United States or any domestic state, city or municipality.

Assumption of Executory Contracts and Unexpired Leases. On the Effective Date, pursuant to Article IX of the Plan and sections 365 and 1123(b) of the Bankruptcy Code, all executory contracts and unexpired leases of the Debtors shall be deemed rejected effective on and subject to the occurrence of the Effective Date of the Plan, unless such contract or lease (a) previously shall have been assumed or rejected by the Debtors, (b) previously shall have expired or terminated pursuant to its own terms before the Effective Date, (c) is identified in the Plan or Plan Supplement as a contract or lease to be assumed; or (d) is one of the Assumed Contracts. Entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of such assumptions pursuant to sections 365(a) and 1123 of the Bankruptcy Code. Each executory contract assumed pursuant to Article IX of the Plan shall revest in and be fully enforceable by the respective Reorganized Debtor in accordance with its terms, except as modified by the provisions of the Plan, or any order of the Bankruptcy Court authorizing and providing for its assumption or applicable federal law.

#### 7. Bar Dates

- (including requests under section 503(b) of the Bankruptcy Code by any entity for making a substantial contribution in the Chapter 11 Cases) shall be thirty (30) days after the Confirmation Date, except with respect to Fee Claims, which shall be subject to the provisions of paragraph 7(b) below. Holders of asserted Administrative Claims, except for Fee Claims, not paid prior to the Confirmation Date shall submit proofs of Claim on or before such Administrative Claims Bar Date or forever be barred from doing so. The notice of Confirmation to be delivered pursuant to Fed. R. Bankr. P. 3020(c) and 2002(f) shall set forth such date and constitute notice of this Administrative Claims Bar Date. The Reorganized Debtors (in consultation with the Plan Committee) shall have sixty (60) days from the Effective Date (or such longer period as may be allowed by order of the Bankruptcy Court upon the request of the Reorganized Debtors or the Plan Committee) following the Administrative Claims Bar Date to review and object to such Administrative Claims before a hearing for determination of allowance of such Administrative Claims.
- Applications. All Professionals and other entities requesting compensation or reimbursement of Fee Claims pursuant to section 327, 328, 330, 331 or 503(b) of the Bankruptcy Code for services rendered prior to the Effective Date shall File and serve on the Reorganized Debtors, counsel for the Plan Administrator and counsel for the Plan Committee, an application for final allowance of compensation and reimbursement of expenses no later than sixty (60) days after the Effective Date, unless otherwise ordered by this Court. Objections to applications of such Professionals or other entities for compensation or reimbursement of expenses must be Filed and served on the Reorganized Debtors, counsel for the Plan Administrator, counsel for the Plan Committee and the requesting Professional or other entity no later

than thirty (30) days (or such longer period as may be allowed by order of this Court) after the date on which the applicable request for compensation or reimbursement was served. Except as otherwise provided in the Plan, Professionals shall be paid pursuant to sections 328, 330, or 331 of the Bankruptcy Code and prior orders of this Court for amounts earned through the Effective Date. After the Effective Date, the Plan Administrator shall, in the ordinary course of business and without the necessity for any approval by this Court, pay the reasonable fees and expenses, incurred after the Effective Date, of the professional persons employed by the Debtors, the Reorganized Debtors, the Creditors' Committee and Plan Committee, and of the Fee Auditor, as the case may be, in connection with the implementation and consummation of the Plan, the claims reconciliation process, and any other matters as to which such professionals may be engaged. The fees and expenses of such professionals shall be paid within ten (10) calendar days after submission of a detailed invoice therefor. If the Plan Administrator disputes the reasonableness of any such invoice, the Plan Administrator shall timely pay the undisputed portion of such invoice, and the Plan Administrator or the affected professional may submit such dispute to this Court for a determination of the reasonableness of such invoice. Upon the Effective Date, any requirement that Professionals comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date shall terminate, and the Plan Administrator, in consultation with the Plan Committee, may employ any Professional in the ordinary course of business.

8. Effective Date. On [\_\_], the Effective Date of the Plan occurred.

VeraSun Energy Corporation

VeraSun Legal Information Website: http://www.kccllc.net/verasun

Dated: Wilmington, Delaware

[\_\_], 2009

SKADDEN, APRS, SLATE, MEAGHER & FLOM, LLP

Mark S. Chehi (I.D. No. 2855) Davis Lee Wright (I.D. No. 4324) One Rodney Square P.O. Box 636 Wilmington, DE 19899 (302) 651-3000

- and -

Felicia Gerber Perlman John K. Lyons 155 North Wacker Drive Chicago, Illinois 60606 (312) 407-0700

Counsel for Debtors and Debtors-in-Possession